Sheffield Hallam University

An integrated framework for improving supply chain performance

ELBEREGLI, Mukhtar Ali

Available from the Sheffield Hallam University Research Archive (SHURA) at:

http://shura.shu.ac.uk/23302/

A Sheffield Hallam University thesis

This thesis is protected by copyright which belongs to the author.

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author.

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given.

Please visit http://shura.shu.ac.uk/23302/ and http://shura.shu.ac.uk/information.html for further details about copyright and re-use permissions.

An Integrated Framework for Improving Supply Chain Performance

Mukhtar Ali Elberegli

A thesis submitted in partial fulfilment for the requirements of

Sheffield Hallam University

For the degree of Doctor of Philosophy

July 2018

Preface

This thesis has been carried out in accordance with the regulations of the Sheffield Hallam University as a part of the requirements for the degree of Doctor of Philosophy. The contents of this research studying have been carried out by the author were supervised by Professor Terrence Perera between May 2014 to July 2018.

Mukhtar A. Elberegli

July 2018

Acknowledgments

I cannot express the gratitude I have towards Allah Almighty for the mercy and blessings He has bestowed upon me to be able to complete my PhD.

To be able to work alongside my Professor Terrence Perera was an absolute pleasure. I'm deeply indebted to him for the time, friendship and inspiration he has consistently given me during my time under his supervision.

I would like to thank Llamafot Support team for their support with Supply Chain Guru Software, and especially to Tin Leung who always offers outstanding assistance.

I sincerely thank my beloved family for their patience and support. They missed my attention and presence throughout the period of my study and for that I am truly indebted to them.

Dedicate

This thesis is dedicated to my late mother (Fatima), my father (Ali) and my brothers and sisters (Fouziya, Husain, Halima, Mohamed, Basma, Hana, and Abdu-Allah).

Last but not least, to my darling wife (Manani) for her patience and sacrifice during my PhD journey. To my beloved children (Abduerehman, and Ali) for their endless love.

Abstract

In 2009, Roland Berger Strategy Consultants [Roland Berger Strategy Consultants, (2009). Global SCM excellence study., p.5.] reported that 40% of 234 companies had the wrong priorities in regard to efficiency vs. responsiveness. In 2014, PricewaterhouseCoopers (PwC) and American Production and Inventory Control Society (APICS) [PwC and APICS, Sustainable supply chains: Making value the priority 2014] found that 76% of 500 supply chain executives identified sustainability as an important aspect of their supply chain. The results highlight the importance of achieving consistency between customer expectations, in terms of cost and service level, and supply chain performance in today's competitive business environment. Despite this, however, no integrated supply chain design framework exists to control majority of the important functions related to supply chain strategy, structure, process and performance.

The literature review showed that simulation is rarely considered at the strategic level, but the research experiments highlighted a number of ways in which simulation tools might be useful at this level, such as exploring the impact of strategic fit and decoupling points, and assessing different supply chain network configurations and policies.

This research contributes to knowledge by designing and developing a framework that integrates strategy, process and resources, and allows the use of simulation tools to consider the three dimensions of efficiency, responsiveness and sustainability concurrently during the design process. The proposed framework is validated using a hypothetical supply chain network. Simulation allows performance to be assessed under a range of scenarios. The simulation experiments showed that under the suggested policies, efficiency improved from 25.38% to 30.58% and responsiveness rose from 18.37% to 32.78%. However, they also indicated that while policies oriented towards improving responsiveness had a positive impact on sustainability, those oriented towards improving efficiency had a negative impact.

The significance of the research lies in its development of a supply chain design framework that could assist companies in achieving the optimum configuration of supply chain resources, thereby helping them reduce inventory, lower costs, enhance responsiveness and improve strategic focus in terms of design, execution and capital investments.

List of abbreviations

ABDS	Agent-Based Decision Support
ABM	Agent-Based Modelling
ADI	Inter-demand interval
AT	Agent Technology
AHP	Analytic Hierarchy Process
ANP	Analytic Network Process
BPR	Business Process Reengineering
CASNs	Complex Adaptive Supply Networks
CLD	causal loop diagramming
CPFR	Collaborative Planning, Forecasting and Replenishment
CRS	Constant Returns to Scale
CV	Coefficient of Variation
DES	Discrete Event Simulation
ECM	Engineering Change Management
ERP	Enterprise Resource Planning
EM	Enterprise Modelling
ETO	Engineer-to-order
FEIs	Fast-Evolving Industries
GBOP	Generic-Bill-Of-Products
GSCF	Global Supply Chain Framework
GSCM	Green Supply Chain Management
HRM	Human Resource Management
JIT	Just-in-time
KPIs	key performance indicators
LARG	Lean, Agile, Resilient and Green)
LSCM	Logistics and Supply Chain Management
LT	Lead Time
МТО	Make-to-order
MTP	Make-to-plan
MTS	Make-to-stock
NPD	New Product Development
PDF	Probability Density Function
ROA	Return On Asset
RP	Reorder Point

List of abbreviations (Continued)

RQ	Reorder Quantity
SA	Sensitivity Analysis
SAP	System and Application Products
SC	Supply chain
SCD	supply chain design
SCG	Supply Chain Guru
SCI	supply chain integration
SCM	supply chain management
SCOR	Supply Chain Operations Reference Model
SCN	supply chain network
SCND	supply chain network design
SCS	supply chain strategy
SD	system dynamics
SM	Simulation Modelling
SPC	Statistical Process Control
SPSCs	Sustainability Performance in Supply Chains
SSCPM	supply chain performance measurement
SSO	Safety Stock Optimization
StDv	Standard Deviation
ТО	Transportation optimization
TQM	Total Quality Management
VAR	value at risk
VMI	vendor Managed Inventory
VRM	Value Reference Model
VRP	Vehicle Routing Problems

Table of Contents

Preface	I
Acknowledgments	
Dedicate	
Abstract	IV
List of abbreviations	V
List of abbreviations (Continued)	VI
List of Figures	X
List of Tables	XII
Chapter 1: Introduction	1
1.1 Background	1
1.2 Research Aim and Objectives	2
1.2.1 Aim	2
1.2.2 Objectives	2
1.3 Thesis Structure	3
Chapter 2: Literature Review	5
2.1 Introduction	5
2.2 Supply Chain Management	5
2.3 Supply Chain Design	6
2.3.1 Concept and scope	7
2.3.2 Frameworks and models	9
2.4 Supply Chain Strategy	13
2.4.1 Strategic fit	
2.4.2 Strategic supply chain management	
2.4.3 Green supply chain strategy	
2.5 Supply Chain Simulation	17
2.5.1 Design applications	
2.5.2 Supply chain enhancement applications	
2.5.3 Supply chain operation applications	
2.5.4 System Dynamics (SD) Simulation	
2.6 Supply Chain Performance	22
2.6.1 Supply chain efficiency	
2.6.2 Supply chain responsiveness	
2.6.3 Supply chain sustainability	

2.6.4 Performance measures of efficiency, responsiveness and sustainability	27
2.7 Conclusions	29
Chapter 3: Research Methodology	30
3.1 Introduction	30
3.2 Development of Supply Chain Management Frameworks	30
3.2.1 Characteristics of SCM framework in general	31
3.2.2 Method of developing SCM Strategy	32
3.3 The proposed conceptual SC framework	35
3.3.1 Supply chain strategy: achieving strategic fit	36
3.3.2 Supply chain operations reference (SCOR) model	41
3.3.3 Supply chain network design	46
3.4 Research Techniques and Tools	47
3.4.1 Conceptual techniques	47
3.4.2 Supply chain modelling	47
3.4.3 Supply chain simulation	50
3.4.4 Supply chain evaluation and what-if analysis	51
3.4.5 Statistical distribution	52
3.4.6 Sensitivity analysis	56
3.5 Conclusions	57
Chapter 4: Integrated Supply Chain Design Framework	58
4.1 Introduction	58
4.2 Strategic Objective Model	60
4.2.1 Demand characteristics	60
4.2.2 Suggested strategies	62
4.3 Process Model	63
4.3.1 Supply chain processes	64
4.3.2 Process configurations for lean strategy	65
4.3.3 Process configurations for agile strategy	66
4.3.4 Process configurations for leagile strategy	67
4.3.5 An overall map for implementing supply chain policy	68
4.4 Network Model	70
4.4.1 Drivers affecting supply chain network design	71
4.4.2 SCN configuration	74
4.5 Performance Indicators Model	76
4.5.1 Efficiency performance indicators	76

4.5.2 Responsiveness performance indicators	76
4.5.3 Sustainability performance indicators	77
4.6 Conclusions	77
Chapter 5: Supply Chain Design: Modelling and Simulation	79
5.1 Introduction	79
5.2 Background	79
5.3 Building the Supply Chain Guru Model	80
5.3.1 Products	80
5.3.2 Sites	81
5.3.3 Demand	81
5.3.4 Sourcing policy	81
5.3.5 Transportation policies	82
5.3.6 Inventory policies	83
5.4 Supply Chain Guru: Modelling, Optimization and Simulation	85
5.4.1 Case study	86
5.4.2 Overall method	86
5.4.3 Baseline model	90
5.4.4 Optimization of baseline model	94
5.4.5 Simulation scenarios in the baseline model1	03
5.5 Results of modelling and simulation10	05
5.5.1 Performance results from the baseline model1	05
5.5.2 Optimization of results from baseline model (predictable demand)1	05
5.5.3 Simulation results for efficient, responsive and sustainable SCD scenarios1	10
5.5.4 Discussion of results1	14
5.5.5 Validation of the proposed framework1	15
5.6 Conclusions1	16
Chapter 6: Conclusions, Recommendations, Contributions to knowledge and Future work1	18
6.1 Introduction1	18
6.2 Summary of Findings1	18
6.3 Contributions to Knowledge12	21
6.4 Limitations and Future Work12	23
References12	25

List of Figures

Figure 2. 1: Scope of supply chain management	6
Figure 2. 2: Supply chain management framework	.10
Figure 2. 3: SCOR map of the supply chain	.11
Figure 2. 4: Dimensions influencing supply chain design	.13
Figure 3. 1: SCM framework	.31
Figure 3. 2: Framework for choosing supply chain strategy	.33
Figure 3. 3: Theoretical structures underlying the framework	.35
Figure 3. 4: Finding the zone of strategic fit	.38
Figure 3. 5: Process of achieving strategic fit	.39
Figure 3. 6: SCOR process type and category	.44
Figure 3. 7: SCOR process hierarchy	.45
Figure 3. 8: Multi-echelon supply chain network	.46
Figure 3. 9: Supply chain modelling techniques	.51
Figure 3. 10: Categorization of demand patterns	.53
Figure 3. 11: Normal probability density function	.53
Figure 3. 12: Lognormal probability density functions	.54
Figure 3. 13: Poisson distribution for λ = 1, 2, 3, 4, 5, 6, 7, 10, and 20	.55
Figure 4. 1: Integrated supply chain design framework	.59
Figure 5. 1: Integrated solutions of Supply Chain Guru	.80
Figure 5. 2: R,Q policy	.84
Figure 5. 3: Demand flow policy	.84
Figure 5. 4: S,s policy	.85
Figure 5. 5: supply chain of the case study	.86
Figure 5. 6: SCG modelling and simulation method	.89
Figure 5. 7: Using a baseline model	.90
Figure 5. 8: Baseline model inputs and outputs	.90
Figure 5. 9: Demand classification	.95
Figure 5. 10: Smooth demand	.96
Figure 5. 11: Erratic demand	.97
Figure 5. 12: Slow demand	.98

Figure 5. 13: Lumpy demand	99
Figure 5. 14: Slow demand	100
Figure 5. 15: Shipment transactions excluding assets	102
Figure 5. 16: Unrouted shipment table	102
Figure 5. 17: Asset summary table	103

List of Tables

Table 5. 18: Different scenarios of unpredictable demand volume
Table 5. 19: Optimized inventory parameters for different unpredictable demand
scenarios108
Table 5. 20: Optimized transportation asset numbers for different unpredictable
demand scenarios109
Table 5. 21: Performance of optimized unpredictable demand baseline model
Table 5. 22: Performance of efficiency strategies under scenarios of predictable
demand111
Table 5. 23: Performance of responsiveness strategies under scenarios of
unpredictable demand112
Table 5. 24: Sustainability performance of efficiency and responsiveness
strategies under different scenarios of predictable and unpredictable demand
Table 5. 25: Profit to revenue (%) under lean strategy, plan and execute policy
Table 5. 25: Profit to revenue (%) under lean strategy, plan and execute policy(R,Q)
Table 5. 25: Profit to revenue (%) under lean strategy, plan and execute policy(R,Q)Table 5. 26: Profit to revenue (%) under lean strategy, continuous
Table 5. 25: Profit to revenue (%) under lean strategy, plan and execute policy(R,Q)
Table 5. 25: Profit to revenue (%) under lean strategy, plan and execute policy(R,Q)
Table 5. 25: Profit to revenue (%) under lean strategy, plan and execute policy(R,Q)
Table 5. 25: Profit to revenue (%) under lean strategy, plan and execute policy (R,Q) 115 Table 5. 26: Profit to revenue (%) under lean strategy, continuous replenishment policy (demand flow) 115 Table 5. 27: Avg. fill qty rate (%) under leagile strategy, postponement policy (T,S) 116 Table 5. 28: Avg. fill qty rate (%) under agile strategy, quick response policy

Chapter 1: Introduction

1.1 Background

Supply chains (SC) and logistics are the drivers of every economy. As global manufacturing has grown, effective and efficient SC and logistics management has become a key priority. Businesses are continually seeking to re-align their business models to ensure they serve their customers' needs better than their competitors. In this journey to improve customer service, businesses have used a wide variety of metrics to measure their performance, first in terms of efficiency and effectiveness, and more recently, in terms of sustainability. The emphasis now being placed on these three dimensions means that all three must be taken into account in the design of any SC.

Broadly speaking, the business of supply chain design (SCD) is to utilize resources efficiently in order to achieve defined outcomes. However, the literature review highlights that there is no general consensus among authors about how the concept should be defined. Baud-Lavigne (2012) and Melnyk (2014) defined SCD as the development, implementation and management of resources, processes and relationships across the SC. Mallidis, Dekker and Vlachos (2012) argued that SCD may be considered at two different levels: the strategic level and the tactical level, while Pistikopoulosb and Stuart (2013) and Leukel and Sugumaran (2013) argued that it has to support the company's strategic objectives. This lack of agreement about how to define SCD is the main reason why there is no comprehensive SCD framework; as Melnyk (2014) pointed out, numerous authors have focused on individual issues such as process, investment and structure, but no one has offered an overall framework to tie these issues together.

Chaharsooghi and Heydari (2011) emphasized that in highly turbulent environments, it is essential for the SC to focus on responsiveness to avoid losing customers. Accordingly, Wieland (2012) proposesd a model that enables companies to select from a range of SC strategies including agility, robustness, resilience and rigidity, depending on their assessment of risk probability and risk impact. Alfalla-Luqu, Medina-Lope and Dey (2013) extended this idea by building a conceptual framework (based on Fisher's (1997) prototypical efficient

and market responsive configurations) that also included social and environmental issues, while Um et al. (2017) developed a conceptual model linking product variety management strategies with SC responsiveness and cost. However, a review of the relevant literature indicates that no one has developed a design framework in which all three categories are considered concurrently.

This has now become possible with the emergence of SC-specific simulation software that is powerful enough to consider efficiency, responsiveness and sustainability simultaneously. Researchers have already employed simulation as a tool to explore a range of SC issues; Chaharsooghi and Heydar (2009), for example, investigated the effects of lead time on SC performance, while Ilaria (2011) used simulation to develop a model for strategic decision making and to investigate operational issues such as inventory and transportation effects. Finally, Tseng, Gung and Huang (2013) used simulation to define the impacts of operator parameters on total cost, penalty cost, fill rate and on-time delivery. Shahi and Pulkki (2013), looked to further extend its application, argue for simulation-based optimization models to provide much better solutions than current industrial practice.

To sum up: while numerous researchers have discussed SCD, there is no consensus on how the concept should be defined. Furthermore, no frameworks have yet been presented that integrate all SC functions or aim to improve multiple dimensions of SC performance. This research seeks to address this gap by developing a framework that adopts a broad and integrated approach to SCD and then using hypothetical supply chain network to validate it.

1.2 Research Aim and Objectives

1.2.1 Aim

The research aims to develop an integrated Framework for improving supply chain performance

1.2.2 Objectives

The following are the main objectives of this research:

- 1. To carry out a comprehensive literature review to establish current knowledge and practice.
- 2. To review the use of simulation in SCD.
- 3. To identify appropriate key performance indicators (KPIs) to consider efficiency, responsiveness and sustainability concurrently.
- 4. To review currently available SCD frameworks.
- 5. To design a preliminary framework.
- 6. To evaluate and improve this framework.
- 7. To validate the framework in the virtual environment using the simulation tool.

1.3 Thesis Structure

The objectives mentioned in the previous section are addressed in six chapters of this thesis. This chapter offers a brief introduction to the aims and objectives of the research. Chapter Two presents a comprehensive literature review discussing the various methodologies that have been employed by previous researchers in SCD. This chapter discusses definitions of SCD, existing frameworks, SC strategies and performance metrics.

Chapter Three discusses the methodology employed in the study. It describes the process of developing an SC management framework and introduces the features of integrated SCD, including the strategic fit concept underlying the strategic model within the proposed framework, the SCOR model (used to clarify SC process configurations) and the supply chain network design (SCND) concept. Finally, the chapter discusses research techniques and tools including SC modelling, optimization and simulation.

Chapter Four presents the proposed integrated SCD framework. This consists of four different models, each of which performs a defined role within the overall SCD. The strategic objective model sets out four different SC strategies to meet customer requirements in both certain and uncertain business environments, taking into account SC capabilities and target performance levels. The process model (the Supply Chain Operations Reference Model – SCOR) assists managers in implementing the right policies to maximize resource utilization and achieve the SC's strategic goals. The network model clarifies the way in which various entities in the supply chain network (SCN) are associated with each other and how their processes interact to achieve the goals of the SC. Finally, the performance indicators model measures SC performance and defines the extent to which it is meeting its objectives.

Chapter Five describes the modelling and simulation process that was undertaken to verify the proposed framework. This process was conducted using Llamasoft's Supply Chain Guru (SCG) software.

Chapter Six summarises and discusses the main points obtained from the research before offering recommendations for further study.

Chapter 2: Literature Review

2.1 Introduction

This chapter reviews the literature pertaining to supply chain design (SCD) and other related topics in the field of supply chain management (SCM). In doing this, it focuses on those issues that are relevant to the current study, rather than attempting to cover multidisciplinary literature.

The survival of a modern business depends on the effectiveness of its supply chain (Farahani et al., 2014). In designing this chain, the aim should be to achieve a level of efficiency that maximizes overall value and gives the company an advantage over its competitors. Since the success of a supply chain is highly dependent on the alignment between strategy and design (Melnyk, 2014), it is essential to adopt an integrated methodology that takes into account strategy, resources and performance.

2.2 Supply Chain Management

The term "supply chain management" was first coined in 1982 when Keith Oliver, a consultant at Booz Allen Hamilton, used it in an interview with the *Financial Times*. Since then, its development as a discipline has occurred primarily in the industrial sector. Melnyk (2014) argueed that:

"Supply chain management is a concept that has been born of practice, grown through need, and changed in response to various challenges, threats and opportunities. Consequently, until recently, it has largely not been theoretically grounded. Rather, attention has been devoted to understanding what supply chain management is (and is not), how it is related to similar approaches such as logistics, operations management and purchasing/sourcing management and how it affects performance".

Figure 2.1 presents an overview of the scope of supply chain management.



Figure 2. 1: Scope of supply chain management

Source: Bowersox, Closs, Cooper and Bowersox (2013)

Lu (2011) argued that as the business environment has changed, competition has become less a matter of organization versus organization than supply chain versus supply chain. As a result, a business's survival is no longer solely dependent on its ability to compete but rather on its ability to cooperate with others in the supply chain.

Wu, Melnyk and Flynn (2010) noted that supply chains have changed from being strategically decoupled and price-driven to strategically coupled and value-driven. They argue that:

"This transition is not simply a 'happy accident'. Rather, it is the result of deliberate management action and strategic corporate investments aimed to procure, develop and configure the appropriate supply chain resources that will allow the firm to compete successfully in the marketplace".

The concept of supply chain design lies at the very heart of these investment decisions.

2.3 Supply Chain Design

Fine (1998) was the first to recognize that supply chain design (SCD) now goes beyond issues of make/buy, buyer-supplier relationships or vertical integration

to encompass investment decision making. How a firm decides to distribute its investment across its various supply chains will affect the capabilities of each of these chains (Wu, Melnyk and Flynn, 2010) and shape the types of relationship that emerge between supply chain partners and the degree of transparency that is achieved (e.g. Closs and Mollenkopf, 2004; Janvier-James, 2012; Lambert, Cooper and Pagh, 1998; Samaranayake, 2005; Spens and Bask, 2002). Researchers have focused on SCD from both theoretical and empirical perspectives, but as Melnyk (2014) pointed out, many have focused on issues such as process, investments and structure without offering an overall framework to tie these aspects together. The following two sections discuss what the concept of supply chain design actually means and the frameworks that have been put forward for understanding this concept.

2.3.1 Concept and scope

Broadly speaking, the business of supply chain design is to utilize resources efficiently in order to achieve defined outcomes. However, the literature review highlights that there is no general consensus among authors about how the concept should be defined. Baud-Lavigne (2012) suggested that SCD may be considered at two different levels: the strategic level (e.g. the choice of production facilities, load/manufacturing capacities and technologies) and the tactical level (e.g. mid-term decision making on issues such as the choice of suppliers, the allocation of products to production facilities and the flow of each product and sub-assembly in the network) (Cordeau et al., 2006).

Leukel and Sugumaran (2013) appeared to take the strategic perspective with their argument that the supply chain has to be designed to support the strategic objectives of the firm, which they suggest involves making long-term decisions about products; process technologies; the number, location and capacity of SC nodes; production rates; and suppliers, markets and partners. Mallidis, Dekker and Vlachos (2012) also saw SCD as encompassing decisions about the number, location, capacity and operation of distribution centres/production facilities, and the selection of intermediaries and partners (suppliers, freight forwarders etc.). Melnyk (2014), meanwhile, defined supply chain design as identifying the desired strategic outcomes for the firm and developing, implementing and managing the resources, processes and relationships (within

the firm and across the supply chain) that will make the attainment of these outcomes inevitable over time.

At the tactical level, Metta and Badurdeen (2013) argued that supply chain design involves identifying product design criteria (e.g. materials, functionality, components and interfaces) and evaluating their impact on SC configuration (e.g. the number and location of SC partners, their capabilities and capacities) to achieve optimum SC performance.Prasad, Venkatasubbaiah and Rao (2014) explained that the design should aim to maximize overall value in the SC by optimizing transportation, inventory, operating facilities and information flow.

Complicating the issue further, the perceived scope of the SC design process seems to have changed over the years. Speier et al. (2011) argued that SCD decision makers have historically focused on how to minimize the total landed cost, for example by considering carefully where to locate facilities such as plants and warehouses and by controlling materials acquisition, production, inventory and logistics costs. However, Closs and McGarrell (2004) claimed that over time, SCD objectives have gone beyond cost, with chains now being expected not only to operate within designated cost parameters but also to meet the unique service requirements of different customer segments. Indeed, these objectives have recently extended even further to include consideration of the dimensions of security, risk and sustainability.

Govindan, Fattahi and Keyvanshokooh (2017) defined three types of uncertain environment in which SCD decision makers must operate. In the first, the decision-making environment has random parameters whose probability distributions are known to the decision maker. These are called stochastic parameters and are described by either continuous or discrete scenarios. In the second, there are again random parameters, but the decision maker has no information about their probability distributions. Under this setting, robust optimization models are usually developed with the purpose of optimizing the worst-case performance of the SC network. The third type of environment is the fuzzy decision-making environment. This is characterized by ambiguity (there is no clear choice between multiple alternatives) and vagueness (boundaries between some domains of interest are not clearly delineated). In this context, fuzzy mathematical programming handles the planner's expectations about the

level of objective function, the uncertainty range of coefficients, and the satisfaction level of constraints.

2.3.2 Frameworks and models

Frameworks and models help researchers to see clearly the essential elements of their interested research and guiding the entire process of the research study in order to achieve its aim.

The framework developed by Toit and Vlok (2014) (see Figure 2.2) offered a simple graphical representation which divides SCD into different components, defines the components and shows the relationships between them. In this way, the framework helps users to make sense of a complex concept in a practical manner. The framework starts with organisational strategy, highlighting the importance of the alignment between this and SC strategy. The next object in the framework is SCM, through whose plans the SC strategy is implemented. SCM has three main components: SC participants, SC life-cycle activities and SC support functions. SC participants link to both SCM plans and life-cycle activities. Performance measurement acts as a feedback loop into continuous improvement, which impacts on both SC strategy and management. The different components within SCM are all affected by enablers that act across functions, activities and participants.



Figure 2. 2: Supply chain management framework

Source: Toit and Vlok (2014)

Naslund and Williamson (2010) presented the Supply Chain Operations Reference (SCOR) model, developed by the Supply Chain Council (SCC) and AMR Research in 1996. According to the SCC, the supply chain operations reference (SCOR) model may be used to identify, measure, reorganize and improve supply chain processes. They claimed that it:

"...provides a unique framework that links business processes, metrics, best practices and technology features into a unified structure to support communication among supply chain partners and to improve the effectiveness of supply chain management and related supply chain improvement activities" (Supply Chain Council, 2009).

The validity of the SCOR model has been confirmed by Zhou et al. (2011), whose empirical findings generally support the relationships it posits between supply chain processes (plan, source, make and deliver).

A number of authors have discussed the SCOR model; Huan, Sheoran and Wang (2004), for example, noted its integration of BPR, benchmarking and process measurement within a cross-functional framework and employed the

analytic hierarchy process (AHP) to demonstrate its strength as decisionmaking tool. However, they also note that the model fails to consider change management or to supply quantifiable measurements of SC performance.



Figure 2. 3: SCOR map of the supply chain

Source: Kocao glu, Gülsün and Tanya (2011)

Fronia, Wriggers and Nyhuis (2008), while acknowledged that the SCOR model initially provides as universal as possible a description of the supply chain, show how it might be extended to offer a more detailed framework for supply chain design. The models they offer give a clearer explanation of each SCOR process. When Long (2014) developed a hierarchical framework for modelling supply chain networks based on an improved version of the SCOR model, he found that these networks generally consist of several entities, each of which may be composed of several departments or workshops. This led him to argue that any supply chain network can be divided into four levels: the supply chain network level, the enterprise level, the workshop level and production. He suggests that any element at any level can be modelled using the five core processes from the SCOR model.

Other attempts at a framework include that by Ivanov (2009), who employed software to develop and validate a complex mathematical model with the aim of increasing the efficiency, consistency, implacability and sustainability of SCD decision making and showing the links between the design, planning and implementation functions. Ivanov pointed to the need for further work to investigate the relationship between business processes and information systems, and suggested that researchers should consider the flow of financial data between departments alongside the flow of materials and information.

A number of authors have proposed five-step models. For Hilletofth (2012), these steps are: develop a segmentation model, collect market information, then specify, select and implement supply chain solutions. For Corominas et al. (2015), they are: define SC objectives and conduct environmental analysis; define SC macrostructure (activity blocks and the relationships between these blocks); define SC mesostructure (product structure and production process); define SC microstructure (demand, production activities and transportation); choose SC configuration and implement. Finally, Marchesini and Alcântara (2016) proposed: identify logistics activities; characterize these activities according to need and their impact on customer value and logistics service; assign logistics activities to companies; identify any gaps in internal coordination and integration; measure the performance of logistics activities.

The framework proposed by Affonso, Liu and Zolghadri (2013) integrated product and supply chain design. It consists of identifying and evaluating product functions, defining relevant SC structures, identifying and evaluating potential suppliers, selecting suppliers, and finally defining the supply chain configuration.

Melnyk, Narasimhan and DeCampos (2014) claimed that supply chain design is shaped by three dimensions that have a hierarchical relationship: these are influencers, design decisions and building blocks. Influencers are those factors that impact on overall SC performance such as the desired SC outcomes and the global environment. Design decisions are the specific decisions that must be made regarding the supply chain as a whole (e.g. network design, sourcing strategies), while building blocks are the investments that are required to implement these decisions and build the supply chain.



Figure 2. 4: Dimensions influencing supply chain design

Source: Melnyk, Narasimhan and DeCampos (2014)

Finally, Sabet, Yazdani and Leeuw (2017) developed a conceptual model specifically for SCM in fast-evolving industries (FEIs). Their model illustrates that the more important the supplied products/services are to a firm's core business, the more closely it must integrate with its suppliers to secure its value-creation processes and protect this core business. At the same time, if supply is associated with a high level of risk and uncertainty, the firm must aim for alignment, adaptability and agility within its SC.

It should be noted that whatever framework or model is applied, all SCD activities must be guided by the supply chain strategy to ensure that all decisions in the design stage contribute positively towards achieving the company's strategic objectives.

2.4 Supply Chain Strategy

Successful companies understand the value of focusing their energies on those dimensions where they can compete most effectively. In a survey of 234 companies from 16 countries, Roland Berger Strategy Consultants (2009) found that companies with supply chain fit achieve a Return on Assets (ROA) of 4-6% higher on average than companies without supply chain fit. Companies without supply chain fit tend to have the wrong priorities when designing their supply chain; companies with standardized products do not focus enough on cost, inventory management or utilization rates. On the other hand, companies with

customized products often do not focus enough on flexibility, delivery reliability and service level improvement.

2.4.1 Strategic fit

Randall, Morgan and Morton (2003) examined the association between product demand characteristics and the initial investment in a supply chain at the time of market entry. Characterizing supply chains as responsive or efficient, they conclude that responsive market entry is associated with lower industry growth rates, higher contribution margins and higher technological demand uncertainty.

Chaharsooghi and Heydari (2011) discussed the concept of strategic fit in supply chain management, concluding that in highly turbulent environments it is essential for the SC to focus on responsiveness to avoid losing customers. Soni and Kodali (2011) explored the strategic fit between competitive strategy (CS) and supply chain strategy (SCS) in the Indian manufacturing industry by investigating the mediating role of SCS between CS and company/supply chain performance. Their findings reveal a causal relationship between CS and SCS and that the choice of both affects both business and supply chain performance. Wagne, Gross-Ruyke and Erhun (2012) also investigated the relationship between supply chain fit and the financial performance of the firm. Their findings indicate that the higher the supply chain fit, the higher the ROA of the firm, and that firms with a negative misfit perform worse than firms with a positive misfit.

2.4.2 Strategic supply chain management

Hwang (2010) discussed how to develop a supply chain's overall competitive strategic direction so as to optimize SC performance. His general SSCM framework comprises three stages, namely strategy formulation, strategy implementation and strategy evaluation. Alfalla-Luqu, Medina-Lope and Dey (2013) identified information integration, coordination and resource sharing, and organisational relationship linkage as the three major dimensions of supply chain integration (SCI) and analysed how these affect overall supply chain performance in terms of efficiency and responsiveness. The authors offer an integratived model that blends together elements of supply chain configuration, stakeholder management and capability development. Their analysis revealed

that the nature of stakeholder exposure determines how social/environmental, technical and relational capabilities impact on social and environmental outcomes. Their framework builds on Fisher's (1997) prototypical efficient and market responsive configurations, expanding them to include social and environmental issues. Taking Apple/Foxconn as an example, they suggested that capabilities based upon responsiveness, such as product improvement and collaboration, may need to be supplemented with efficiency-oriented capabilities, such as process improvement and monitoring, to satisfy the demand for economic, social and environmental outcomes.

Wieland (2013) proposed a model that enables companies to select a supply chain strategy based on risk probability and risk impact. He identified four supply chain strategies – agility, robustness, resilience and rigidity – advising resilience where supply chain risk probability and impact are high, and rigidity where both values are low. When only risk impact is low, robustness is optimal, whereas agility is optimal when only risk probability is low.

Um et al. (2017), investigated the impact of product variety strategy on supply chain performance, developed a conceptual model that links product variety management strategies with supply chain responsiveness, cost and customer service in high- and low-customization environments. They found that product variety strategy influences supply chain cost and customer service performance only when mediated by internal and external responsiveness capabilities, and that its impact on performance depends on the level of product customization. In a low-customization environment, both supply chain flexibility and agility have a significant influence on cost efficiency, while in a high-customization environment, these dynamic capabilities have a significant influence on customer service.

Given the importance of environmental aspects in supply strategy, green supply chain strategy is discussed in a separate section.

2.4.3 Green supply chain strategy

Corporations are coming under increasing pressure to design their supply chains for sustainability. This means designing processes to use environmentally friendly inputs and to create outputs that can be recycled and

that do not contaminate the environment. One approach has been proposed by Cabral, Griloand and Cruz-Machado (2011), who offered an integrated Lean, Agile, Resilient and Green (LARG) analytic network process (ANP) model to help companies choose the most appropriate practices and KPIs for their SCs. They presented a case study (based on a car manufacturer's supply chain) to showcase the model's ability to prioritise enablers, KPIs, practices and paradigms in complex situations.

Yu et al. (2014) extended previous green supply chain management (GSCM) research by developing and empirically testing a conceptual framework that investigated the relationships between iGSCM (internal GSCM, GSCM with customers and GSCM with suppliers) and multiple dimensions of operational performance in terms of flexibility, delivery, quality and cost. Varsei et al. (2014), meanwhile, offered a framework that adopts a multidimensional approach, considering economic, environmental and social dimensions.

Youn, Yang and Roh (2012) explored how Fisher's perspective of efficient versus responsive supply chains can be a stepping stone to the development of eco-efficient and eco-responsive supply chains. In an eco-efficient supply chain, the focus is on maintaining high environmental standards across the chain, while the emphasis in an eco-responsive supply chain is on collaboration among suppliers and distributors.

Gracia and Quezada (2016) proposed combining three analytical tools (analytic hierarchical process, fuzzy multi-objective optimization and clustering methods) for the strategy formulation process and integrating corporate and supply chain strategy on the basis of sustainability. The results from their case study indicated that the methodology is a valid tool for generating a coordinated strategy for the management of a sustainable supply chain.

The literature emphasizes the link between achieving a green supply chain and improving economic performance; the European Freight and Logistics (EF&L) and United Nations Commission for Latin America and the Caribbean (UNCLAC) (2014), for example, co-authored a study designed to demonstrate the direct link between sustainability and efficiency/cost reduction. However, there is as yet no integrated strategic framework available for achieving supply chain management that is both green and competitive. It is therefore the aim of

this research to design and develop an integrated framework that encompasses not just the efficiency and responsiveness dimensions, but also that of sustainability.

The question of whether a company's supply chain strategy is indeed as efficient, responsive and sustainable as its designers intend can only be answered if it is continuously evaluated and improved. Simulation is an invaluable tool for observing and assessing the effectiveness of a company's strategic decision making.

2.5 Supply Chain Simulation

Researchers and practitioners in SCM employ a range of techniques to assess the impact and effectiveness of SCD decisions, including surveys, simulations, mathematical models, case studies and conceptual models. Simulation is considered a particularly valuable tool because it can be used to assess the potential cost of system changes and to model scenarios that would be difficult to apply in reality. Changes can be executed and systems can be observed, and the developments of years can be presented within hours. For these reasons, simulation is widely employed in SCM. In this study, simulation is used at the strategic level for the purpose of supply chain improvement, while followed sections investigate various simulation applications.

2.5.1 Design applications

As noted above, simulation is frequently used as a decision-making tool in supply chain design. Ilaria (2010) claimed that SCM integration is best pursued by adopting either an operative or an organizational approach. However, the way the SC is organized influences its performance. He employs an NK simulation model (in which N represents the decisions that SC firms should make and K the interdependencies among SC integration decisions), to identify the best forms of governance for tackling SC integration problems. Tako and Robinson (2012) concluded that both discrete event simulation (DES) and system dynamics (SD) have been used to model logistics and supply chain management (LSCM) issues. Hilletofth and La⁻⁻ttila (2012) investigated the use of agent-based decision support (ABDS) systems in the supply chain context,

finding that they enable increased versatility in the system architecture, improve supply chain visibility and allow users to conduct experiments and what-if analyses.

Another application of simulations is evaluating supply chain networks. Li et al. (2010), argued that supply chains should be treated as complex adaptive supply networks (CASNs), modelled their evolution using complex adaptive system and fitness landscape theory. They then conducted a case study of the evolution of a supply network in the emerging Chinese market. Their results led them to suggest that external environmental factors and firm-internal mechanisms appear to be the dominant forces shaping the evolution of CASNs, with cost and quality considerations being the primary factors influencing their structure, complexity, centralization and formalization. Pirard, lassinov-ski and Riane (2011) studied the problem of strategic network design in multi-site enterprises by modelling various supply network designs. Their simulation allows the computation of performance measures such as profitability and customer service, but the authors claim that the model could be improved by incorporating operational decision making, valuation of tardiness used in the allocation rules, transportation system capacity and the possibility of grouping orders. Finally, Porras and Zelaya (2012) offered a standardized simulation model for analysing distribution networks which is both designed to assist strategic decision making and can also take into account operational issues such as inventory and transportation effects.

Among those employing simulations to assess supply chain configurations, Persson and Araldi (2009) developed an Arena-based SCOR template, while Mittermayer and Rodriguez-Monroy (2013) presented a simulation-based evaluation method for the comparison of different organizational forms and software support levels. They found that coordination schemes based only on ERP systems are a valid alternative in industrial practice, and that these schemes represent a significant saving in terms of IT investment. Indeed, the authors conclude that spending more on coordination does not always result in improved logistics performance. The results confirm the importance of considering all dimensions when evaluating SCM concepts and IT tools.

2.5.2 Supply chain enhancement applications

Eulalia et al. (2010) investigated the robustness of different tactical planning and control policies for a softwood supply chain using an agent-based environment. Their simulations were modelled using a novel agent-based methodology combined with a robust experimental design approach. Their results indicated that supply chain control levels play a role in defining robust service levels, while the planning horizon and the planning method have lower impact in this context. Ferreira and Borenstein (2011) presented an agentbased simulation framework for supply chain planning. The study sought to investigate the role regulation plays in SCs by modelling the actors involved in the regulation of SCs using normative agents to allow evaluating the potential benefits of alternative strategies for planning of regulated SCs. The authors suggest that the developed model can be expanded to consider logistics by adding new agents and control agents, and by adding and removing norms. Rashid and Weston (2012) presented an integrated methodology for modelling complex supply chains which deploys enterprise modelling (EM), causal loop diagramming (CLD) and simulation modelling (SM) techniques.

Reddi and Moon (2012) studied the interactions between the various new product development (NPD) and engineering change management (ECM) process parameters by modelling the processes and simulated the model to understand the parameter interactions. The results indicate that most of the variables and interactions among the variables have a significant influence on the NPD is lead time. Shahza and HadjHamou (2013) proposed the notions of generic-bill-of-products (GBOP) to implement the concept of sustainable mass customization. Simulation results provided an optimum GBOP, its respective segments and decisions on the opening or closing of the market segments to sustain mass customization efforts.

Simulation is a powerful tool for mitigating uncertainty; Colicchia, Dallari and Melacini (2010) identified a set of approaches (mitigation actions and contingency plans) for managing risk in order to enhance supply chain resilience. They then apply a simulation-based framework to assess the effectiveness of the proposed approaches. Shahi and Pulkki (2013) reviewed the literature related to supply chain models in the forest products industry,

concluding that studies that focus on optimization are mostly deterministic in nature and do not account for uncertainty in either the supply of raw materials or the demand for forest products. They pointed to a need for the development of simulation-based optimization models that will meet industrial expectations and provide much better solutions than current industrial practice.

Simulations have also been used to improve performance. Chaharsooghi and Heydar (2009) investigated the effects of LT mean and LT variance on supply chain performance indices using simulations and multivariate models and found that LT variance has the stronger impact. This result may help practitioners develop investment strategies to reduce LT mean and variance. Finally, Vidalakis, Tookey and Sommerville (2011) used the simulation technique to investigate the applicability of logistics management in the construction sector. They utilized pre-existing data to build a model, which they then analysed using discrete-event simulation modelling. Their analysis shows that logistics costs are exponentially related to the level of material demand and the number of vehicle movements.

2.5.3 Supply chain operation applications

Simulation is considered an important tool in inventory management. Petrovic (2001) developed a simulation tool for analysing SC behaviour and performance in the presence of uncertainty. Fuzzy analytical models were employed to determine optimal order-up-to levels in a fuzzy environment, followed by a simulation model to evaluate SC performance over time at the order-up-to levels recommended by the fuzzy models. Lyu, Ding and Chen (2009) proposed three collaborative replenishment mechanism models for use in the collaborative supplier and store-level retailer environment. The models, which were developed based on a case grocery company, explore the impacts of different scenarios. The authors suggested further research is needed to multi-store-level-retailer discuss the multi-supplier and collaborative replenishment mechanism in which each supplier adopts an individual inventory control policy for different products.

Gumus, Guneri and Ulengin (2010) proposed a methodology for multi-echelon inventory management and presented a neural network simulation of a model

which they claimed represents an improvement on similar models (demand and lead time are not constant and it allows orders that arrive out of phase to be expedited). Heath and Ciarallo (2010) presented an initial agent-based modelling (ABM) simulation of individual order pickers and their interactions to better understand the drivers affecting warehouse cost and operating efficiency. Their simulation demonstrated the ability of the ABM paradigm to be utilized in the development, testing and evaluation of new warehouse operating and design strategies at a level of detail and aggregation. Mula et al. (2013) proposed a simulation approach based on system dynamics for operational procurement and transport planning. Tseng, Gung and Huang (2013) focused on the application of the make-to-plan (MTP) supply chain strategy and agent technology (AT) based technique. The researchers defined the impacts of operator parameters (e.g. throughput improvement, forecast accuracy improvement, demand variability management and safety stock level adjustment) on total cost, penalty cost, fill rate and on-time delivery.

2.5.4 System Dynamics (SD) Simulation

SD is a continuous simulation approach which allows the quantities of interest or variables to change over time. This approach is concerned with overall (aggregate and trend) system behaviour under the influence of given policies (Abd El-Aal et al., 2008). Sabry and Beamon (2000) developed an integrated multi-objective SC model for use in simultaneous strategic and operational SC planning. They added decision analysis to the model to allow the use of a performance measurement system that covers cost, customer service levels (fill rates) and flexibility (volume or delivery). The model incorporates production, delivery and demand uncertainty, and provides a multi-objective performance vector for the entire SC network. Wilson (2004) applied SD simulation in order to investigate how transportation disruption between 2 echelons in a 5-echelon SC affects performance in both traditional SCs and vendor-managed inventory (VMI) systems.

The previous sections show that while simulation has been put to a variety of applications, it has only rarely been deployed in strategic supply chain management. It is therefore the aim of this study to use the technique at the
strategic level for the purpose of supply chain design. To do this, the study employs the Supply Chain Guru software.

The fundamental objective of supply chain strategy is to ensure smooth flow at minimum cost. Since the measurement of SC performance is central to achieving this objective, this is the focus of the next section.

2.6 Supply Chain Performance

The identification of appropriate performance metrics is crucial for monitoring and improving supply chain performance. These metrics play an important role in setting objectives and determining future trends. Attempts have been made to survey the main performance metrics currently used in SCM (see Elrod, Murray and Bande, 2013; Gopal and Thakkar, 2012), and a number of authors have called for new measures to be introduced in response to the evolving business environment. Akyuz and Erkan (2010), for example, suggested that new performance measurement systems are needed to take account of qualities such as agility, flexibility, information productivity, business excellence and collaborative/partnership capacity. Kim, Kumar and Kumar (2010) developed a framework for assessing the comprehensive performance of supply chain partnerships (SCP). Their framework is based on the self-assessment dimensions and approaches of the business excellence model developed by the European Foundation for Quality Management (EFQM).

Drawing on his review of the literature, Leończuk (2016) compiled a list of the various indicators that have been proposed for measuring SC performance (see Table 2.1).

As outlined in Chapter 1, this research aims to focus equally on the dimensions of efficiency, responsiveness and sustainability. The following sections therefore discuss these three dimensions and their metrics in more detail.

2.6.1 Supply chain efficiency

The measurement of SC efficiency is vital, not just to give an insight into how the chain is performing but also to identify any problems in a timely fashion. Lichocik and Sadowsk (2013) attempted to explain the problem of supply chain management efficiency in the context of general theoretical considerations relating to supply chain management. The authors highlight the determinants and practical implications of supply chain management efficiency, concluding that efficiency means being cost-effective and streamlining processes while ensuring that service remains high quality. Mishra (2012) employed data envelopment analysis to measure SC efficiency in Indian pharmaceutical companies, using the constant returns to scale (CRS) assumption and variable returns to scale (VRS) assumption to calculate a technical efficiency score.

Danese and Romano (2011) analysed the impact of customer integration on efficiency and the moderating role of supplier integration by employing hierarchical regression analysis to test two hypotheses. The integration includes upstream and downstream operations in both suppliers and customer's sites. Their analysis revealed that supplier integration positively moderates the relationship between customer integration and efficiency, but did not support the hypothesis that in general, customer integration has a positive impact on efficiency. Where supplier integration is low, customer integration can even reduce efficiency.

Perspective of Performance Indicators	Subdivision 1	Subdivision 2	Subdivision 3
Qualitative / quantitative	Qualitative	Customer satisfaction, flexibility, information and material flow integration, effective risk management, supplier performance	
		Associated with the cost	Cost, sales, profit, inventory, investment maximization
	Quantitative	Associated with the customer	Product lateness, fill rate, customer response time, lead time
		Related to productivity	Capacity utilization, resource utilization
Based on SCOR model	Related to process	Planning, sourcing, manufacturing, delivery and returns	
	Related to performance attributes	Reliability, responsiveness flexibility, cost and asset management efficiency	
Performance measure type	Resources	Goal: high level of efficiency Purpose: impact on profitability	
	Output	Goal: high level of customer service	
		transition of customers to other supply chains	
	Flexibility	Ability to respond to a changing environment	
		changes	

Table2. 1: Categories and sub-categories of performance indicators

Perspective of Performance Indicators	Subdivision 1	Subdivision 2	Subdivision 3
Level of the decision-making	Strategic		
	Tactical		
proceed	Operational		
		Sales	
	Economic	Waste costs	
		Resource efficiency	
Implementation Extent	Environmental	Compliance with environmental standards	
		Consumption of hazardous/toxic materials	
		Energy consumption	
	Social	Product image	
		Customer loyalty	
		Relationship with surroundings	
	Operational	Operating cost	
		Response time	
		Inventory turnover rate	
		Order fulfilment	

Cont. Table2. 2: Categories and sub-categories of performance indicators

Source: Leończuk (2016)

2.6.2 Supply chain responsiveness

Modern supply chains must be able to respond rapidly, effectively and efficiently to changes in the marketplace if they are to endure and create competitive advantage (Adebambo and Adebayo, 2013). The relationship between responsiveness and competitive advantage is illustrated by Sukati et al. (2012), who found that supply chain integration positively impacts on the responsiveness and competitive advantage of the chain as a whole, and that supply chain responsiveness is positively associated with competitive advantage at firm level. Ghosh, Das and Deshpande (2014) offered an integrative framework that incorporates chain responsiveness, process integration, supply chain coordination and performance, but acknowledge that more research is needed to understand and explore the quantitative relationships between these constructs. Danes, Romano and Formentini (2012) argued that in supply networks, both external and internal integration practices have a significant and positive impact on responsiveness. However, since external integration has a bigger impact on company responsiveness than internal integration, they advised managers to adjust the level of adoption of according to the degree of integration practices supplier network internationalization.

Yi, Ngai and Moon (2011) asserted that supply chain responsiveness is best raised by reducing uncertainties and improving supply chain flexibility. The authors identify four types of flexibility strategy (laggard, conservative, agile and aggressive) that are adopted by SC participants in response to environmental uncertainties, and proposed a theoretical framework to assist managers in properly diagnosing and deploying these strategies. Singh and Sharma (2013), meanwhile, employed the analytical network process approach to decide where companies' priorities should lie in terms of flexibility. They concluded that organizations should give top priority to manufacturing, followed by customers and suppliers.

In their research model, Roh, Hong and Min (2013) set out the drivers, strategy, practices and performance outcomes associated with SC responsiveness. They suggested that the level of SC responsiveness is mainly influenced by firm size, industry characteristics and the customer and supplier bases rather than the location of manufacturing firms. The study showed that implementing a responsive supply chain strategy involves the integration of inter organization and sources (i.e. socio-relational and techno process integration) across the global supply chain to enhance pull production capabilities.

26

2.6.3 Supply chain sustainability

Environmental concerns such as climate change, environmental contamination and resource depletion are having an increasing impact on the activities of supply chains. The UN defines sustainability as "meeting the needs of the present without compromising the ability of future generations to meet their own needs" (Cabral et al.,2011). As Baud-lavigne et al. (2012) explained, there was a growing focus on the ways in which human and economic activities have the potential to adversely impact the long-term sustainability of the planet.

Taticchi, Tonelli and Pasqualino (2013) aim to developed the body of knowledge in the area of sustainable supply chains by conducting a critical review of the literature addressing sustainable supply chain performance measurement (SSCPM). These authors analysed the evolution of the research field, revealing that it is immature but growing very quickly. Schaltegger and Burritt (2014) were among those contributing to this growth with their analytical framework for the assessment of approaches to the measurement and management of sustainability performance in supply chains (SPSCs). They outlined five SPSC designs which may be used individually or in combination, but acknowledge that some measures need systematic development (they explained that eliminating or replacing existing products and SC participants may create fundamental changes that cannot be captured with simple SPSC measures but may require a different set of indicators). Finally, Zailani et al. (2012), investigated the extent and impact on performance of environmental purchasing and sustainable packaging, found that the former has a positive effect on economic, social and operational outcomes, while the latter has a positive effect on environmental, economic and social outcomes.

2.6.4 Performance measures of efficiency, responsiveness and sustainability

As is evident from the previous sections, the researcher can choose from a wide range of performance metrics. Table 2.3 lists those metrics that are most relevant to the dimensions addressed in the current research, proposed framework; SC performance is assessed using the efficiency, responsiveness and sustainability indicators, they are defined as following:

- Efficiency is measured using the profit to revenue ratio indicator (calculated by dividing cross profit by revenue).
- Responsiveness is measured using the average fill quantity rate (calculated as the ratio of demanded items filled from available inventory to the total number of items demanded over a particular period).
- Sustainability, although sustainability is a broad concept that encompasses economic, social and environmental concerns, the proposed framework focuses only on the environmental aspect, measuring it in terms of CO2 emissions.

Table 2-3	Attributes and	mansuras of	officioncy	racnoncivanasc	and sustainability
Table 2. 5	Attributes and	measures or	eniciency,	responsiveness	and sustainability

Source	Performance	Measures	
	Attribute	Dorfoot ordor fulfilm ont	
	Reliability		
	Responsiveness		
		Upside supply chain flexibility	
	Agility	Upside supply chain adaptability	
SCOR		Downside supply chain adaptability	
(version 11)		Overall value at risk (VAR)	
	Cost	Total cost to serve	
	Asset	Cash-to-cash cycle time	
	Management	Return on supply chain fixed assets	
	Efficiency	Return on working capital	
		Perfect order fulfilment	
Behrouzi,		from suppliers	
Wong and	Delivery and reliability	On-time production	
Behrouzi		On-time delivery to customers	
(2011)		Perfect order fulfilment to customers	
		Customer delivery lead time	
α		Fill rates	
Ambe	Flexibility	Volume flexibility	
(2014)		Product-mix flexibility	
(2011)		Delivery flexibility	
	Sustainability	Proportion of reusable/recyclable	
Rao (2014)		materials to total material input	
		Raw material efficiency	
		Proportion of cost of energy in production	
		to total value of output	
		Volume of air emissions per year (NOx,	
		SOx, CO2, VOC, etc.)	
		Use of vehicles that run on renewable	
		energy, electricity and natural gas	

2.7 Conclusions

The chapter demonstrates that while numerous researchers have discussed SCD, there is still much to do in this area, this is summarized as following:

- Although researchers have addressed various aspects of SCD, including strategic and operational SCD, SCN design, and designing for performance improvement, there is still no consensus on how the concept of SCD should even be defined.
- A number of frameworks have considered some of these functions, no single framework has yet been presented that ties all of these functions together (Melnyk, 2014).

This research aims to address this gap by developing an integrated SCD framework that considers strategy, process, network and performance concurrently.

- 3. Although the aim of SC strategy is generally to improve efficiency and customer service across the chain while keeping negative environmental impacts to a minimum, no framework has yet been developed that captures all of these performance dimensions. Accordingly, the proposed framework seeks to improve SC performance in terms of efficiency, responsiveness and sustainability.
- 4. The review indicates that despite simulation's importance as a tool for evaluating and improving SC outcomes, it has rarely been exploited in the field of strategic SC management. This research uses a simulation tool to build and simulate different supply chain scenarios.

Chapter 3 sets out the methodology that was employed to develop the framework, while Chapter 5 presents the results of the simulation that was run to validate it.

Chapter 3: Research Methodology

3.1 Introduction

The research methodology comprises the sequence of steps that are taken by the researcher to answer the research problem (Kothari, 2004). In this case, the research was conducted in two main stages. The first of these was the development of an integrated framework for supply chain design. The framework includes proposed models for supply chain strategy, processes, resources and performance. The strategic model studies consistency between customer needs and supply chain capabilities; the process model is based on the SCOR model; the resource model investigates the elements that make up the supply chain network; and the performance model offers sets of measurements for efficiency, responsiveness and sustainability. The second stage using hypothetical supply chain network to validate the proposed framework. The modelling employed statistical tools, probability distributions (to model the uncertain variables) and sensitivity analysis (to vary variables for the purpose of building different scenarios).

3.2 Development of Supply Chain Management Frameworks

In the following, key frameworks in Supply Chain management are discussed.

Although the term is frequently used in the SCM literature, there seems to be a lack of consensus about what a framework actually is. As Chapter 4 presents a proposed framework, it is essential to discuss the general concept and how they are developed.

A framework is a set of basic assumptions or fundamental principles of intellectual origin in which discussions and actions can proceed (Popper, 1994). Very often, the terms model and framework are used interchangeably, but for the purpose of this research, the two are regarded as distinct concepts. The framework is made up of four models; these answer "what is" questions, while the overall framework answers "how to" questions. Soni and Kodali (2013) suggest that a framework should: depict the complete structure of relationships between elements of the system under study (not just identify the elements that

make up the system); describe the steps/stages/sequence of activities that need to be undertaken to achieve the designated purpose; and describe the activities connecting the various elements within the framework.

3.2.1 Characteristics of SCM framework in general

Soni and Kodali (2013) reviewed a number of SCM framework articles and propose a framework that possibly suggests a way to achieve coherency in use of SCM frameworks. They noticed a massive use of sets of elements (or constructs) in SCM frameworks and tried to find out a possible set of standard constructs that make SCM by the aid of SCM professionals, the efforts were directed towards finding out the broad area, a particular construct may belong. This broad area is referred as a pillar of SCM and that leads to emergence of a comprehensive SCM framework (see Figure 3.1).

At the top of the framework is the mission and vision of the company. This informs its competitive strategy, whether this is based on cost structure or product differentiation. Once the competitive strategy and its priorities have been established, the company then formulates a supply chain strategy that will promote supply chain efficiency and effectiveness.



Figure 3. 1: Characteristics of SCM framework in general

Source: Soni and Kodali (2013)

Once the supply chain strategy is in place, the SCM pillars are used to build the capabilities of the chain and help the organization achieve its mission and vision. However, this will only happen if a strategic fit is achieved between its competitive strategy and supply chain strategy.

3.2.2 Method of developing a SCM Strategy

The supply chain strategy is part of the company's overall business strategy, but unlike most company strategies, it requires the coordination and commitment of many different firms (Ambe and Badenhorst-Weiss 2011), they suggested that developing a strategic SCM framework involves three steps: understanding the market and customer demand; defining the company's core competencies; and choosing the most appropriate strategy (see Figure 3.2).



Figure 3. 2: Method of developing SCM Strategy

Source: Ambe and Badenhorst-Weiss (2011)

Step 1: Understand the market and the nature of customer demand: Six key market variables determine the attributes of a supply chain structure: volume, time, variety, service level required, price and rate of change, innovation and new product development. If it to choose the right type of supply chain strategy, an organization must understand both its customers and supply chain uncertainty. Supply chain uncertainty is strongly affected by product life cycle; new products have higher supply uncertainty because design and production processes are still evolving, whereas mature products tend to have less supply uncertainty. Different market requirements demand different kinds of supply chain. Choosing the wrong strategy for a product may lead to mismatch in the supply chain. Mismatch is the root cause of supply chain problems.

Step 2: Define core competencies and capabilities of a supply chain: Supply chains have different characteristics, but all supply chains have two important attributes: cost and service. Supply chain capabilities include the ability to respond to a wide range of demanded quantities, meet short lead times, handle a large variety of products, build highly innovative products, meet a high service level and handle supply uncertainty. Where products are predictable, the ability to produce these products at low cost becomes the dominant consideration. The capabilities of a supply chain are determined by the trade-off, its participants are prepared to make, between responsiveness and cost. The so-called efficient frontier marked the lowest possible cost that can be achieved for a given level of responsiveness.

Step 3: Choose the applicable strategy: The level of responsiveness that can be achieved in the supply chain depends upon the level of cost incurred; raising costs lowers efficiency but increases responsiveness. To achieve complete strategic fit, an organization must ensure that all its functions maintain consistent strategies that support the competitive strategy. All sub-strategies within the supply chain, such as manufacturing, inventory and purchasing, need to be consistent with the supply chain level of responsiveness to reduce uncertainties and cost while satisfying the end customer's needs.

A supply chain can be lean (efficient), agile (responsive) or a combination of the two. An organization can achieve a competitive advantage by strategically employing a leagile supply chain model that minimizes cost and maintains stability while still being flexible and responsive to customer demand. This model will allow the organization to compete on innovation, cost, service and quality.

3.3 The proposed conceptual SC framework

The literature review revealed that there is no standard methodology for designing supply chains; some researchers discuss SCD in terms of strategic objectives (e.g. how to design supply chains to be lean, agile or sustainable), others focus on SC network design (i.e. the number, location, capacity and operation of different supply chain nodes) and still others argue that SCD involves identifying product design criteria and evaluating their impact on SC configuration.

It is the position of this research to take into account all of these perspectives in the design of the proposed framework, as the various functions they describe are mutually complementary; the strategic model assists the chain in achieving efficiency and responsiveness, process model employs SCOR model that determines process configurations, the network model determines the resources that are required to deliver the defined strategy, while the performance model shows whether the SC is achieving its objectives.

The aim is to develop an integrated framework that incorporates the various kinds of strategy SCs can use to achieve their goals along with the process configurations and networks they can employ to implement these strategies. The theoretical fundamentals of the proposed framework are shown in Figure 3.3 and discussed in the following sections.



Figure 3. 3: The components of the proposed SC framework

The strategic fit concept is employed to explain the strategic role of the framework, the SCOR model to clarify supply chain configurations and performance, and the supply chain network design concept to illustrate how supply chain networks are structured.

3.3.1 Supply chain strategy: achieving strategic fit

Supply chain management requires the strategic management of the various aspects of the coordination process, including information, technology, distribution, products, raw materials, finance and, most of all, relationships. Successful companies understand that they cannot compete effectively in all dimensions but know where to focus their energies.

The survival of the supply chain depends on the consistency between customer expectations (what customers want) and SC performance (what the chain is able to deliver). This is the concept of strategic fit: the company must ensure that its supply chain capabilities enable it to meet the needs of its customers. Evidently, the company must have a clear understanding of what these needs and capabilities are (Chopra and Meindl, 2007).Strategic fit requires the competitive and supply chain strategies of the company to have aligned goals. There are three steps to achieving strategic fit (Chopra and Meindl, 2007):

3.3.1.1 Understanding customer needs

Customer demand can vary in a number of ways: for example, lot quantity may vary from small (e.g. customised or emergency orders) to large planned orders, while response time (the amount of time that customers are willing to wait for orders) may be longer for customized products. A company may have to hold a wide range of products to appeal to different customer segments, particularly if the business environment is unstable. A high level of product availability usually requires high inventory levels and more detailed and frequent information sharing, reducing competitive advantage. On the other hand, customers who expect a high level of service, more product variety and short response times tend to be less sensitive to product price. Customer demands regarding product innovation tend to vary according to product purpose, with less being expected of functional products than of consumed products. All of these attributes can be combined in one key metric: implied demand uncertainty. Unlike demand uncertainty, which reflects the uncertainty of customer demand for a product, implied demand uncertainty describes the uncertainty only for that portion of demand that the supply chain plans to satisfy based on the attributes the costumer desires.

3.3.1.2 Understanding the supply chain's capabilities

Creating strategic fit is about finding the supply chain strategy that best meets the demand a company has targeted, given the uncertainty it faces. If it is to find the balance between responsiveness and efficiency that best supports its competitive strategy, the company must have a clear understanding of the logistics and cross functional drivers that affect SC capability. These are:

- Facilities: where the product is stored or fabricated. Decisions regarding the role, location, capacity and flexibility of facilities have significant impact on the supply chain's performance.
- Inventory: changing inventory policies can dramatically alter the supply chain's efficiency and responsiveness. High inventory levels, for example, can increase a company's responsiveness and raise service levels but may reduce its efficiency.
- Transportation: the SC may employ multiple combinations of modes and routes, each with its own performance characteristics. This has a direct impact on SC efficiency and responsiveness; faster transport modes, for example, may make the chain more responsive, but as they tend to be more expensive they are also less efficient.
- Information: managers must use the available data and analysis concerning facilities, inventory, transportation, costs, prices and customers to make the supply chain more efficient and responsive. For example, using information to better match supply and demand will improve responsiveness while keeping production and distribution costs down.
- Sourcing: the choice of who will perform a particular supply chain activity such as production, storage or transportation can impact on both responsiveness and efficiency. Opting to source some products from a far distant supplier because this is cheaper may improve a company's efficiency but it will also compromise

its responsiveness. Going back to the supplier with small or urgent orders is likely to increase transportation costs.

 Pricing: pricing affects the behaviour of the buyer of the goods or services, thus affecting supply chain performance. For example, if a haulage company varies its charges based on the lead time demanded by the customer, it is likely that customers who value efficiency will order early and customers who value responsiveness will be willing to wait and order just before they need the product to be transported.

3.3.1.3 Achieving strategic fit

The goal of strategic fit is to target high responsiveness for a supply chain facing high implied uncertainty, and efficiency for a supply chain facing low implied uncertainty. An increase in implied uncertainty from customers and supply sources is best dealt with by improving the responsiveness of the supply chain (see Figure 3.4). To achieve a high level of performance, companies should aim to move their competitive strategy and supply chain strategy towards the zone of strategic fit.



Figure 3. 4: Finding the zone of strategic fit

Source: Chopra and Meindl (2007)

Strategic fit is the optimum combination of efficiency and responsiveness; achieving it requires companies to have a clear understanding of their customers' needs (in terms of both demand characteristics and certainty) and

the capabilities of the SC. Designing the right combination of logistics drivers is vital for achieving responsiveness and efficiency, first in the company and then across the supply chain as a whole. The company must ensure that all its functions are implementing consistent strategies, and that these support the company's competitive strategy. Figure 3.5 presents the process by which strategic fit is achieved.



Figure 3. 5: Process of achieving strategic fit

The strategy of any single supply chain member is closely connected with those of the chain's other members, both upstream and downstream. A given level of responsiveness can be achieved within the chain by adjusting the respective roles played by each stage; for example, allowing one stage to absorb most of the uncertainty will make it more responsive, while at the same time allowing upstream and downstream stages to become more efficient. Table 3.1 presents multiple options for designing supply chain drivers; different designs can be directed to achieve specific strategies and SC outcomes.

Design of SC drivers	Efficient / Responsive Strategy for supplier, manufacturer, distributor and retailer	SC measurements	SC aim
Facilities: • Single / multiple facility location (plant, warehouse, retailer). • Flexible / inflexible process. • Product / function-focused process. • Low / high investment in facilities. Inventory:	Efficient Strategy Low costs Limited items 		
 Finished products, parts or raw materials. Transportation: Slow cheap / fast expensive transportation modes. Low-cost full truckload / higher-cost less than full truckload quick shipments. Fixed / flexible numbers and types of trucks 	 Varying supply time Fixed batch size Responsive Strategy High costs 	 KPIs of whole SC KPIs of SC members 	Desired outcome
 Information: Pull process (rely on information) / push process. Low / high level of information sharing. Sourcing: In house / out sourcing. Pricing: Low steady / high changeable price 	 Quick response Various items Short and fixed delivery time Varying batch size 		

Table 3. 1: Supply chain strategies and designs

Based on: Chopra and Meindl (2007)

3.3.2 Supply chain operations reference (SCOR) model

The supply chain operations reference model (SCOR) is the product of the Supply Chain Council (SCC). The model provides a unique framework that links business processes, metrics, best practices and technology into a unified structure to support communication among supply chain partners and to improve the effectiveness of supply chain management and related supply chain improvement activities. The SCOR model consists of four major sections (SCC, 2012): performance (standard metrics to describe process performance and define strategic goals), processes (standard descriptions of management processes and process relationships), practices (management practices that produce significantly better process performance) and people (standard definitions for the skills required to perform supply chain processes). These four sections are discussed below.

3.3.2.1 Performance

The performance section of SCOR consists of two types of elements: performance attributes and metrics. A performance attribute is a grouping of metrics used to express a strategy. An attribute itself cannot be measured; it is used to set strategic direction. Examples of business strategies applied to supply chains include superior performance for supply chain reliability and advanced performance for agility. Metrics measure the ability of a supply chain to achieve these strategic attributes. Superior performance for reliability can be expressed in the performance objective: perfect order fulfilment: X%. Reliability is the performance attribute and perfect order fulfilment is the metric. Benchmarking is a commonly used method to calculate the value of X in the reliability example. Table 3.2 shows the performance attributes and metrics used within the SCOR model.

Performance Attribute	Definition	Level-1 Strategic Metric
Reliability	The ability to perform tasks as expected. Reliability focuses on the predictability of the outcome of a process. Typical metrics for the reliability attribute include: on-time, the right quantity, the right quality.	Perfect order fulfilment
Responsiveness	The speed at which tasks are performed. The speed at which a supply chain provides products to the customer. Examples include cycle-time metrics.	• Order fulfilment cycle time
Agility	The ability to respond to external influences, the ability to respond to market place changes to gain or maintain competitive advantage. SCOR agility metrics include flexibility and adaptability.	 Upside supply chain flexibility Upside supply chain adaptability Downside supply chain adaptability Overall value at risk
Costs	The cost of operating the supply chain processes. This includes labour costs, material costs, management and transportation costs. A typical cost metric is cost of goods sold.	• Total cost to serve
Asset management efficiency (assets)	The ability to efficiently utilize assets. Asset management strategies in a supply chain include inventory reduction and in-sourcing vs. outsourcing. Metrics include inventory days of supply and capacity utilization.	 Cash-to-cash cycle time Return on supply chain fixed assets Return on working capital

Table 3. 2: Performance attributes and metrics in SCOR

3.3.2.2 Practices

The practices section, formerly known as best practices, provides a collection of industry-neutral practices companies have recognized for their

value. A practice is a unique way to configure a process or a set of processes. The uniqueness can be related to the automation of the process, a technology applied in the process, special skills applied to the process, a unique sequence for performing the process, or a unique method for distributing and connecting processes between organizations. SCOR recognizes that several different practices may exist within any organization. These practices may be classified as emerging practices, best practices, standard practices, and declining practices.

3.3.2.3 Process

The model is organized around the five primary management processes of plan, source, make, deliver and return. Planning processes balance aggregate demand and supply to develop the course of action which best meets sourcing, production and delivery requirements, while source processes are concerned with the procurement of goods and services to meet planned or actual demand. Make processes transform products into their finished state to meet planned or actual demand, while deliver processes provide the finished goods and services to meet planned or actual demand. This typically involves the management of orders, transportation and distribution. Finally, return processes are concerned with the returning of (or the receiving of returned) products for any reason. These processes extend into post-delivery customer support. Figure 3.6 shows the five SCOR processes distinguished by process type/ category.



Figure 3. 6: SCOR process type and category

Planning processes balance aggregated demand and supply and generally occur at regular intervals. Execution processes are triggered by planned or actual demand and involve changing the state of materials and goods. They generally involve scheduling/sequencing, transforming products and moving them onto the next process. Enabling processes focus on preparing, maintaining and managing the information and relationships on which the planning and execution processes rely.

Zhou et al. (2001) stated that the SCOR model includes four levels of process detail. Level 1, the top level (process type), defines the scope and content of the SCOR and is where the performance targets are set. Level 2 is the configuration level (process categories); this level defines the configuration of planning and execution processes in the material flow. Standard approaches include make-to-stock (MTS) (production is based on sales forecasts e.g. the fashion industry), make-to-order (MTO) (the customer defines the specs and the product is manufactured to order e.g. tailoring) and engineer-to-order (ETO) (the customer defines the specs and the factory buys materials, designs and manufactures the product e.g. manufacture of lifts).

Level 3 (the process element level) determines the company's ability to compete successfully in its chosen markets. It comprises: process element definitions, process element information inputs and outputs, process performance metrics, best practices (where applicable), the system capabilities required to support best practice, and systems/tools. Finally, level 4 (the implementation level) defines the specific practices the company needs to implement to achieve competitive advantage and to adapt to changing business conditions.

Figure 3.7 shows the hierarchical structure of the SCOR model with specific boundaries regarding of process scope.



Figure 3. 7: SCOR process hierarchy

3.3.2.4 People

The people section of the SCOR model, which was introduced in SCOR 10, provides a standard for describing the skills that are required to perform tasks and manage processes. These standard definitions focus on aptitude, experience, training and competency level. SCOR recognizes five commonly accepted competency levels: trainee (untrained beginner, no experience, requires and follows detailed written instructions), beginner (performs the work with limited situational perception), competent (understands the work and can determine prioritize based on situational aspects) and expert (intuitive understanding, able to apply experience patterns to new situations).

3.3.3 Supply chain network design

Network design involves determining the elements, numbers, locations and material flow quantities within the supply chain. The term "supply chain" implies that there is only one player at each stage of the chain, but in practice, manufacturers may source materials from a range of suppliers and work with several different distributors. In other words, most supply chains are more accurately described as networks (Chopra and Meindl, 2004). An SC network is made up of suppliers, manufacturers, distribution centres and retailers; it comprises a series of processes and stages, which starts with the material/information supplier and ends with the customer. Mid-stage participants play a dual role as the customer of the next stage and supplier of the previous stage.

Wang (2009) claims that SC network design is one of the company's biggest strategic tasks and central to the long-term efficiency of the whole SC. It involves working out the optimal number, capacity, layout and type of factories, warehouses and distribution centres required, setting up distribution channels and calculating the quantity of materials which will be consumed in the production process, the quantity of materials which will be transported from suppliers to customers, and the quantity of materials which will be produced. Figure 3.8 shows the different stages that make up a typical supply chain.



Figure 3. 8: Multi-echelon supply chain network

Long (2014) describes supply chain networks as consisting of several entities ranged from upstream to downstream. Each entity is composed of several departments or workshops, which are in turn made up of several production units. SC networks may therefore be divided into four levels: the supply chain network level, the enterprise level, the workshop level and the production unit level. Long argued that any entity in any level can be modelled using the SCOR model's five core processes to obtain its required function by selecting different process elements and determining different parameters. He proposed a hierarchical framework with four levels, each of which consists of several elements. Elements in the upper level can be decomposed into a set of elements in the lower level, while elements in the lower level can be aggregated to form an element in the upper level. The integrated supply chain design framework proposed in this study draws on Long's framework (see Figure 4.1).

3.4 Research Techniques and Tools

3.4.1 Conceptual techniques

A conceptual framework may be defined as a network of interlinked concepts that together provide a comprehensive understanding of a phenomenon. Jabareen (2009) argued that rather than offering a theoretical explanation, as do quantitative models, conceptual frameworks provide full understanding of all the concepts proposed and the interrelationship amongst them. Conceptual frameworks may be developed and constructed through a process of qualitative analysis. This study's conceptual framework for improving SC management, presented in Chapter 4, addresses a gap in the literature by integrating the strategy, process, network and performance functions.

3.4.2 Supply chain modelling

When a company is designing a new logistics network, it will take into account its objectives, all the decision variables such as network structure, facility location and service requirements, and supply chain constraints. Extraneous information should be omitted in order to limit the complexity of the model (Min and Zhou, 2002). No model can capture all aspects of supply chain processes, so the model should be defined in such way that it answers the question it is designed for. In this study, modelling was used to create a hypothetical supply chain in the virtual environment. There are three stages to the modelling process (Fitkov-Norris, 2010). These are discussed below.

3.4.2.1 Model identification

The first step is to identify the model objectives and the best approach for modelling a particular event. This stage also includes defining the model boundaries (i.e. the key variables, scope and time frame). Min and Zhou (2002) explained that the main objects likely to feature in a SC model are customer service (may be represented by product availability and response time), monetary value (generally defined as a ratio of revenue to total cost and measured in terms of asset utilization, ROI or cost), information transactions (the sharing of information in real time allows supply chain partners to coordinate their actions and integrate SC processes) and risk (SC integration helps mitigate risks such as risk of quality failure or risk of information failure).

3.4.2.2 Building the model

The second step is the building of the model. This involves representing the real world links between the variables of interest in an appropriate format. This can be done using a quantitative approach such as linear programming or a qualitative approach such as a structural dependency representation using causal diagrams. This stage comprises the identification of supply chain decision variables, the collection of supporting data, and the identification of supply chain constraints.

Supply chain decision variables: Since decision variables generally set the limits on the range of decision outcomes, they are functionally related to supply chain performance. Thus, the performance measures (or objectives) of a supply chain are generally expressed as functions of one or more

decision variables. Decision variables might include (but are not limited to) (Min and Zhou, 2002):

- Location: determining where plants, warehouses (or distribution centres (DCs)), consolidation points and sources of supply should be located.
- Allocation: determining which warehouses (or DCs), plants and consolidation points should serve which customers.
- Network structuring: centralizing or decentralizing the distribution network and determining which combination of suppliers, plants, warehouses and consolidation points should be utilized.
- Policies: determining how the supply chain will achieve its objects. Policies include sourcing policies (e.g. make or source), transportation policies (e.g. full/less than full truckload) and inventory policies (e.g. periodic review-based or level of units maintained).
- Number of facilities and equipment: determining how many plants, warehouses and consolidation points are needed to meet the needs of customers and market segments.
- Number of stages (echelons): determining the number of stages that will comprise a supply chain.
- Service sequence: determining delivery or pickup routes and schedules for vehicles serving customers or suppliers.
- Volume: setting the optimal purchasing volume, production and shipping volume at each node of the supply chain.
- Inventory level: determining the optimal amount of inventory to be stored at each supply chain stage.
- Size of workforce: determining the number of truck drivers or order pickers needed for the system.
- The extent of outsourcing: determining which and how many suppliers should be used for long-term outsourcing contacts.

Collection of supporting data: If the company is to make the correct decisions concerning decision variables, it must collect all the relevant data about product demand, customer value, transportation costs, transportation

times, warehousing costs, inventory costs, production costs and procurement costs etc.

Identifying supply chain constraints: Min and Zhou (2002) defined supply chain constraints as limitations that restrict the range of decision alternatives open to the firm. These constraints may affect the feasibility of some decision alternatives. They may include the company's financial, production, supply or technical capacity (or those of another SC member), service compliance issues (e.g. delivery time windows, manufacturing due dates and the number of driving hours permitted for truck drivers) and the extent of demand (the company may have to balance its demand against supply capacity at the preceding stage).

3.4.2.3 Model analysis and interpretation

The third step in the modelling process involves the derivation of solution(s) for the mathematical equations and/or simulation of the dependencies between variables, in order to answer the particular questions set out at the beginning of the process. This step may also involve a number of extra steps such as model validation. In this study, supply chain optimization was employed to identify the best solution for the model, which was then evaluated by means of simulation.

3.4.3 Supply chain simulation

Whereas an optimized solution is only valid for a defined scenario, a simulation model can treat different scenarios in order to find an optimal solution. A solution derived from a simulation can therefore be made more sensitive to environmental changes than a solution obtained through optimization. Simulation can be defined as the process of creating a model of an existing or a proposed system (e.g. a project, a business, a forest) in order to identify and understand those factors which control the system and/or to predict its future behaviour (EI-Aal et al., 2008). In this research, the SCG software package was used as a simulation tool to investigate the impact of different SCD scenarios on efficiency, responsiveness and sustainability performance.

50

3.4.4 Supply chain evaluation and what-if analysis

Optimization is: "Narrowing your choices to the very best when there are virtually innumerable feasible options and comparing them is difficult" (Institute of Operations Management and Management Science). An important component of SCD is determining how to achieve an effective design, given a performance measure or a set of performance measures (Beamon, 1998). Optimization models answer questions about plant location, product mix, choice of technology, distribution methods, inventory planning and control, choice of suppliers, configuration and reverse logistics (EI-Aal et al., 2008). In this study, the focus was on finding optimum/near optimum values for the inventory and transportation variables in the studied supply chain.



Figure 3. 9: Supply chain modelling techniques

3.4.5 Statistical distribution

The type of statistical distribution, together with the distribution parameters (mean, standard deviation, minimum and maximum values), define a probability density function (PDF) for a random variable. A PDF describes the distribution of possible values that a random variable may assume, for a hypothetical, infinite set of observations of the variable. Usually, the features of the population under investigation can be summarized by the parameters. Hence, the research problem usually becomes an investigation of the values of parameters. Since these population parameters are unknown, sample statistics are used to make inferences about them.

In this research, statistical distributions were used to model different demand patterns. Syntetos et al. (2005) characterized demand as intermittent, erratic or lumpy. A demand is intermittent when it appears randomly with many time periods having no demand; an erratic demand pattern is characterized by highly variable demand size; and lumpy demand is both intermittent and erratic. The authors quantify the categories using two parameters: average inter-demand interval (ADI) and squared coefficient of variation (CV^2).The cut-off values are set as ADI = 1.32 and CV^2 = 0.49, as shown in Figure 3.10.



Figure 3. 10: Categorization of demand patterns

Source: Syntetos et al. (2005)

3.4.5.1 Normal distribution

Normal distribution is the most common type of probability density function (PDF). For a normal distribution, about 68% of observations should fall within one standard deviation of the mean, and about 95% of observations should fall within two standard deviations of the mean. Normal probability density function, demonstrating standard deviation ranges, is shown in Figure 3.11 (Evans, 2000).



Figure 3. 11: Normal probability density function

Source: Forbes (2011)

Parameters: The mean (μ) is specified as a real number and the standard deviation (σ) is specified as a positive number.

Range: (-∞, +∞)

Mean: µ

Variance: σ^2

3.4.5.2 Bounded normal distribution

A truncated normal distribution can be defined by setting the desired minimum and/or maximum values for the variable. For practical purposes, if the minimum and maximum values are at least three standard deviations away from the mean, a complete normal distribution will be obtained. If the minimum/maximum values are less than three standard deviations away from the mean, the distribution will be visibly truncated (Duncan, 2000).

3.4.5.3 Lognormal distribution

If a random variable has a lognormal distribution, then its natural logarithm has a normal distribution. This is the meaning of the term lognormal. The lognormal distribution can only be used for variables which are always positive. A lognormal distribution can be useful for modelling variables such as cohesion, which may have a large peak in the distribution near zero and then narrow off gradually for larger values. Figure 3.12 illustrates lognormal probability density functions (Evans, 2011).



Figure 3. 12: Lognormal probability density functions

Source: Forbes (2011)

Parameters: The mean ($\mu l > 0$) is specified as a real number and the standard deviation ($\sigma l > 0$) is specified as a positive number.

Range: [0, +∞)

Mean: µl

Variance: $\sigma^2 l$

3.4.5.4 Poisson distribution

The Poisson distribution is applied when counting the number of rare but open-ended events. An example might be the number of faults in a batch of materials. It is also used to represent the number of arrivals, say, per hour, at a service centre. In practice, arrival rates may vary according to the time of day or year, but a Poisson model will be used for periods that are reasonably homogeneous. The mean and variance are equal and can be estimated by observing the characteristics of actual samples of "arrivals" or "faults" (Evans, 2011).



Figure 3. 13: Poisson distribution for λ = 1, 2, 3, 4, 5, 6, 7, 10, and 20.

Source: Hoffman (2015)

Parameters: The mean (λ) is specified as a positive real number.

Range: {0,1,...}

Mean: **λ**

Variance: **λ**

3.4.6 Sensitivity analysis

The parameter values and assumptions of any model are subject to change and error. Sensitivity analysis (SA), broadly defined, is the investigation of these potential changes and errors and their impacts on conclusions to be drawn from the model (Pannell, 1997). Sensitivity analysis has many uses, including decision making, improving understanding of a system, and model development. Pannell (1997) argues that the technique may be used to test the robustness of an optimal solution, identify sensitive or important variables, investigate sub-optimal solutions, assess the riskiness of a strategy or scenario, and understand relationships between input and output variables. In this research, it was used to understand the influence of specific factors on SC performance.

Sensitivity analysis is particularly valuable where parameters are uncertain, as it can highlight both the circumstances under which the optimal solution will change and how these circumstances will affect the optimal solution. However, the modeller needs to determine what changes to make in order to obtain the required information. These changes might include any or all of the following: the contribution of an activity to the objective, the objective itself (e.g. minimizing risk of failure instead of maximizing profit), constraint limits (e.g. the maximum availability of a resource), the number of constraints (e.g. adding or removing a constraint designed to express the personal preferences of the decision maker for or against a particular activity), and technical parameters (Pannell, 1997). Whichever items the modeller chooses to vary, many different aspects of the model output can be observed; for example, the value of the objective function and the values of decision variables.

Sensitivity analysis starts with a list of the key factors or parameters. If the aim is to estimate the likely profitability of a project, these factors might be market growth rate, market share, selling price and the costs of direct labour and direct materials. The most likely values are then attached to each of these parameters and used to predict the most likely level of profits. The effect is then calculated of varying the values of all or a selected few of these

parameters. This may be done by working out what the impact would be if all the values varied equally by, say, 1, 3 or 5 per cent. Different incidences of variation between the values may be calculated if appropriate. The outcomes of the alternative assumptions are listed and a subjective assessment made of their likelihood. Finally, the modeller draws conclusions regarding what if any actions are required to make the achievement of the better outcomes more likely.

3.5 Conclusions

This chapter discussed the methodology that was employed to develop the proposed framework. It presents the main SC components and characteristics, and the method for developing supply chain strategy, before describing the conceptual fundamentals of the proposed framework. These are the strategic fit concept (employed to fulfil the strategic role of the framework in achieving consistent strategy in terms of efficiency and responsiveness), the SCOR model (to clarify supply chain policies and process configurations) and the supply chain network design concept (to illustrate how supply chain networks are structured). Finally, the chapter discusses the techniques and tools that were utilised to refine and validate optimisation and framework: modelling, simulation. Probability the distributions were also utilised to model different demand patterns, and sensitivity analysis was used as a basis for building simulation scenarios.
Chapter 4: The proposed Integrated Supply Chain Design Framework

4.1 Introduction

As highlighted in the literature review (see section 2.7), previous studies have failed to identify essential linkages of SCD (i.e. strategy, process and network). Accordingly, this research proposes an integrated approach that combines all the most important elements within one framework. The proposed framework (shown in Figure 4.1) is composed of four different models, each of which performs a defined role. It takes into account the following considerations:

1. Resources: are all those physical investments that make up the SCN, including production entities, warehouses and distribution centres, as well as all means of transporting materials from suppliers to manufacturing centres and on to the end consumer. All decisions regarding resource location, capacity and technology must be directed towards achieving the strategic objective of the SC.

2. Processes: covers management policies for ensuring that resources are maximized to achieve the SC's strategic goal. The SCOR model is considered the standard template for SC processes (plan, source, make, deliver and return). The model defines the best process configurations for different levels of the SCN to support selected strategy in achieving the supply chain outcomes.

3. Relationships: covers the ways in which various entities within the SCN are linked and how their processes interact to achieve the goal of the SC. The proposed framework uses a network model to show how these relationships might be structured more effectively.



Figure 4. 1: The Proposed Integrated supply chain design framework

4.2 Strategic Objective Model

The strategic objective model outlines the ways in which different SC strategies attempt to meet customer requirements while still taking into account demand uncertainty and SC capabilities and performance. Several strategic models are offered in the literature, including those by Ivanov (2010), who proposed a model to support decision making on SC strategy, design, tactics and operations; Hwang (2010), who focused on the role of cost leadership, differentiation and focus in shaping overall strategic direction; and Sabet, Yazdani and Leeuw (2017), who considered the role of uncertainty and product importance in shaping SC strategy. However, as none of these models take into account the market environment, the proposed framework is instead based on Christopher's (2006) modeled, which recommends strategies for different demand characteristics.

4.2.1 Demand characteristics

Since customer demand is the main driver of strategic SC decision making, it is essential to understand the features of this demand and how they affect SC performance.

Uncertainty

Demand uncertainty focuses on the difficulty of predicting customer demand. Lee (2002) distinguished between functional products, which generally have long product life cycles and therefore stable and predictable demand, and innovative products, which tend to have a short life and therefore more unpredictable demand. Obviously, different supply strategies are required for the two categories of products. Functional products tend to offer less product variety than innovative products, which are often trend-led or produced to respond to customer demands for wider choice. Companies have three safety buffers for handling these uncertainties: safety inventory, safety capacity and safety time. These buffers are used to reduce variations in the SC and meet customer demand for better service at lower cost.

Lead Time

The order lead time limits the extent to which the supply chain can be orderdriven. If a very short order lead time is required, it may be necessary to make-to-stock (MTS) and provide local warehousing or vendor-managed inventory. In some cases, however, it is not possible to MTS because the product is customized or provided in such wide variety that finished stocks are not economically viable. In this case, the product is made-to-order (MTO) and the manufacturing process may require buffers in terms of excess manufacturing capacity and raw material stocks to support a short order lead time.

Reducing product development lead time means that a product can get to market earlier. This has a number of important advantages: the sales life of the product is extended; a higher price can be charged; new customers can be won; and a high market share can be won by building upon the initial lead. Moreover, by reducing overall lead time, product complexity and process setup times, the production of a particular product can be scheduled more frequently with smaller production batches. This improves the variety of products available to a customer over a given time (Kampen, and Donk, 2014).

Variety

Cooper and Griffiths (1994) stated that: "Issues of variety and complexity are strongly linked." An increase in external variety (i.e. in the choice being offered to the end customer) has implications for the level of internal variety that will be required of the SC (Kampen, and Donk, 2014).

Increasing variety makes logistics operations more complex and so increases both direct and indirect costs, though these may be mitigated to some degree by redesigning systems. Ideally, variety should be increased only when it adds value.

Variability

Where the demand for a product is stable and significant, SC members may be able to rely on a small supply base to provide a high volume of standard ship-to-stock components and materials (Kampen, and Donk, 2014). These high volumes can be leveraged to reduce ordering frequency, allowing a more efficient operation in which inventory turns are high and there is little exposure to excess and obsolete inventory. However, if customer demand is volatile, for example because the product is specialized, a low-volume approach is more sensible. SC members who are forced to rely on a wide range of suppliers, each producing unique components, are particularly exposed to the risk of excess or obsolete inventory.

4.2.2 Suggested strategies

Birhanu, Lanka and Rao (2014) argued that providing the right degree of responsiveness and efficiency simultaneously is difficult, since increased responsiveness is generally perceived to come at the expense of efficiency, and vice versa. The strategic objective model presents four SC strategies, allowing managers to choose the option that is best suited to the combination of supply/demand conditions they face.

Lean strategy (plan and execute)

This is the most appropriate strategy where demand is high-volume, lowvariety and predictable, and lead times are long. Materials, components and products can be ordered in advance and manufacturing and transportation facilities can be optimized (Christopher, 2006).

Lean strategy (continuous replenishment)

In cases where demand is predictable and replacement lead times are short, a lean strategy of continuous replenishment is possible. At its extreme, products are replaced as they are sold or used (Christopher, 2006). Christopher, Peck and Towill (2006) suggested that point-of-sale data facilitates this strategy as it allows vendors to manage their own inventory and rapidly replenish individual stores.

Agile strategy (quick response)

Christopher (2000) defined agility as the ability of an organization to respond rapidly to changes in demand both in terms of volume and variety. Where demand is unpredictable and lead times are short, the SC can adopt a quickresponse strategy such as MTO. Agile SCs must be capable of reading and responding to real demand, virtual which is information-based rather than inventory-based. Processes are integrated, with buyers and suppliers working collaboratively and products being developed jointly, and SC members sharing common systems and information. The agile SC is network-based; individual businesses are no longer competing as standalone entities but as part of a larger chain (Christopher, 2006).

Leagile strategy (postponement)

Leagile strategy is an option where lead times are long and demand is unpredictable, highly variable and outside the organization's control. A hybrid lean/agile strategy requires the SC to be "decoupled"; strategic inventory is held in some generic or unfinished form, with the final configuration being completed rapidly once the real demand is known. If the final physical configuration cannot be postponed in this way, it may be possible to postpone the distribution of the product instead by holding it in fewer locations and using express transportation to move it to the final market or point of use once the actual demand is known. The goal of the hybrid strategy should be to build an agile response upon a lean platform by following lean principles up to the decoupling point and agile practices after that point (Christopher, 2006).

4.3 Process Model

Davenport et al. (1995) defined a business process as a set of activities with specified business outcomes for customers. SC processes may thus be defined as the set of activities by which material is moved through the SCN. A variety of process models have been highlighted in the literature and are employed in industry, including SCOR (Supply Chain Operations Reference Model), GSCF (Global Supply Chain Framework), CPFR (Collaborative Planning, Forecasting and Replenishment), VRM (Value Reference Model) and SAP (System and Application Products). These models cover a range of areas such as organizational cooperation within the SC, product development, inventory management and manufacturing operations. SCOR

was selected to represent the process dimension within the proposed framework because it encompasses a wide range of policies that are suitable for use in different business environments and because it clarifies the relationship between SC processes and network structure.

4.3.1 Supply chain processes

Plan

Planning processes balance aggregate demand and supply to develop the sequence of actions which best meet sourcing, production and delivery requirements. An important part of SC planning is the running of full-stream supply/demand simulations. What-if analysis helps firms to prepare for various possible scenarios. Sharing of the resulting information is crucial to rebalance the chain and improve performance (Fawcett et al., 2011).

Source

Sourcing processes are critical because they connect manufacturers with suppliers. (Dong et al., 2001) showed that the benefit companies gain by giving the bulk of their business to a few suppliers and using long-term contracts outweighs the costs. Just-in-time (JIT) delivery from suppliers is also considered a good sourcing practice. The benefits of JIT delivery have been widely documented.

Make

The make process covers the transformation of raw materials into finished goods to meet SC demand in a timely manner. Relevant practices highlighted in the literature include JIT production, total quality management (TQM) and human resource management (HRM). JIT production practices include pull systems, cellular manufacturing, cycle time reduction, agile manufacturing strategy and bottleneck removal; TQM practices include statistical process control (SPC) and continuous improvement programmes; and HRM practices emphasize employee teamwork and workforce capabilities.

64

Deliver

Delivery processes are a critical part of SC management. Effective processes rely on SC partners sharing real-time information, which enhances visibility and improves order tracking. Agility is an important competence; other best delivery practices identified by the SCOR model include employing a single contact point for all order inquiries, order consolidation and the use of automatic identification.

4.3.2 Process configurations for lean strategy

The primary emphasis in lean strategy processes is on minimizing costs and maximizing production and logistics efficiency. Lean principles were originally aimed at improving manufacturing processes by eliminating waste but were extended to cover the development of a set of associated logistics tools. Lean SCs typically require close, collaborative relationships between manufacturers and suppliers because of the large-volume, long-term commitments involved.

Source-to-stock

The intention of source-to-stock is to maintain a pre-determined level of inventory for certain materials, sub-assemblies or products. The process involves ordering, receiving and transferring raw material items, sub-assemblies, products and/or services based on aggregated demand requirements.

Make-to-stock

The production processes for mature, highly standardized products focus primarily on achieving low-cost operations, which is typically accomplished with high-volume transformation. Production processes may be continuous if large numbers of similar products are required; alternatively, large batch processes allow some variety.

Deliver-to-stock

Manufacturers in these supply chains tend to push products onto retailers and to rely heavily on distribution centres and retailers to deliver products to consumers in the most cost-efficient manner. The multi-layered nature of lean supply chains makes efficient operations and information sharing challenging. Consequently, this type of supply chain is the most liable to the bullwhip effect.

4.3.3 Process configurations for agile strategy

An agile SC may have fewer opportunities to practise lean principles, but it should still look for efficiencies wherever possible. One way to achieve efficiency is by establishing collaborative relationships with key suppliers, as this not only ensures reasonably priced, high-quality raw materials but can also improve delivery times. However, these relationships should not be allowed to compromise the flexibility of the chain; it must be able to reduce the number of suppliers it deals with if it considers this to be necessary.

Source-to-order

Source-to-order inventory is ordered specifically for customer orders. To ensure satisfied customers, delays are minimized by purchasing raw materials in large quantities from multiple suppliers and maintaining a large inventory of work in progress units and other components. In the case of unique or very low-volume products, the engineer-to-order process might be utilized if raw materials need to be sourced specifically for the product.

Make-to-order

These products are usually customized from a combination of standardized components and additional elements that are specifically produced to meet individual customer requirements. In agile supply chains, the raw material and components may already be on hand (to ensure quick customization), but actual production does not begin until the customer's order is received. Job shop or project processes ensure the right balance between efficient and flexible production by allowing different designs to be produced from a small number of components.

Deliver-to-order

There may be no distributors or retailers in this type of process to allow for a faster response to customer requests, as well as easier sharing of information among SC members. Distribution flexibility is critical as the availability of a range of delivery options with multiple carriers reduces the likelihood of late deliveries.

4.3.4 Process configurations for leagile strategy

The major management challenge in terms of leagile SC processes is achieving timely production and delivery at low cost while still offering wide product variety. Assemble-to-order processes are commonly utilized to allow a limited number of choices in the configuration of the final product. In order to offer customers a number of options, companies typically delay the final assembly of products until orders are received.

Make assemble-to-order process

Production processes are designed to produce standardized components in appropriate batch sizes, which are then assembled to fulfil individual customer orders. Since products are only differentiated after the decoupling point, components can be used for multiple products, reducing inventory and total production lead time. On the other hand, the reliance on standardized components limits the degree of customization possible for individual products.

Source/deliver assemble-to-order process

The assemble-to-order supply chain is typically controlled by the firms doing the assembly. These firms send the end products to the retailer/dealer for delivery to the end customer or directly to the end customer. Adopting a postponement strategy and shipping inventory direct to the customer can help companies improve their on-time delivery of complete orders, achieve more reliable and shorter lead times, introduce new products more quickly, reduce inventory costs and stabilize transportation costs. On the other hand, it can raise shipping costs, as it involves shipping in smaller quantities and using faster transport modes to reduce lead time (Balland and Lindholm, 2012).

4.3.5 An overall map for implementing supply chain policy

The framework takes a demand-driven approach; it assumes that SCs are designed to satisfy certain demand characteristics. The main featured strategies are lean, agile and leagile. For each of these, a specific configuration of policies must be implemented for the SC to achieve its strategic objectives.

In the case of lean strategy, the aim is to lower costs and find the most efficient way of utilizing the available resources. SC capabilities tend to be pre-planned well in advance, unvarying, and fixed with no excess capacity. Functions are operated within these capability restrictions: production is massive and standard; inventory is high-level and unvaried; transportation processes utilize low-cost modes, adopt a full truckload policy, and aim for the quickest possible transport time. In contrast, agile strategy aims to meet unpredictable customer demand in a short time and to achieve perfect fulfilment. The SC's resources tend to be variable, varied and excess. Production is varied, low-volume and has a short process time. Inventory is kept at a low level by means of postponement or quick response, with products being transported using fast and flexible modes.

Chapter 5 discusses the modelling and simulation for the framework, showing how it was used to model and simulate various demand characteristics and different SC policies in order to achieve certain objectives. Table 4.1 shows how the SC policies discussed above can be implemented for each of the featured strategies.

Demand Characteristics		Supply	22		Supply Chain Capabilities								
Qty	Vari. ety	Lead Time	Cert. ainty	Chain Strategy	Policies	Activities	Sched.	Capacity	Production	Inventory	Transportation		
					Source To Stock	Schedule product deliveries Receive product Transfer product		Preplanned Rigid Long period					
high	low	long or short	high	Efficiency: •Low cost •Asset utilization •Perfect order	Make To Stock	 Demand forecasting Define production rate (U/T) Define capacity level (U/T) Issue required materials Order scheduling Carry out production activity Release product to deliver Waste disposal 	•Preplanned •Rigid •Long period		High Unvaried Planned or continuous replenishm	•Low-cost mode •Reduce transportation time •Utilize assets			
				fulfilment	Deliver To Stock	 Receive order Determine delivery date Consolidate orders Plan product load and shipment Receive product Ship product Verify product received by customer 			•Push and mass production process	ent policy	(FTL)		
					Source To Order	Schedule product deliveries Receive product Transfer product							
low	high	short	short	short	short	Ag •Sa unp nort low e of sho	Agility: Satisfy Unpredictabl e orders in short time	Enter customer order Define production rate (WH/T) Define capacity level (WH/T) Issue required materials Order scheduling Carry out production activity Release product to deliver Waste disposal	•Based on order •Flexible •Short period	•Variable •Varied • Excess	 Group of products Low- volume Small lot size Low setup time Pull and batch production process 	•Low •Varied •Postpone ment or quick- response policy	•Fast mode •Flexible mode (LTL)
						•High level of service	Deliver To Order •F	Receive and configure order Determine delivery date Consolidate orders Plan product load and shipment Receive product Ship product Verify product received by customer					

Table 4. 1: Supply chain policies for sourcing, manufacturing and delivery

4.4 Network Model

Most studies in the area of supply chain network design (SCND) focus on one particular objective; for example, those addressing lean SCND focus on how to establish the optimal number and location of members so as to minimize overall SC costs (Shen, 2007). Those focusing on agile/responsive SCND aim to reduce products' time to market while achieving minimum total cost. In this category are Gunasekaran et al. (2015), who listed networking of partnering firms, information technology and knowledge management as the three major enablers of SC responsiveness. Finally, studies investigating sustainable SCND look at how SCs can be designed so that they meet current requirements without impacting on future generations. Neto et al. (2008) identified transportation, manufacturing, use of products, testing and end-of-use activities as all having a major impact on not just the economic but also the environmental performance of logistics networks.

The framework presented in Figure 4.1 draws on Long's (2014) hierarchical network model, itself based on the SCOR model, to arrive at a multiple objective network that integrates with SCOR to achieve different process policies. The network model shows that each SCOR process in network entities in the upper level can be decomposed into a set of five processes of lower level elements.

In the SCOR model, decomposition and aggregation are carried out based on process, but the proposed network model accomplishes the decomposition and aggregation based on both the entity and the process. Therefore, a simulation tool was used to integrate the SCOR process with the dynamic complex SCN. In order to fulfil the mentioned integration, two steps needed to be carried out in the SCN modelling. The first step was to determine the structure model with the proposed hierarchical framework that mainly describes the modules composing the SCN and their relationships without involvement of process element selection. The second step was to fulfil the structure model with corresponding functions using the simulation tool.

4.4.1 Drivers affecting supply chain network design

Strategic decision making in SCND is affected by a number of drivers. Those related to demand are discussed in section 4.2.1; this section describes other drivers and their impacts on SC decisions at the global level.

Completeness (item fill rate)

Item fill rate refers to the probability of having a product in stock when an order arrives. Where demand is volatile and unpredictable, selecting the most appropriate SC structure is particularly important as it affects the overall delivery reliability within the network (Lovell, Saw and Stimson, 2005). Centralizing inventories can help pool the risk and increase delivery reliability.

Delivery frequency

This is defined as the number of deliveries performed within a certain time unit (e.g. week, month or year). A high-frequency policy keeps inventory holding costs low but increases transportation costs. In these circumstances, efficiency and economy will be maximized if distribution is kept local. Low delivery frequency will incur lower transportation costs, which is preferable in global supply chains where goods must travel long distances.

Endowment of purchased items

This driver defines the availability of resources. Some countries/regions have geographical, technological or underground sources advantages and availability, while others face scarcity and the risk of "running out". If this is the case, manufacturing facilities should be located near suppliers with the easiest and cheapest access to the required resources. This may mean locating these facilities overseas.

Source quality

In the same way that availability issues can force a firm to look further afield for suppliers, it can also be forced to source from foreign suppliers if domestic resources do not satisfy its quality standards.

Table 4.2 summarizes the various SC drivers and analyses their impact on SCN strategy and SC performance.

Drivers	Direct effects	Suitable strategy	Strategy results	Effect on SC performance)
		Canabio charogy		Efficiency	Respon.	Sustain.
High product variety	 Wider inventory and suppliers Increased replenishment lead time 	Centralization	 Reduced duplication Higher transportation costs 	Increased (unless trans. cost is higher than inv. cost in case of decentralization strategy)	- ive	+ ive
Short lead time	 Adapt to changes quickly Intro. new products quickly 	 Locate plants near to market Local distribution 	Shorter lead times	- ive	+ ive	+ ive
Unpredicta ble demand	 Less accurate forecasting Impact of lost sales greater with innovative products 	 Locate distribution facilities closer to market 	 Shorter lead times Pooling the demand variations from different areas 	- ive	+ ive	+ ive
High demand	Increases cost because high levels of safety stock are required	Decentralization to achieve responsiveness and agility	More responsive to changeable customer requirements	- ive	+ ive	+ ive
variability	Increases demand volatility, especially in a more global SC	Centralization	Reduces the impact of variation	+ ive	- ive	- ive

Table 4. 2: Impact of supply chain drivers on network strategy

Drivore	Direct offects	Suitable strategy	Stratogy regults	Effect on SC performance			
Dilvers	Direct effects	Suitable Strategy	Strategy results	Efficiency	Respon.	Sustain.	
High delivery frequency	 Lower inventory holding costs Higher transportation costs (more frequent and fewer full loads) 	Local distribution	 Increased delivery frequency Reduced transportation distance 	+ ive	+ ive	+ ive	
High rate of completenes s (item fill rate)	Fewer stock outs	Centralization (among inventories)	Pooled risk and increased reliability	- ive	+ ive	+ ive	
Endowment of purchased items	Resources advantages and availability	Overseas manufacturing where required resources are unavailable locally	Easier and cheaper access to resources	+ ive	+ ive	+ ive	
Source quality	Improved product quality	Global sourcing when domestic resources are not up to quality standards	Achieves quality standards	+ ive	+ ive	Reduction in local sourcing but increase in global sourcing	

Cont. Table 4.2: Impact of supply chain drivers on network strategy

4.4.2 SCN configuration

Companies seeking to develop a global SC strategy must decide where to source raw materials and components, where to locate manufacturing facilities and which markets to serve. The answers to these three strategic questions will determine SC configuration; that is, whether the key functions within the operational process (sourcing, manufacturing and distribution) are located locally or globally. The decision where to locate these functions is critical. A number of factors must be taken into account, such as intended market, supply chain capabilities and competitive strategy. Global SCNs are complicated and need a high level of investment, but they are crucial if businesses are to take advantage of the cost, quality and availability advantages of foreign sources. On the other hand, a local SC is the most efficient way of serving the local market as it allows savings in transportation and inventory costs. Sourcing raw materials and components from local suppliers also reduces lead time and enables the firm to respond more quickly to the market. Table 4.3 presents various SCN configurations and their associated characteristics.

The table presents a number of strategies for combining efficiency and responsiveness. When SC design is entirely global, strategic emphasis tends to be given to high-volume production as a way of mitigating transportation and inventory costs. Where the SC is entirely local, on the other hand, lead time is short and only a low inventory level is needed to serve the market. In both local and global SCs, responsiveness requires increasing transportation frequency and consequently costs. Those chains that do not require frequent transportation will incur lower transportation costs and find it easier to pursue efficiency.

Supp	ly chain configura	Characteristics		
Supply	Manufacturing	Distribution	Characteristics	
Global	Local	Global	Huge investment, complex and sophisticated products	
Global	Global	Global	Highest level of complexity in terms of organization management, planning and coordination; tends to be adopted by global and large- scale companies	
Global	Global	Local	Global sourcing is inevitable, large volume in local market to be served	
Global	Local	Local	Exploits cost, quality and availability advantages of foreign sources in order to serve local market in best way possible	
Local	Local	Local	Low complexity and internalized cost efficiency; adopted by companies which have rigid manufacturing facilities, high inventory cost and high transportation cost	
Local	Global	Local	Short lead time, meets customer needs better; local advantages in terms of low labour cost, low taxes, better environmental norms and regulations	
Local worldwide	Local worldwide	Global	Global brands use "unique" local suppliers to add value; global distribution to be close to foreign markets	
Local	Local	Global	Adopted by strong global brands; local roots add value	

Table 4. 3: Network configurations

Where the SC combines global manufacturing with local distribution, the former raises efficiency while the latter reduces efficiency (because it incurs

higher inventory and transportation costs) but raises responsiveness. If manufacturing is located locally and serves both local and global markets, the SC can be efficient and responsive within the local market and efficient in the global market due to high inventory and transportation costs.

4.5 Performance Indicators Model

The last model included in the framework is a performance indicators model. This measures SC efficiency, responsiveness and sustainability and determines the extent to which the chain is achieving its objectives.

4.5.1 Efficiency performance indicators

Cost

A critical performance indicator, cost is tracked more carefully and comprehensively than any other aspect of competitive performance. SC costs include all costs associated with operating the SC, including the costs of planning, sourcing, material landed, production, order management, fulfilment and return.

Asset management

This refers to an organization's ability to manage its assets so that it is able to satisfy demand. Three indicators that measure SC asset management efficiency are cash-to-cash cycle times, inventory days of supply and asset turns. Asset turns are calculated by dividing revenue by total assets, including both working capital and fixed assets (Bolstorff and Rosenbaum, 2007).

4.5.2 Responsiveness performance indicators

Responsiveness refers to how quickly the SC is able to deliver products to the customer. It is measured as the time that elapses from a customer's order being received to completed delivery (order fulfilment lead time) (Cohen and Rousell, 2012).

4.5.3 Sustainability performance indicators

Sustainability indicators measure the impact of SC processes on the environment. Common indicators include: proportion of reusable/recyclable materials to total material input, raw material efficiency, proportion of cost of energy in production to total value of output, volume of air emissions per year (NOx, SOx, CO2, VOC, etc.) (Rao et al., 2008).

Table 4.4 summarizes the performance indicators utilized in the proposed model

Performance Attribute		Indicators	Sub-indicators
Efficiency	Cost	Total cost to serve	 Planning cost Sourcing cost Material landed cost Production cost Order management cost Fulfilment cost Returns cost
	Asset management efficiency	 Cash-to-cash cycle time Return on supply chain fixed assets Return on working capital 	
Respor	nsiveness	Order fulfilment cycle time	 Source cycle time Make cycle time Deliver cycle time Deliver retailer cycle time
Susta	iinability	 Reusable/recyclable materials to total material input Raw material efficiency Cost of energy to total value of output Air emissions (NOx, SOx, CO2, VOC, etc.) 	

Table 4. 4: Indicators of supply chain performance

4.6 Conclusions

Previous SC design efforts have only been able to achieve partial improvement in SC performance because they focus on isolated aspects of performance (e.g. cost or service) and do not take into account the complexities of the business environment. Crucially, they do not integrate strategy, process, network and performance into a single framework. The proposed framework addresses this problem by covering different aspects of SC performance and combining strategies to achieve the optimal levels of efficiency, responsiveness and sustainability throughout the whole supply chain.

The framework integrates the four key elements of strategy, process and resources. The strategic objective model suggests four SC strategies that can be deployed in response to different combinations of supply/demand conditions to achieve set goals and objectives. The process model then shows how these strategies can be implemented through different process configurations. It allows each entity in the SCN (e.g. supplier, manufacturer, distributor and retailer) to be modelled to ensure that SC resources are being deployed in accordance with the chosen policies and strategy.

The framework was subjected to modelling and simulation in order to validate it and demonstrate its applicability. Chapter 5 discusses the methodology that was employed, and presents the results of the modelling and simulation.

Chapter 5: Supply Chain Design: Modelling and Simulation

5.1 Introduction

This chapter describes the process through which the proposed framework, described in Chapter Four, was validated. This involved using the framework to design an SC for a hypothetical case study company and then using the Supply Chain Guru (SCG) software program to assess the performance of this design in terms of efficiency, responsiveness and sustainability. The first part of the chapter describes the building of a baseline model in SCG, the optimization process and the deployment of the program's simulation function to test the framework's performance under a range of demand scenarios. The second part of the chapter presents and discusses the results of the modelling and simulation stages and considers the extent to which they validate the framework.

5.2 Background

The SCG software package allows a single network model to be optimized and simulated without user interaction. It integrates network optimization, safety stock optimization, transportation optimization and simulation functions into a single SC optimization and simulation tool (see Figure 5.1), enabling companies to improve cost, service, sustainability and risk mitigation.



Figure 5. 1: Integrated solutions of Supply Chain Guru

The network optimization function determines the optimal network structure, flows and policies required to meet a defined demand, while the safety stock optimization function calculates the required safety stock, compares inventory cost to the achieved service level, and recommends inventory policy and associated parameters. The transportation optimization function aims to consolidate shipments through vehicle routing to minimize mileage, thereby reducing transportation costs. Finally, the simulation function allows the real-world system to be modelled over time for the purposes of validation and assessment.

5.3 Building the Supply Chain Guru Model

The SCG model consists of six main components: products, sites, demand, sourcing policy, transportation policy and inventory policy. This section discusses the data required for each of these components.

5.3.1 Products

Products are a key element in any SC model. Travelling across the nodes and lanes of the network, they are generally sourced, manufactured and shipped at defined cost and delivered to customers at a price to generate revenue. They can be represented by attributes such as name, value, price, weight and cubic volume.

5.3.2 Sites

Sites are essential within the SCN. The nodes between which products flow around the network, they may be customers, distributors, centres, factories, suppliers or ports. They may be defined in the model by name, type (e.g. existing facility, potential facility or customer) or location (address, geographic coordinates etc.).

5.3.3 Demand

Demand is central to the SCN because it describes the relationship between products and customers that drives the flow of the model. Demand information is sent through the network via sourcing policies until a facility in the network can satisfy the demand. The product is then sent back to the demanding customer via transportation policies. The demand relationship has four key elements: customer, product, quantity and time (i.e. when the customer places the order, though due date may also be factored into the simulation to determine whether a shipment is on time).

5.3.4 Sourcing policy

Sourcing policies link customers, distributors, manufacturers and suppliers, defining where a product is acquired from and determining the behaviour of source site, destination site and product. Sourcing policies can be represented in the model by attributes such as: source (origin of product), site (destination requesting the product), method by which a source is selected (e.g. make, multiple source or single source – see Table 5.1), product and lead time (time required before a request can be satisfied).

Sourcing Policy	Description
Single Source	Replenishment orders are filled by only one source facility
Multiple Sources	Multiple source policies vary; they may based on: - most inventory at potential suppliers - defined preference - random probabilities - split according to defined ratios - Fastest Path - Close to Due Date
Source by Transfer	Replenishment orders are never placed, regardless of inventory levels
Make	Allows for production by filling incoming orders within the site. Make policies include: - Make by Schedule - Make (Single Process): the first process that has enough capacity is selected - Make (Order of Preference) - Make (Process - Probability) - Make (Single BOM): the first BOM that has enough capacity is selected - Make (BOM - Probability)

Table 5. 1: Types of sourcing policy

5.3.5 Transportation policies

Transportation policies define how products travel through the network and the behaviour of the source site and destination site. Each non-production sourcing policy requires at least one corresponding transportation policy to allow for flow between the two sites. Every transportation policy must consider the source site, the destination site, the mode of transport between the two, and time. SCG provides a range of transportation policies to model the many different modes of transportation seen in any SC. Table 5.2 presents the most common transportation policies.

Transportation Policy	Description
Parcel	Product bundles are shipped immediately
LTL (Less than Truckload)	Product bundles are not aggregated
Full TL (Full Truckload)	Product bundles are aggregated to Full TL for shipment
Aggregate Container	Product bundles are assembled into containers for simpler shipping

Table 5. 2: Types of transportation policy

5.3.6 Inventory policies

Inventory policies define the relationship between products and sites. Because all non-customer sites have the potential to hold inventory, a policy must be defined to specify how the inventory is held. Inventory policies must have the following elements: site name, product name, initial inventory, reorder point (RP – the stock level at which to place another order), reorder amount/order up to quantity (RQ – the amount of product to order at each reorder) and reorder policy (the method by which reorders are placed). Inventory policies are the method by which RQ, RP and review period levels are set. Potential inventory policies include:

R,Q: a fixed replenishment point/fixed replenishment quantity inventory policy. When the inventory level on-hand falls below a certain replenishment point, R, the site will generate a replenishment order for a certain quantity, Q, of this product. Figure 5.2 presents R,Q policy.



Figure 5. 2: R,Q policy

Demand flow: a one-for-one replenishment policy; if one product is shipped, one is ordered to replace. This means that every order that arrives at a site for a certain product will generate a request for a replenishment order for the exact same quantity. Figure 5.3 presents demand flow policy.



Figure 5. 3: Demand flow policy

Although it seems straightforward, demand flow represents complex behaviour, especially when combined with review period. Setting a review period can produce batching as replenishment orders accumulate during the review period and are filled at its end.

S,s: a minimum/maximum inventory policy. When the inventory level onhand falls below a minimum, s, the site will generate a request for a replenishment order that will restore the on-hand inventory to a target, or maximum, number, S. Figure 5.4 presents S,s policy.



Figure 5. 4: S,s policy

T,S: periodic review order up to level is relatively simple and reflects realworld practice to a great extent. This policy is preferred in an intermittent demand context. SCG uses this policy to handle the lumpy demand class. The periodic policy provides the convenience of regular ordering days for the stock list and for the supplier who can plan efficient routing of the delivery vehicles.

Base stock: is preferable when economies of scale in the supply system are negligible relative to other factors. If individual units are very valuable, for example, holding and backorder costs may outweigh any fixed order costs. A base stock policy is also recommended for a slow-moving product (one with a low demand rate) where the economics of the situation rule out large batch sizes.

5.4 Supply Chain Guru: Modelling, Optimization and Simulation

This section discusses the steps involved in applying the SC modelling and simulation to the chosen case study. The section begins by introducing the case study SC before outlining the steps taken to build the baseline model and optimize the inventory and transportation functions. It then describes how the simulation scenarios were built and run to develop solutions for multiple SCD configurations.

5.4.1 Case study

The case study is a hypothetical SC manufacturing children's clothing, (see figure 5.5). The case study focuses specifically on a boys' cotton clothing set. Although the SC was created for the purpose of the study, the data used were taken from real websites; for example, production data were taken from the website of a real manufacturing company producing the same product, while transportation data were taken from a real haulage company. These data are presented in the input data tables in section 5.4.3.



Figure 5. 5: Supply chain of the case study

5.4.2 Overall method

The SCG modelling method comprised a sequence of steps, as described below (see Figure 5.6).

1. Demand modelling

As there were no historical demand data for the study's hypothetical SC, probability distributions were employed to generate different demand patterns (e.g. predictable and unpredictable). Normal distribution was

used to create smooth demand, lognormal distribution to produce erratic demand, bounded normal distribution to generate lumpy and slow demand, and Poisson distribution to produce slow, low variable demand.

2. Demand analysis

The demand analysis tool was then used to classify the nature of this demand (e.g. as smooth, erratic, slow or lumpy) according to defined statistical criteria (see Figure 5.9).

3. Safety stock optimization (SSO)

Once demand analysis had been completed, the safety stock function was run to calculate the required stock under the recommended inventory policy and its associated parameters (reorder point, reorder quantity). SSO was run for both predictable demand (smooth) and unpredictable demand (lumpy), with three scenarios in each demand category.

4. First-run simulation

In the absence of comprehensive real data, this step allowed the model to generate shipment details for use in the transportation optimization (TO) function (see Figure 5.15). The simulation ran three different time scenarios for each volume scenario, generating nine different shipment details in order to be populated to Transportation optimization function.

5. Transportation optimization (TO)

This provided a range of solutions in terms of routes, vehicles and shipment optimization, allowing the user to choose the most cost- and logistics-efficient. Transportation optimization involved the following substeps:

- Shipment details were divided into inbound and outbound and optimized separately (SCG does not support whole shipment optimization).
- Any unrouted shipments, which mostly happens because of vehicles capacity, were split manually and re-entered.
- All shipments could then be delivered and transportation optimization could be completed.

The focus on finding optimal solutions for inventory and transportation reflects the impact of these functions on SC efficiency and responsiveness. In inventory's case, SSO is employed to study the effect of demand volume on efficiency and responsiveness under a range of demand size scenarios. Similarly, TO is used to find the transportation solution that will best solve several scenarios for the length of time required to meet demand. Having optimized the problem in terms of inventory and transportation, the resulting data were used to run a second-run simulation.

6. Second-run simulation

This final step aimed to simulate different demand patterns (e.g. predictable and unpredictable) under different scenarios (e.g. changes in the demand level and time period) and to develop alternative solutions in each scenario to raise efficiency or responsiveness as required.



Figure 5. 6: SCG modelling and simulation method

5.4.3 Baseline model

A baseline model allows comparison between current SC performance and the results produced by the simulation under different scenarios. This allows the user to test changes and plan for improvements (see Figure 5.7).



Figure 5. 7: Using a baseline model

The baseline model imitates reality by replicating the real-life relationships between variables. The accuracy of this imitation is confirmed by comparing the model's performance with the real-life chain's historic performance under identical conditions. Once validated, the model can be employed to assess and improve the real-life SC. Figure 5.8 shows example inputs and outputs for a baseline model.



Figure 5. 8: Baseline model inputs and outputs

Tables 5.3 to 5.9 present the data inputs that were used in the baseline model for this case study.

Name *	Value (\$)	Price (\$)	Weight (Kg)	Cubic (m^3)
Р	7.5	9	0.75	0.02
R1	0.65	0.65	0.5	0.0133
R2	0.35	0.35	0.25	0.0066
R3	0.35	0.35	0.1	0.0001

Table 5. 3: Model data inputs - product

P: End product R: Raw material

Table 5. 4: Model data inputs - demand

Product Name	Site Name	Quantity*	Order Time	Occurrence	Time Between	Unit Price
					Orders	(\$)
Р	DIS1- Karachi	N(1000,70)	1/1/2018	INF	10 Days	9
Р	DIS2- Shanghai	N(1000,50)	1/1/2018	INF	10 Days	9
Р	DIS3- London	N(1000,150)	1/1/2018	INF	10 Days	9

* Based on normal distribution N (μ , σ)

Source Name	Site Name	Product Name	Sourcing Policy	BOM Name	Source Lead Time	Production Time*
	SUP1 New Delhi	R1	Make		1 Day	N(9,2) Day
	SUP2 Tokyo	R2	Make		1 Day	N(5,1) Day
	SUP3 Jakarta	R3	Make		1 Day	N(8,2) Day
	MAN- Beijing	Р	Make	P_BOM	1 Day	N(8,2) Day
SUP1 New Delhi	MAN- Beijing	R1	Single Source		8 HR	
SUP2 Tokyo	MAN- Beijing	R2	Single Source		8 HR	
SUP3 Jakarta	MAN- Beijing	R3	Single Source		8 HR	
MAN- Beijing	DIS1- Karachi	Р	Single Source		8 HR	
MAN- Beijing	DIS2- Shanghai	Р	Single Source		8 HR	
MAN- Beijing	DIS3- London	Р	Single Source		8 HR	

Table 5. 5: Model data inputs - sourcing policy

* Based on normal distribution N (µ, $\sigma)$

Table 5. 6: Model	data inputs - bi	ll of material (BOM)
-------------------	------------------	----------------------

Name	Product Name	Quantity	Туре
P_BOM	R1	1	Component
P_BOM	R2	2	Component
P_BOM	R3	1	Component

Source Site	Destina. tion Site	Product Name	Mode Rule	Source Policy	Cost (\$)*	Ship. Size	Distan. ce	Tran. Time (day)	Asset Nam e
SUP1- New Delhi	MAN- Beijing	R1	First	LTL	0.9	2400	11815	19	A1
SUP2 - Tokyo	MAN- Beijing	R2	First	LTL	0.0016	4836	1296	4	A2
SUP3- Jakarta	MAN- Beijing	R3	First	LTL	0.13	319200	5736	9	A3
MAN- Beijing	DIS1- Karachi	Ρ	First	Full TL	0.31	1596	10478	17	A4
MAN- Beijing	DIS2- Shanghai	Ρ	First	Full TL	0.43	1596	1220	2	A5
MAN- Beijing	DIS3- London	Р	First	Full TL	5.30	1596	20360	30	A6

Table 5. 7: Model data inputs - transportation policy

* Transportation cost per unit

Name	Units	Unit Fixed	Quantity	Quantity	Speed	Home Asset Base	
		Cost (\$)	Fill Level	Capacity	(Distance / H)		
A1	7	2165	1920	2400	26	SUP1-New Delhi	
4.2	25	FOG	2960	4926	1.4		
AZ	25	506	3869	4830	14	SUP2 -TOKYO	
A3	17	641	255360	319200	27	SUP3- Jakarta	
A4	7	500	1277	1596	26	MAN- Beijing	
A5	7	686	1277	1596	26	MAN- Beijing	
A6	7	8500	1277	1596	29	MAN- Beijing	

Table 5. 8: Model data inputs - transportation assets
Site Name	Prod. Name	Inven. Policy	Reorder Point	Reorder Qty	Initial Inven.	Safety Stock Basis	Service Req.	DOS Window
SUP1- New Delhi	R1	R,Q				Days Of Supply	0.95	25
SUP2 - Tokyo	R2	R,Q				Days Of Supply	0.95	25
SUP3- Jakarta	R3	R,Q				Days Of Supply	0.95	25
MAN- Beijing	Р	R,Q				Days Of Supply	1.00	25
MAN- Beijing	R1	R,Q				Days Of Supply	1.00	25
MAN- Beijing	R2	R,Q				Days Of Supply	1.00	25
MAN- Beijing	R3	R,Q				Days Of Supply	1.00	25

Table 5. 9: Model data inputs - inventory policy

5.4.4 Optimization of baseline model

The baseline model was optimized for two main functions, inventory and transportation. SSO was run to find optimum inventory policy solutions, while TO was run to find the optimum level of transportation resources required.

Safety stock optimization (SSO)

Safety stock is buffer stock that is maintained to mitigate the risk of stockouts due to uncertainties in supply and demand. It insures against variability in demand and lead time. Prior to SSO, demand analysis must be performed to propagate customer demand to upstream sites, calculate demand statistics and classify demand into different categories. SSO helps the user identify demand characteristics and determine whether demand is intermittent or non-intermittent, smooth, erratic, slow or lumpy. Figure 5.9 presents the statistical parameters underlying these classifications.



Figure 5. 9: Demand classification

Table 5.10 shows one of demand analysis outputs for the baseline model in this study; that is, the customer demand profile.

Table 5. 10: Customer demand profile

Customer Name	Prod. Name	Intermittency	Demand Class	Inter-demand Interval Mean	Non-Zero Demand <i>CV</i> ²	Non-Zero Demand Std Dev
DIS1- Karachi	Р	Non- Intermittent	Smooth	1	0.03	532.7293
DIS2- Shanghai	Р	Non- Intermittent	Smooth	1	0.02	471.9285
DIS3- London	Р	Non- Intermittent	Smooth	1	0.03	1100.874

Demand analysis was performed to model a range of demand patterns, as shown in Figures 5.10 - 5.14.

DIS	Normal		Intermittency	Non Intermittent
Mean	500		ADI	1
StDv	25	300	CV ²	0.0037
-	-	200	Demand Class	Smooth
-	-	0	-	-

Figure 5. 10: Smooth demand



Figure 5. 11: Erratic demand



Figure 5. 12: Slow demand



Figure 5. 13: Lumpy demand



Figure 5. 14: Slow demand

SSO also helps the user determine when to replenish inventory (RP) and by how much (RQ) in order to optimize inventory holding cost and reduce the risk of stock-out. The SSO function in SCG recommends an inventory policy that optimizes inventory cost and service level for a given demand pattern. Table 5.11 presents the inventory policy recommended in the baseline model for the case study supply chain.

Site Name	Prod.	Recommended	Parameter1	Parameter2
Site Name	Name	Policy	(R)	(Q)
MAN- Beijing	Р	R,Q	0	3796
MAN- Beijing	R1	R,Q	12252	14425
MAN- Beijing	R2	R,Q	7662	6074
MAN- Beijing	R3	R,Q	6885	6833
SUP1-New Delhi	R1	R,Q	0	3796
SUP2 -Tokyo	R2	R,Q	0	7592
SUP3- Jakarta	R3	R,Q	0	3796

Table 5. 11: Inventory policy summary

Transportation optimization (TO)

TO allows the user to resolve the vehicle routing problems (VRP) associated with the consolidation of shipments into multi-stop routes. The aim is to minimize the total transportation cost by finding the optimum number of vehicles (assets) required for the routed shipments. The first step in this process is to generate a shipment transaction table from the simulation model. Figure 5.15 shows the shipment transaction table generated for the case study company, assuming a smooth demand scenario.

			~ I						
Run model Simulation metings Staphs and Reports Custom Reporting									
hipment Trans	actions [Sim _no Ass	ets]							- 0
	🛌 🗛 🕺 🎍	2 10 11 1		Ur	filtered)	- 7 6		Row Limit: 1	000
Scenario	Source Name	Destination Name	Product Name	Quantity	Time /	CO2	Weight	Cubic Mor	de Shinment Number Transports
Sections	Source Hame	Destination marie	1 ICODECTION IC	Gounday				CODIC MON	se supriere realizer rransporte
seline	SUP2-TOKYO	MAN-BEIJING	R2	2116.00	1/9/2018	0.00	529.00	13.97 1	2
seline	SUP2-TOKYO	MAN-BEIJING	R2	46.00	1/11/2018	0.00	11.50	0.30 1	4
seline	SUP2-TOKYO	MAN-BEIJING	R2	2116.00	1/11/2018	0.00	529.00	13.97 1	4
seline	SUP2-TOKYO	MAN-BEIJING	R2	2186.00	1/11/2018	0.00	546.50	14.43 1	4
oclinc	SUP2 TOKYO	MAN BEIJING	R2	2162.00	1/11/2018	0.00	540.50	14.27 1	6
seline	SUP3-JAKARTA	MAN-BEIJING	R3	1081.00	1/15/2018	0.00	108.10	0.11.1	3
seline	SUP3-JAKARTA	MAN-BEIJING	R3	1058.00	1/16/2018	0.00	105.80	0.11 1	5
seline	SUP3-JAKARTA	MAN-BEIJING	R3	1116.00	1/16/2018	0.00	111.60	0.11 1	5
seline	SUP3-JAKARTA	MAN-BEIJING	R3	1058.00	1/18/2018	0.00	105.80	0.11 1	8
seline	SUP2-TOKYO	MAN-BEIJING	R2	3723.00	1/19/2018	0.00	930.75	24.57 1	10
seline	SUP2-TOKYO	MAN-BELUNG	R2	146 00	1/19/2018	0.00	36 50	0.96 1	10
seline	SUP2-TOKYO	MAN-BEIJING	R2	2892.00	1/19/2018	0.00	723.00	19.09 1	11
seline	SUP2-TOKYO	MAN-BEIJING	R2	831.00	1/19/2018	0.00	207.75	5.48 1	12
seline	SUP2-TOKYO	MAN-BEIJING	R2	2068.00	1/19/2018	0.00	517.00	13.65 1	12
seline	SUP2-TOKYO	MAN-BEIJING	R2	1247.00	1/19/2018	0.00	311.75	8.23 1	12
seline	SUP2-TOKYO	MAN-BEIJING	R2	801.00	1/20/2018	0.00	200.25	5.29 1	13
seline	SUP2-TOKYO	MAN-BEIJING	R2	2922.00	1/20/2018	0.00	730.50	19.29 1	13
seline	SUP2-TOKYO	MAN-BEIJING	R2	1224.00	1/20/2018	0.00	306.00	8.08 1	14
seline	SUP2-TOKYO	MAN-BEIJING	R2	824.00	1/20/2018	0.00	206.00	5.44 1	14
seline	SUP2-TOKYO	MAN-BEIJING	R2	2068.00	1/20/2018	0.00	517.00	13.65 1	15
oclinc	SUP2 TOKYO	MAN BEIJING	R2	977.00	1/21/2018	0.00	244.25	6.45 1	17
seline	SUP2-TOKYO	MAN-BEIJING	R2	1228.00	1/21/2018	0.00	307.00	8.10 1	17
seline	SUP2-TOKYO	MAN-BEIJING	R2	2641.00	1/22/2018	0.00	660.25	17.43 1	18
oolino	CUED TOKYO	MAN DELING	ph .	1000 00	1/00/0010	0.00	207.00	0 10 1	10

Figure 5. 15: Shipment transactions excluding assets

The shipment table is then separated into two: inbound shipments of raw materials from all suppliers (R1, R2 and R3) and outbound shipments of products (P) for distributors. These are fed into the TO model for a first-run simulation (see figure 5.6). The important output from this step is the identification of unrouted shipments – these are usually the result of limited vehicle capacity. Figure 5.16 shows an unrouted shipment table for the case study company.

Nort Total Modal Bula Operation Smalland Transportation Operation Smalland Transportation Operation Statu Scanson Tormany	🕦 🔡 TO in_	Base BLT (first i	run)	- 🚊 🛝	🥥 🕫				Supply Ch	ain Guru [SCD N	Aodels_predi	ctable dem	and]			-	er X
Statu Scrawing Sc	File Hom	e Tools	Model Data	Optimization	Simulation	Transportation Op	timization							s	earch LLam	asoft Support Website	Q 0
One of the state (10 in the state (10 in the state)) Image: State (10 in the state) Image: State (10 in the state) <th>State</th> <th>us Scenarios</th> <th>TO Summary Optim</th> <th>Routes Shipm</th> <th>Routes</th> <th>Assets Shipments</th> <th>Tableau Graphs</th> <th>Relationship Constraint Editor</th> <th>Shipments Builder</th> <th>Distance B Matrix Calculator S Tools</th> <th>Business P Hours Icheduler</th> <th>eriodic VRF Scheduler</th> <th></th> <th></th> <th></th> <th></th> <th></th>	State	us Scenarios	TO Summary Optim	Routes Shipm	Routes	Assets Shipments	Tableau Graphs	Relationship Constraint Editor	Shipments Builder	Distance B Matrix Calculator S Tools	Business P Hours Icheduler	eriodic VRF Scheduler					
Scenario Name Product Name Source Site Destination Site Vieight Cubic Off Direct Shipping Co Comp2 City Comp2 City <td>🔀 Unrouted Sh</td> <td>nipments ITO in</td> <td>Base BLT (first</td> <td>run)]</td> <td></td> <td>1 88</td> <td></td> <td></td>	🔀 Unrouted Sh	nipments ITO in	Base BLT (first	run)]											1 88		
Scenario Name Product Name Scansistic Name Product Name Scansistic			71 17 5										1000			-	
Scenario Name Product Name Description Description Cubic Gty Direct Shaping Co Comp2 Cubit Cubit Comp2 Cubit			AU 25	W 🕺 🚇	- X		nfiltered)		16 6		× 1	Kow Limit:	1000				
Stateline Sh2 P2 SUP2-Tokyo MAN-Beging O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O O	Scenario	o Na	ame Pr	roduct Name	Source Site	Destination Site	Weigh	1	Cubic	Qty	Direct Sh	ipping Co	Comp2 Qty	Comp2 Cubi	<u> </u>		
Statiline Statiline <t< td=""><td>Paseline</td><td>SF5</td><td>D2</td><td></td><td>SUP2-Tokyo</td><td>MAN-Rejijing</td><td></td><td>0.00</td><td>0.00</td><td>10620</td><td>00</td><td>0</td><td></td><td>I</td><td></td><td></td><td></td></t<>	Paseline	SF5	D2		SUP2-Tokyo	MAN-Rejijing		0.00	0.00	10620	00	0		I			
Statine St27 R2 SUP7-Tayo VAA-Bajing 0.00 1050.00 0 Baseline Sk28 R2 SUP7-Tayo VAA-Bajing 0.00 0.00 1050.00 0 Baseline Sk88 R2 SUP7-Tayo VAA-Bajing 0.00 0.00 1053.00 0 Baseline Sk88 R2 SUP7-Tayo VAA-Bajing 0.00 0.00 6074.00 0 Baseline Sk88 R2 SUP7-Tayo VAA-Bajing 0.00 0.00 6072.00 0 Baseline Sk55 R2 SUP7-Tayo VAA-Bajing 0.00 0.00 6072.00 0 Baseline Sk56 R2 SUP7-Tayo VAA-Bajing 0.00 0.00 6076.00 0 Baseline Sk56 R2 SUP7-Tayo VAA-Bajing 0.00 0.00 6076.00 0 Baseline Sk77 R2 SUP7-Tayo VAA-Bajing 0.00 0.00 2885.00 0	Raseline	Sh3	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6074	.00	0			=		
State State R2 SUP2-Takyo MAN-Beijing COO COO 6074.00 O Baseline SN88 R2 SUP2-Takyo MAN-Beijing COO COO 6074.00 O Baseline SN89 R2 SUP2-Takyo MAN-Beijing COO COO 6074.00 O Baseline SN55 R2 SUP2-Takyo MAN-Beijing COO COO 6072.00 O Baseline SN55 R2 SUP2-Takyo MAN-Beijing COO COO 6072.00 O Baseline SN56 R2 SUP2-Takyo MAN-Beijing COO COO 6072.00 O Baseline SN68 R2 SUP2-Takyo MAN-Beijing COO COO 6075.00 O Baseline SN43 R2 SUP2-Takyo MAN-Beijing COO COO 6075.00 O O D D D D D D D D D D	Baseline	Sh27	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	10630	.00	0					
Sk8 R2 SUP2-Tayo MAN-Beijing O.00 1033.00 0 Baseline Sk69 R2 SUP2-Tayo MAN-Beijing O.00 1033.00 0 Baseline Sh10 R2 SUP2-Tayo MAN-Beijing O.00 6074.00 0 Baseline Sh10 R2 SUP2-Tayo MAN-Beijing O.00 6072.00 0 Baseline Sh66 R2 SUP2-Tayo MAN-Beijing O.00 6072.00 0 Baseline Sh66 R2 SUP2-Tayo MAN-Beijing O.00 6076.00 0 Baseline Sh67 R2 SUP2-Tayo MAN-Beijing O.00 6076.00 0 Baseline Sh77 R2 SUP2-Tayo MAN-Beijing O.00 2885.00 0 Baseline Sh32 R1 SUP1-Mevo Delhi MAN-Beijing O.00 2885.00 0 Baseline Sh32 R2 SUP2-Tayo MAN-Beijing O.00 Co.00	Baseline	Sh28	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6074	.00	0					
Stepsine	Baseline	Sh58	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	10630	.00	0					
Shi0 R2 SUP2-Tayo MAN-Bajing 0.00 6072.00 0 Baseline Sh55 R2 SUP2-Tayo MAN-Bajing 0.00 6072.00 0 Baseline Sh65 R2 SUP2-Tayo MAN-Bajing 0.00 6072.00 0 Baseline Sh65 R2 SUP2-Tayo MAN-Bajing 0.00 6072.00 0 Baseline Sh66 R2 SUP2-Tayo MAN-Bajing 0.00 6076.00 0 Baseline Sh67 R2 SUP2-Tayo MAN-Bajing 0.00 6076.00 0 Baseline Sh35 R2 SUP2-Tayo MAN-Bajing 0.00 6076.00 0 Baseline Sh37 R1 SUP1-Mayo MAN-Bajing 0.00 0.00 28850.00 0 Baseline Sh307 R1 SUP1-Mayo MAN-Bajing 0.00 6074.00 0 Baseline Sh42 R2 SUP2-Tayo MAN-Bajing 0.00 6074.00<	Baseline	Sh59	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6074	.00	0					
Baseline Sh35 R2 SUP2-Tayo MAN-Bajing 0.00 6072.00 0 Baseline Sh66 R2 SUP2-Tayo MAN-Bajing 0.00 6072.00 0 Baseline Sh11 R2 SUP2-Tayo MAN-Bajing 0.00 6072.00 0 Baseline Sh11 R2 SUP2-Tayo MAN-Bajing 0.00 6076.00 0 Baseline Sh77 R2 SUP2-Tayo MAN-Bajing 0.00 6076.00 0 Baseline Sh32 R2 SUP2-Tayo MAN-Bajing 0.00 6076.00 0 Baseline Sh33 R1 SUP1-Mev Delhi MAN-Bajing 0.00 600 2855.00 0 Baseline Sh36 R2 SUP2-Tayo MAN-Bajing 0.00 600 2855.00 0 Baseline Sh36 R2 SUP2-Tayo MAN-Bajing 0.00 6074.00 0 Baseline Sh37 R2 SUP2-Tayo MAN-Bajin	Baseline	Sh10	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6072	.00	0					
Seadine Sh6 R2 SUP2-Tayo MAN-Bajing 0.00 6072.00 0 Baseline Sh11 R2 SUP2-Tayo MAN-Bajing 0.00 6076.00 0 Baseline Sh36 R2 SUP2-Tayo MAN-Bajing 0.00 6076.00 0 Baseline Sh36 R2 SUP2-Tayo MAN-Bajing 0.00 6076.00 0 Baseline Sh33 R1 SUP1-Tayo MAN-Bajing 0.00 0.00 8076.00 0 Baseline Sh37 R1 SUP1-Mayo MAN-Bajing 0.00 0.00 28850.00 0 Baseline Sh307 R1 SUP1-Mayo MAN-Bajing 0.00 0.00 28850.00 0 Baseline Sh307 R2 SUP2-Tayo MAN-Bajing 0.00 0.00 6074.00 0 Baseline Sh429 R2 SUP2-Tayo MAN-Bajing 0.00 0.00 6074.00 0 Baseline Sh427	Baseline	Sh35	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6072	.00	0					
Baseline Sh11 R2 SUP2-Tokyo MAN-Bajing 0.00 6076.00 0 Baseline Sh26 R2 SUP2-Tokyo MAN-Bajing 0.00 6076.00 0 Baseline Sh76 R2 SUP2-Tokyo MAN-Bajing 0.00 6076.00 0 Baseline Sh77 R2 SUP2-Tokyo MAN-Bajing 0.00 6076.00 0 Baseline Sh33 R1 SUP1-Mev Delhi MAN-Bajing 0.00 600 28850.00 0 Baseline Sh48 R1 SUP1-Mev Delhi MAN-Bajing 0.00 600 28850.00 0 Baseline Sh40 R2 SUP2-Tokyo MAN-Bajing 0.00 600 6074.00 0 Baseline Sh43 R2 SUP2-Tokyo MAN-Bajing 0.00 600 674.00 0 Baseline Sh23 R2 SUP2-Tokyo MAN-Bajing 0.00 600 674.00 0 674.00 0 0 <td>Baseline</td> <td>Sh66</td> <td>R2</td> <td></td> <td>SUP2-Tokyo</td> <td>MAN-Beijing</td> <td></td> <td>0.00</td> <td>0.00</td> <td>6072.</td> <td>.00</td> <td>0</td> <td></td> <td></td> <td></td> <td></td> <td></td>	Baseline	Sh66	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6072.	.00	0					
Baseline Sh.6 R2 SUP2-Tayo MAN-Beging 0.00 6076.00 0 Baseline Sh.67 R2 SUP2-Tayo MAN-Beging 0.00 6076.00 0 Baseline Sh.33 R1 SUP1-Mayo MAN-Beging 0.00 0.00 6076.00 0 Baseline Sh.33 R1 SUP1-Mevo Delhi MAN-Beging 0.00 0.00 28850.00 0 Baseline Sh.37 R1 SUP1-Mevo Delhi MAN-Beging 0.00 0.00 28850.00 0 Baseline Sh.39 R2 SUP2-Tayo MAN-Beging 0.00 0.00 6074.00 0 Baseline Sh.49 R2 SUP2-Tayo MAN-Beging 0.00 0.00 6074.00 0 Baseline Sh.27 R2 SUP2-Tayo MAN-Beging 0.00 0.00 4553.00 0 Baseline Sh.27 R2 SUP2-Tayo MAN-Beging 0.00 0.00 4553.00 0 </td <td>Baseline</td> <td>Sh11</td> <td>R2</td> <td></td> <td>SUP2-Tokyo</td> <td>MAN-Beijing</td> <td></td> <td>0.00</td> <td>0.00</td> <td>6076.</td> <td>.00</td> <td>0</td> <td></td> <td></td> <td></td> <td></td> <td></td>	Baseline	Sh11	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6076.	.00	0					
Baseline Sk7 R2 SUP2-Tokyo MAN-Beijing 0.00 60760 0 Baseline Sk3 R1 SUP1-Havo Delhi MAN-Beijing 0.00 0.00 28550.00 0 Baseline Sk48 R1 SUP1-Havo Delhi MAN-Beijing 0.00 0.00 28550.00 0 Baseline Sk49 R1 SUP1-Havo Delhi MAN-Beijing 0.00 0.00 28550.00 0 Baseline Sk00 R2 SUP1-Tokyo MAN-Beijing 0.00 0.00 6074.00 0 Baseline Sk24 R2 SUP1-Tokyo MAN-Beijing 0.00 0.00 6074.00 0 Baseline Sk27 R2 SUP1-Tokyo MAN-Beijing 0.00 0.00 6475.00 0 Baseline Sk57 R2 SUP1-Tokyo MAN-Beijing 0.00 0.00 4553.00 0 Baseline Sk57 R2 SUP1-Tokyo MAN-Beijing 0.00 0.00 6074.0	Baseline	Sh36	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6076.	.00	0					
Baseline SN23 R1 SUPT-Heve Delhi MAH-Beijing 0.00 28850.0 0 Baseline SN37 R1 SUPT-Heve Delhi MAH-Beijing 0.00 0.00 28850.0 0 Baseline SN377 R1 SUPT-Heve Delhi MAH-Beijing 0.00 0.00 28850.0 0 Baseline SN377 R1 SUPT-Heve Delhi MAH-Beijing 0.00 0.00 28850.0 0 Baseline SN459 R2 SUP2-Tokyo MAH-Beijing 0.00 0.00 6574.00 0 Baseline SN457 R2 SUP2-Tokyo MAH-Beijing 0.00 0.00 6574.00 0 Baseline SN457 R2 SUP2-Tokyo MAH-Beijing 0.00 0.00 4553.00 0 Baseline SN457 R2 SUP2-Tokyo MAH-Beijing 0.00 0.00 4553.00 0 Baseline SN450 R2 SUP2-Tokyo MAH-Beijing 0.00 600	Baseline	Sh67	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6076.	.00	0					
Baseline Baseline Baseline Baseline Sh00 Sh48 R R1 SUPI-Have Delhi MAN-Bajing 0.00 0.00 2850.00 0 Baseline Baseline Sh00 R2 SUPI-Avayo MAN-Bajing 0.00 0.00 6074.00 0 Baseline Baseline Sh249 R2 SUPI-Avayo MAN-Bajing 0.00 0.00 6074.00 0 Baseline Sh257 R2 SUPI-Avayo MAN-Bajing 0.00 0.00 6474.00 0 Baseline Sh257 R2 SUPI-Avayo MAN-Bajing 0.00 0.00 6475.00 0 Baseline Sh257 R2 SUPI-Avayo MAN-Bajing 0.00 0.00 4553.00 0 Baseline Sh250 R2 SUPI-Avayo MAN-Bajing 0.00 0.00 4553.00 0 Baseline Sh260 R2 SUPI-Avayo MAN-Bajing 0.00 0.00 6074.00 0 Baseline Sh260 R2 SUPI-Avayo MAN-Bajing 0.00 0.00 6074.00 0 Baseline Sh260 R	Baseline	Sh23	R1		SUP1-New Delhi	MAN-Beijing		0.00	0.00	28850	.00	0					
Baseline SN07 R1 SUPT-Nov MAN-Baijing 0.00 288500 0 Baseline SN40 R2 SUP2-Tokyo MAN-Baijing 0.00 0.00 65074.00 0 Baseline SN424 R2 SUP2-Tokyo MAN-Baijing 0.00 0.00 65074.00 0 Baseline SN422 R2 SUP2-Tokyo MAN-Baijing 0.00 0.00 65074.00 0 Baseline SN48 R2 SUP2-Tokyo MAN-Baijing 0.00 0.00 65074.00 0 Baseline SN57 R2 SUP2-Tokyo MAN-Baijing 0.00 0.00 4553.00 0 Baseline SN57 R2 SUP2-Tokyo MAN-Baijing 0.00 0.00 4553.00 0 Baseline SN65 R2 SUP2-Tokyo MAN-Baijing 0.00 0.00 6074.00 0 Baseline SN65 R2 SUP2-Tokyo MAN-Baijing 0.00 6074.00 0	Baseline	Sh48	R1		SUP1-New Delhi	MAN-Beijing		0.00	0.00	28850	.00	0					
Baseline Sk0 R2 SUP2-Takyo MAN-Baijing 0.00 6074.00 0 Baseline Sk249 R2 SUP2-Takyo MAN-Baijing 0.00 6074.00 0 Baseline Sk22 R2 SUP2-Takyo MAN-Baijing 0.00 6074.00 0 Baseline Sk22 R2 SUP2-Takyo MAN-Baijing 0.00 6074.00 0 Baseline Sk28 R2 SUP2-Takyo MAN-Baijing 0.00 6074.00 0 Baseline Sk577 R2 SUP2-Takyo MAN-Baijing 0.00 0.00 4553.00 0 Baseline Sk578 R2 SUP2-Takyo MAN-Baijing 0.00 0.00 4553.00 0 Baseline Sk581 R2 SUP2-Takyo MAN-Baijing 0.00 0.00 6074.00 0 Baseline Sk581 R2 SUP2-Takyo MAN-Baijing 0.00 6074.00 0 Baseline Sk581 R2	Baseline	Sh307	R1		SUP1-New Delhi	MAN-Beijing		0.00	0.00	28850.	.00	0					
Baseline SN429 R2 SUP27-tokyo MAN-Bajing 0.00 6074.00 0 Baseline SN422 R2 SUP27-tokyo MAN-Bajing 0.00 6074.00 0 Baseline SN488 R2 SUP27-tokyo MAN-Bajing 0.00 0.00 4553.00 0 Baseline SN57 R2 SUP27-tokyo MAN-Bajing 0.00 0.00 4553.00 0 Baseline SN575 R2 SUP27-tokyo MAN-Bajing 0.00 0.00 4553.00 0 Baseline SN575 R2 SUP27-tokyo MAN-Bajing 0.00 0.00 4553.00 0 Baseline SN630 R2 SUP27-tokyo MAN-Bajing 0.00 0.00 6074.00 0 Baseline SN630 R2 SUP27-tokyo MAN-Bajing 0.00 6074.00 0 - Baseline SN630 R2 SUP27-tokyo MAN-Bajing 0.00 6074.00 0 - <td>Baseline</td> <td>Sh80</td> <td>R2</td> <td></td> <td>SUP2-Tokyo</td> <td>MAN-Beijing</td> <td></td> <td>0.00</td> <td>0.00</td> <td>6074.</td> <td>.00</td> <td>0</td> <td></td> <td></td> <td></td> <td></td> <td></td>	Baseline	Sh80	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6074.	.00	0					
Baseline Sh422 R2 SUP2-Tokyo MAN-Bajing 0.00 607400 0 Baseline Sh48 R2 SUP2-Tokyo MAN-Bajing 0.00 0.00 6455100 0 Baseline Sh57 R2 SUP2-Tokyo MAN-Bajing 0.00 0.00 4455100 0 Baseline Sh57 R2 SUP2-Tokyo MAN-Bajing 0.00 0.00 4455100 0 Baseline Sh57 R2 SUP2-Tokyo MAN-Bajing 0.00 0.00 4455100 0 Baseline Sh671 R2 SUP2-Tokyo MAN-Bajing 0.00 0.00 607400 0 Baseline Sh280 R2 SUP2-Tokyo MAN-Bajing 0.00 607400 0 Baseline Sh280 R2 SUP2-Tokyo MAN-Bajing 0.00 607400 0 Baseline Sh280 R2 SUP2-Tokyo n 0.00 607400 0 Baseline Sh281	Baseline	Sh249	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6074.	.00	0					
Baseline Sh88 F2 SUP2-Tokyo MAN-Beijing 0.00 4553.00 0 Baseline Sh257 F2 SUP2-Tokyo MAN-Beijing 0.00 4553.00 0 Baseline Sh257 F2 SUP2-Tokyo MAN-Beijing 0.00 4553.00 0 Baseline Sh257 F2 SUP2-Tokyo MAN-Beijing 0.00 4553.00 0 Baseline Sh268 F2 SUP2-Tokyo MAN-Beijing 0.00 600 6074.00 0 Baseline Sh269 F2 SUP2-Tokyo MAN-Beijing 0.00 600 6074.00 0 Baseline Sh269 F2 SUP2-Tokyo MAN-Beijing 0.00 600 6074.00 0	Baseline	Sh922	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6074.	.00	0					
Baseline Baseline Baseline Baseline Sh673 Sh275 R2 SUP2-Tokyo SUP2-Tokyo MAN-Bajing MAN-Bajing 0.00 0.00 455:00 0 Baseline Baseline Sh670 Sh671 R2 SUP2-Tokyo MAN-Bajing 0.00 0.00 6074:00 0 Baseline Sh670 R2 SUP2-Tokyo MAN-Bajing 0.00 0.00 6074:00 0 Baseline Sh670 R2 SUP2-Tokyo MAN-Bajing 0.00 0.00 6074:00 0 Baseline Sh670 R2 SUP2-Tokyo NAN-Bajing 0.00 0.00 6074:00 0 Baseline Sh670 R2 SUP2-Tokyo NAN-Bajing 0.00 0.00 6074:00 0 Baseline Sh670 R2 SUP2-Tokyo NAN-Bajing 0.00 6070 0 7 Row1 of 122 ID (Unfiltered) 0.00 6070 0 7 0 7	Baseline	Sh88	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	4563.	.00	0					
Baseline SN/578 R2 SUP2-Tokyo MAN-Beijing 0.00 4563.00 0 Baseline SN/81 R2 SUP2-Tokyo MAN-Beijing 0.00 0.00 65074.00 0 Baseline SN/80 R2 SUP2-Tokyo MAN-Beijing 0.00 0.00 65074.00 0 Baseline SN/80 R2 SUP2-Tokyo MAN-Beijing 0.00 0.00 65074.00 0 Causian D United Stream 0.00 0.00 6574.00 0	Baseline	Sh257	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	4563.	.00	0					
Descrine Baseline Descrine Sk50 Sk51 R2 SUP2-Tokyo SUP2-Tokyo MAN-Beijing 0.00 6074.00 0 Descrine Sk59 P2 SUP2-Tokyo Sk59 P2 SUP2-Tokyo Sk59 0.00 0.00 6074.00 0 Descrine Sk59 P2 SUP2-Tokyo Sk59 0.00 0.00 6074.00 0 Row1 of 122 ID (Unfiltered) 0.00 0.00 6072.00 0	Baseline	Sh578	R2		GUP2-Tokyo	MAN-Beijing		0.00	0.00	4563.	.00	0					
Baseline Sk80 R2 SUP2-Tokyo MAN-Baijing 0.00 6074.00 0 c ID clip2 Table III 0.00 control crossing	Baseline	Sh91	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6074.	.00	0					
Operation Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk Disk <thdisk< th=""> <thdisk< th=""> Disk<td>Baseline</td><td>Sh260</td><td>R2</td><td></td><td>SUP2-Tokyo</td><td>MAN-Beijing</td><td></td><td>0.00</td><td>0.00</td><td>6074.</td><td>.00</td><td>0</td><td></td><td></td><td></td><td></td><td></td></thdisk<></thdisk<>	Baseline	Sh260	R2		SUP2-Tokyo	MAN-Beijing		0.00	0.00	6074.	.00	0					
Row 1 of 122 ID (Unfiltered)	Pasalina	CLE01	D 1		CIIDO Takua	MAN Dailing		0.00	0.00	6074	00	0					
	Row 1 of 122				ID (Unfilte	red)									,		

Figure 5. 16: Unrouted shipment table

In order to find the accurate optimized asset number, unrouted shipments should be split manually as SCG does not support this function. Once this has been done, the input shipment is adjusted accordingly and TO can be run again to find the optimum asset number. Figure 5.17 shows the asset summary table obtained from the inbound TO model for the case study company.

🚯 🔚 TO in_Base BLT 🔹 🏣 🛝 🥥 🗢		Supply Chain Guru [SCD Models_predictable demand]	- @ ×
File Home Tools Model Data Optimization Simulation	Transportation Optimization		Search LLamasoft Support Website 🔍 😧 🔹
Status Scenarios To Foundation Foundation	Assets Shipments Tableau Graphs and Reports	Shipmeng Builder Distance Buinness Periodic/VPP Calculator Scheduler Cole	
			• X
	· (Unfiltered)	🏹 🏹 🏹 🏋 🥅 📰 Row Limit: 1000 🔂	
Scenario Asset Availability I Site Name Asset Name	Assets Available Assets Used Sce	narioID Sub-Scenario ID ID /	
Baseline 1 SUP2-Tokyo A2 Baseline 2 SUP3-Jakarta A3 Baseline 0 SUP1-New Delhi A1	1000 30 1000 2 1000 83	0 1 129 0 1 130 0 1 131	
Row 1 of 3 ID Unfilter	ed)		

Figure 5. 17: Asset summary table

5.4.5 Simulation scenarios in the baseline model

Having identified policy parameters and achieved optimized solutions for the inventory and transportation functions, the next step was to build a range of simulation scenarios to test the framework's ability to deal with certain and uncertain demand characteristics.

Smooth demand scenarios (efficient strategy)

Smooth demand, which is characterized by a high level of certainty and highvolume quantities, generally requires that a lean strategy be implemented. According to the strategic objective model in the proposed framework, lean strategy can be applied by adopting different inventory policies depending on the required demand lead time. A plan and execute policy, for example, suits a long demand lead time, while a policy of continuous replenishment is more appropriate where demand lead time is short. SCG's best option for a plan and execute inventory policy under conditions of smooth demand is R,Q (see section 5.2.6), as it ensures sufficient inventory (R) and replenishment by quantity (Q) where R,Q values are set according to demand level and lead time. A demand flow policy, on the other hand, ensures that inventory is continuously replenished by ordering a quantity equal to the amount withdrawn.

Variable demand scenarios (responsive strategy)

Variable demand is characterized by uncertainty and a high level of variability. Since lumpy demand is considered the most uncertain and changeable type, and therefore the most difficult to satisfy, this was the scenario modelled in the simulation. The proper strategic response to this type of demand is agile or leagile, depending on the level of certainty and volumes involved, and the most appropriate inventory policies are postponement and quick response. Postponement suits uncertain, variable demand with a long lead time (SCG suggests T,S inventory policy for downstream sites and S,s policy for upstream sites), while quick response is more appropriate where customers are not willing to wait long for their orders. In this situation, SCG recommends base stock policy, which ensures continuous review of inventory levels and immediate inventory replenishment, allowing a rapid response to customer demand.

Impact of implemented strategy on SC sustainability

SC sustainability was measured in terms of CO2 emissions. In real life, CO2 emissions come from various sources, but as transportation is responsible for a large proportion of the total CO2 produced by SC activities, this was the only metric used in the simulation. SCG provides input data related to CO2 emission calculations; in the case study, CO2 was calculated based on weight-distance.

5.5 Results of modelling and simulation

5.5.1 Performance results from the baseline model

Table 5.12 depicts the current performance of the case study company, as reflected in the results from the baseline model.

Performance	Performance	Replications of Indicators Result					
Aspect	Aspect Indicator		2	3	4	5	Avg.
Efficiency	Profit to Revenue (%)	31.57	31.90	31.08	31.78	31.65	31.60
Responsiveness	Avg. Fill Qty Rate (%)	39.91	39.82	39.64	40.70	40.91	40.20
Sustainability	CO2 Emissions (kg)	734	721	739	731	726	730

Table 5. 12: Results from baseline model

5.5.2 Optimization of results from baseline model (predictable demand)

Table 5.13 shows various scenarios of predictable demand volume for the baseline model. These were modelled using normal distribution for different values of mean and standard deviation. The baseline mean was based on real demand data from a similar manufacturer, while the standard deviation was chosen to generate a smooth predictable demand pattern (see Figure 5.9). Based on the baseline model data, the other values assumed in order to generate increased and decreased demand scenarios.

Table 5. 13: Different scenarios of predictable demand volume

Demand Volume/ Distributors	DIS1	DIS2	DIS3
Baseline Demand	N(1000,70)	N(1000,50)	N(2000,150)
Increased Demand	N(2000,140)	N(2000,100)	N(4000,300)
Decreased Demand	N(500,35)	N(500,25)	N(1000,75)

Optimization of inventory parameters

The same scenarios were then optimized to find the optimized inventory parameters for each scenario (see Table 5.14).

		Baseline		Incre	ased			
Sites	Products	Der	mand	Dem	and	Decreased Demand		
		R	Q	R	Q	R	Q	
SUP1	R1	0	3796	0	3309	0	1898	
SUP2	R2	0	7592	0	6618	0	3796	
SUP3	R3	0	3796	0	3309	0	1898	
	Р	0	3796	0	3309	0	7213	
ΜΔΝ	R1	12252	14425	10688	12574	6126	3037	
WAN	R2	7662	6074	6686	5295	3831	3417	
	R3	6885	6833	6007	5956	3443	6833	

Table 5. 14: Optimized inventory parameters for different predictable demandscenarios

R: Reorder point Q: Reorder quantity

SUP3 - MAN

MAN - DIS1

MAN – DIS2

MAN – DIS3

Optimization of transportation assets

This optimization was carried out for a range of lead time scenarios where lead time was assumed to be restricted to transportation time. Table 5.15 shows the assumed lead time values.

	Baseline	Longer	Shorter	
Destination	Lead Time (BLT)	Lead Time (LLT)	Lead Time (SLT)	
	(days)	(days)	(days)	
SUP1- MAN	19	27	8	
SUP2 - MAN	4	6	2	

14

25

3

45

5

9

1

15

9

17

2

Table 5. 15: Different scenarios for transportation lead time

The optimized asset number was established for each lead time and demand volume scenario, as shown in Table 5.16.

Asset Name	Baseline Demand	Increased	Demand	Decreased Demand		
	BLT	LLT	SLT	LLT	SLT	
A1	83	62	48	22	43	
A2	30	27	15	6	16	
A3	2	2	5	3	5	
A4	21	13	17	9	10	
A5	9	4	6	3	5	
A6	48	27	34	15	20	

Table 5. 16: Optimized transportation asset numbers for different predictabledemand scenarios

Performance results from the optimized baseline model

Once the baseline model had been optimized in terms of inventory and transportation functions, the optimum values were fed into the model and a simulation run to get performance indicators (see Table 5.17).

Performance	Performance Replications of Indicators Result						
Aspect	Indicator	1	2	3	4	5	Avg.
Efficiency	Profit to Revenue (%)	31.57	31.90	31.08	31.78	31.64	31.78
Responsiveness	Avg. Fill Qty Rate (%)	45.28	46.31	45.79	46.10	45.89	45.87
Sustainability	CO2 Emissions (kg)	717	694	715	714	706	709

 Table 5. 17: Performance of optimized baseline model

Optimization of results under scenarios of unpredictable (lumpy) demand

The validation process also involved repeating the same steps for various scenarios of unpredictable demand. Table 5.18 shows the demand values for three scenarios. These were modelled using Bounded Normal (BN) distribution and Bounded Lognormal (BL) distribution. The distribution types and their related parameters including mean, standard deviation, minimum value, and maximum value were chosen in order to generate an unpredictable lumpy demand pattern (see Figure 5.9).

Demand			
Volume /	DIS1	DIS2	DIS3
Distributors			
Baseline	BL(2500,10000,0,	BN(2900,45000,0,	BN(2000,10000,0,
Demand	5000)	21000)	10000)
Increased	BL(5000,20000,0,	BN(5800,90000,0,	BN(4000,20000,0,
Demand	10000)	42000)	20000)
Decreased		BN(1450,22500,0,	
Demand	BL(1230,3000,0, 2300)	11500)	ыцтооо,зооо,о, зооо)

 Table 5. 18: Different scenarios of unpredictable demand volume

Optimization of inventory parameters

Table 5.19 shows the optimized R and Q values for a range of unpredictable demand scenarios.

Table 5. 19: Optimized inventory parameters for different unpredictable demand
scenarios

		Base	eline	Incre	ased	Decreased		
Sites	Products	Demand		Dem	Demand		Demand	
		R	Q	R	Q	R	Q	
SUP1	R1	0	5755	0	11510	0	2976	
SUP2	R2	0	11510	0	23019	0	5951	
SUP3	R3	0	5755	0	11510	0	2976	
	Р	2427	5755	4853	11510	1294	2976	
MAN	R1	53006	21868	106011	43735	27305	11306	
(Manufacturer)	R2	4853	14061	9705	28120	2587	7348	
	R3	23019	10359	46038	20717	11931	5356	

R: Reorder point Q: Reorder quantity

Optimization of transportation assets

Optimized transportation asset numbers for different unpredictable demand scenarios are shown in Table 5.20.

Asset	Bas	Baseline Demand			eased De	mand	Deci	reased D	emand
Name	BLT	LLT	SLT	BLT	LLT	SLT	BLT	LLT	SLT
A1	63	54	95	129	127	164	44	35	54
A2	28	24	27	19	40	44	18	16	18
A3	3	3	4	3	2	4	3	3	4
A4	12	12	11	13	12	16	4	9	10
A5	9	8	8	6	8	12	5	7	8
A6	20	18	21	25	21	39	19	17	24

Table 5. 20: Optimized transportation asset numbers for different unpredictabledemand scenarios

Performance results from the optimized unpredictable (lumpy) baseline model

Again, the optimized values were entered into the model and a simulation run to get performance indicators. These are shown in Table 5.21.

Table 5. 21: Performance of	optimized	l unpredict	able d	lemand	basel	line mod	el

Performance	Performance	Replications of Indicators Result						
Aspect	Indicator	1	2	3	4	5	Avg.	
Efficiency	Profit to Revenue (%)	40.57	40.69	41.02	34.55	37.90	38.94	
Responsiveness	Avg. Fill Qty Rate (%)	22.33	23.80	20.83	30.60	30.10	25.53	
Sustainability	CO2 Emissions (kg)	966	920	929	642	739	939	

5.5.3 Simulation results for efficient, responsive and sustainable SCD scenarios

This section presents the results of the simulations investigating different aspects of SC performance. Table 5.22 shows what happened when efficiency strategies (R,Q and demand flow) were implemented under six different scenarios of predictable demand, while Table 5.23 shows the results of implementing responsiveness strategies (T,S and base stock) under six different scenarios of unpredictable demand. Table 5.24 shows the results of the simulation investigating the environmental impact (i.e. the sustainability) of these strategies under the various scenarios. The mentioned tables present performance results by using the suggested policies within the developed framework, these results were summarized and 5.5.5 to validate presented in section the framework.

Scenario	Demand	Lead	Utilized	Performance	nance Replications of Indicators Result					
No	Volume	Time	Policy	Indicator	1	2	3	4	5	Avg.
1	Lower	SLT	R,Q	Profit to Revenue (%)	29.42	30.10	29.83	29.90	30.05	29.86
2	Base	BLT	R,Q	Profit to Revenue (%)	31.90	32.37	31.64	31.60	31.41	31.78
3	Higher	LLT	R,Q	Profit to Revenue (%)	32.00	32.87	32.68	32.35	31.94	32.37
4	Lower	LLT	Demand Flow	Profit to Revenue (%)	25.42	25.48	25.47	25.50	25.033	25.38
5	Base	BLT	Demand Flow	Profit to Revenue (%)	28.12	28.13	28.14	27.80	27.84	28.00
6	Higher	SLT	Demand Flow	Profit to Revenue (%)	30.60	30.60	30.79	30.39	30.52	30.58

Table 5. 22: Performance of efficiency strategies under scenarios of predictable demand

Scenario	Demand	Lead	Utilized	Performance	Replications of Indicators Result					
No	Volume	Time	Policy	Indicator	1	2	3	4	5	Avg.
1	Higher	SLT	T,S	Avg. Fill Qty Rate (%)	27.30	32.00	26.50	45.44	36.76	33.60
2	Base	BLT	T,S	Avg. Fill Qty Rate (%)	28.61	31.30	29.26	50.19	39.67	35.80
3	Lower	LLT	T,S	Avg. Fill Qty Rate (%)	31.14	34.56	32.63	46.26	40.87	37.10
4	Higher	LLT	Base Stock	Avg. Fill Qty Rate (%)	13.52	15.45	13.16	26.63	23.10	18.37
5	Base	BLT	Base Stock	Avg. Fill Qty Rate (%)	22.33	23.79	20.83	30.6	30.10	25.53
6	Lower	SLT	Base Stock	Avg. Fill Qty Rate (%)	31.75	32.78	30.47	33.68	35.23	32.78

Table 5. 23: Performance of responsiveness strategies under scenarios of unpredictable demand

Scenario	Stratogy	Utilized	Dem. Lead Performance				Replications of Indicators Result				
No	Strategy	Policy	Volume	Time	Indicator	1	2	3	4	5	Avg.
1			Lower SLT		602	593	597	591	591	595	
2		R,Q	Base	BLT	CO2 Emission (kg)	717	694	715	714	706	709
3	Efficiency		Higher	LLT		1016	1003	1027	1031	1017	1019
4	Linclency	Demand Flow	Lower	LLT		704	699	707	705	701	703
5			Base	BLT	CO2 Emission (kg)	870	861	876	871	867	869
6			Higher	SLT		1210	1190	1220	1202	1209	1206
7			Higher	SLT		1460	1383	1415	830	1024	1222
8		T,S	Base	BLT	CO2 Emission (kg)	907	867	879	595	692	788
9	Posnonsivonoss		Lower	LLT		659	637	641	501	541	596
10	Responsiveness	Base Stock	Higher	LLT		1532	1455	900	1475	1104	1293
11			Base	BLT	CO2 Emission (kg)	966	920	929	642	739	839
12			Lower	SLT		678	670	682	524	572	625

Table 5. 24: Sustainability performance of efficiency and responsiveness strategies under different scenarios of predictable andunpredictable demand

5.5.4 Discussion of results

This section discusses the results presented in the previous section. Table 5.10 shows the results of the demand analysis for the baseline model. This is smooth, as demand is non-intermittent (ADI < 1.32) (see Figure 5.8). Figures 5.10 through to 5.14 show different demand classes that were modelled to simulate various SC designs.

Table 5.14 shows the optimized reorder point (R) and reorder quantity (Q) for different predictable demand scenarios. It can be observed that the inventory cost of raw material has moved downstream at MAN site side, allowing suppliers to be more efficient and rendering MAN more responsive. Table 5.14 also shows that the values of R and Q generally decrease when demand volume decreases but not when it increases, suggesting that the optimum solution is sensitive to the former but not the latter. It can be observed from Table 5.19 that optimized values of R and Q are generally proportional to demand volume, given that if demand increases, ROP and Q also increase so as to mitigate uncertainty risks. In terms of optimized transaction assets, Tables 5.16 and 5.20 show that asset number is generally proportional to demand volume but inversely proportional to lead time. Where lead time is short, it may be necessary to increase the asset number, and therefore the cost, in order to respond to customer demand.

Table 5.17 shows that performance is better in the optimized model than in the baseline model. This is especially the case with responsiveness performance, where avg. fill qty rate increases from 40% up to 45%. This value can be improved even further if the SC can bear higher cost and reduced overall efficiency. Against this backdrop, it is important that SC managers consider various scenarios and select the one which best meets the SC strategy and requirements of their enterprise. It is perhaps surprising that efficiency was slightly higher in the optimized unpredictable demand baseline model (see Table 5.21) than in the optimized predictable demand model (Table 5.17). One explanation for this is that in the unpredictable demand model, a large enough increase in demand volume can generate enough revenue to offset fixed costs. On the other hand, service level is

lower in the unpredictable demand model because of the higher level of uncertainty.

5.5.5 Validation of the proposed framework

The framework suggests that the adoption of lean strategy through the implementation of a plan and execute policy is the best approach where demand is predictable, volume is high and lead time is long. The output of the simulation supports this; as indicated in Table 5.25, efficiency improves significantly as demand volume and lead time increase.

Table 5. 25: Profit to revenue (%) under lean strategy, plan and execute policy(R,Q)

Demand volume/Lead time	Lower volume	Baseline volume	Higher volume
Shorter lead time	29.86		
Baseline lead time		31.78	
Longer lead time			32.37

In instances where demand is predictable, volume is high but lead time is short, the framework suggests that a continuous replenishment policy is a more suitable lean strategy. This is again supported by the simulation, which shows efficiency is improving when the demand volume is higher and the lead time is shorter (see Table 5.26).

Table 5. 26: Profit to revenue (%) under lean strategy, continuous replenishmentpolicy (demand flow)

Demand volume/Lead time	Lower volume	Baseline volume	Higher volume
Longer lead time	25.38		
Baseline lead time		28.00	
Shorter lead time			30.58

Where demand is unpredictable, volume is low and the lead time is long, the framework suggests that the optimal approach is to adopt a leagile strategy such as postponement. This finding is in line with the simulation outputs, which show responsiveness is improving as demand volume declines and lead time extends.

Demand volume/Lead time	Higher volume	Baseline volume	Lower volume
Shorter lead time	33.60		
Baseline lead time		35.80	
Longer lead time			37.10

Table 5. 27: Avg. fill qty rate (%) under leagile strategy, postponement policy(T,S)

Where demand is unpredictable, volume is low and the lead time is short, the framework suggests that an agile strategy, implemented in the form of a quick response policy, is the best way to optimize SC performance. The simulation supports this, showing that responsiveness improves as demand volume declines and lead time shortens (see Table 5.28).

Table 5. 28: Avg. fill qty rate (%) under agile strategy, quick response policy
(base stock)

Demand volume/Lead time	Lower volume	Baseline volume	Higher volume
Shorter lead time	32.78		
Baseline lead time		25.53	
Longer lead time			18.37

Table 5.24 indicates that policies oriented towards improving efficiency tend to reduce the sustainability of the SC, while those policies that are geared towards responsiveness tend to have the opposite effect. These results are in alignment with the proposed framework's assumption that the search for efficiency negatively affects sustainability, but that this is not the case when implementing responsive strategy.

5.6 Conclusions

This chapter described how the SCG program was employed to validate the proposed framework. It discusses the steps involved, including the creation of a baseline model describing the current performance of the case study company, the use of the program's SSO and TO functions to optimize the

model, and the simulations that were run using the optimized model to assess SC performance under a range of scenarios and policies. The chapter concluded by presenting the results of these simulations and showing how they validate the developed framework.

Chapter 6: Conclusions, Recommendations, Contributions to knowledge and Future work

6.1 Introduction

Previous SCD research has focused on developing frameworks that aim to improve some rather than all aspects of SC performance. This framework takes a more integrated approach to SCD by incorporating the strategy, process and resource dimensions, thereby enabling the assessment and improvement of SC performance across multiple areas. This chapter summarizes the findings of the research before discussing how it contributes to knowledge and offering suggestions for further study.

6.2 Summary of Findings

No consensus among authors on what SCD means

Although the concept of SCD has been known since 1998, no one has yet produced a widely accepted definition of either the concept or its scope. Much research has been carried out in this field, most of which has historically focused on questions such as facility location, and whether SCD should consider only operational dimensions such as scheduling and resource allocation, or whether it should also encompass strategic issues. Recently, however, the scope of SCD research has extended to cover service requirements, SC security, risk and sustainability. A number of researchers have discussed SCND from the perspective of process, investments and structure.

• There is no single framework that addresses all three dimensions of efficiency, responsiveness and sustainability

SCs must reconcile the competing pressures of responsiveness, which usually incurs higher cost, and cost-efficiency, which is often achieved at the expense of market responsiveness. Efficiency is also likely to be impacted by the growing pressure to meet "green" demands and operate sustainably. Any attempt at optimization must therefore consider all three dimensions simultaneously, but so far no one framework has been developed to improve all three performance aspects. The closest so far has been Ambe and Badenhorst-Weiss's framework, which considers efficiency and responsiveness.

No integrated framework has been developed to tie the important aspects of SCD together

The literature review demonstrates that no one has yet produced a framework integrating the strategy, process and network aspects of SCD. The framework proposed in this study comprises a strategic model, a process model and a network model, each of which performs a defined but complementary role. Together, they offer an integrated model of SCD.

Need for the development of a simulation-based framework in strategic SCD

Numerous researchers have employed simulation tools to investigate SCs at the operational level, examining areas such as inventory management, production planning and management, performance measurement, location and transportation, warehouse operations and process improvement. At the strategic level, however, this approach has only been used to investigate complex adaptive supply networks (CASN) and the impact of SC integration on responsiveness. The current research responds to this gap by developing a simulation-based framework that integrates SC strategy, process and resources in one applicable template.

The proposed framework contains a strategic model that offers strategies for different demand characteristics and market environments

A few models have been designed that aim to assist SCs in achieving multiple objectives, but these models do not differentiate between market environments or give any guidance on the best strategy for any given objective. Christopher's model (2006), upon which the strategic model proposed in this framework was based, suggests different strategies for different environments, but it remains conceptual; it was not implemented or validated.

 SCOR model has been utilized to represent the process element within the framework since it identifies policies for different business environments

The strategic model recommends the strategy that is most appropriate given the desired objectives and the prevailing business environment. SCOR suggests the most suitable policies and process configurations to complete the sourcing, making and delivering functions in such a way as to achieve this strategy.

 The framework utilizes Long's model to design a multiple-objective network that integrates with SCOR to achieve different process policies

To facilitate integration, the network model was designed to allow the network to be operated to implement SCOR policies; each entity in the SCN (e.g. supplier, manufacturer, distributor, retailer) can be modelled using the suggested process to ensure that SC resources are employed in accordance with the chosen policies and strategy.

• The results of the modelling and simulation validate the framework:

- The optimized inventory model (see Table 5.14) demonstrates that the optimum strategies for the case study SC are lean strategy for supplier sites (allowing it to keep stock levels at a minimum) and agile strategy for the manufacturer (allowing it to keep enough stock to immediately satisfy any unpredictable demand). The SC can adopt a leagile strategy by decoupling the supplier and manufacturing sites. It can attain efficiency and responsiveness by building an agile response upon a lean platform that is, by following lean principles up to the decoupling point and agile practices after that point.
- Transportation asset number is proportional to demand volume and inversely proportional to lead time (see Tables 5.16 and 5.20). In Table 5.16, where demand is predictable, total transportation cost

increases with demand volume, but cost per unit decreases because assets are being utilized more efficiently. This case supports the adaptation of an efficient strategy, as suggested in the proposed framework. On the other hand, where demand is unpredictable (see Table 5.20), a responsive strategy is advised; the aim should be to achieve short lead time, regardless of the increase in transportation cost.

- Table 5.21 shows that efficiency is slightly higher in the unpredictable demand scenario than in the predictable demand scenario. This unexpected result may be explained by the fact that an increase in the volume of unpredictable demand can generate significant revenue, which can potentially offset fixed costs. This opportunity is less likely in the predictable demand scenario.
- Verification and validation of the proposed framework was a vital part of this project, as the findings of the experiment would have been useless if the framework did not perform as expected. All the results obtained support the framework:
 - Efficiency improves significantly when demand volume is high and lead time is long (Table 5.25).
 - Efficiency also improves when demand volume is high and lead time is short (Table 5.26).
 - Responsiveness improves when demand volume is low and lead time is long (Table 5.27).
 - Responsiveness also improves when demand volume is low and lead time is short (Table 5.28).
 - An efficient strategy reduces sustainability, while a responsive strategy improves sustainability (Table 5.24).

6.3 Contributions to Knowledge

The proposed SCD framework integrates strategy, process and resources and allows the use of simulation tools to investigate efficiency, responsiveness and sustainability concurrently during the design process. Demand-driven, it assumes that SCs are designed to satisfy certain demand characteristics. The proposed framework comprises a strategic model, a process model, a network model and a performance model, each of which performs a defined role but integrates with the others.

The strategic objective model seeks to identify the most effective way of meeting customer requirements, taking into account SC capabilities and uncertainty. Its main strategic offerings are lean, agile and leagile strategies. Where demand is predictable, SCs are advised to adopt a lean strategy; if lead time is short, they may implement a continuous replenishment policy of replacing products as they are sold or used, but where lead time is longer, a plan and execute policy may be more appropriate.

In either scenario, the process model suggests that companies following lean strategy should adopt a make-to-stock policy in which processes are configured so as to reduce costs and make the maximum use of available resources. SC capabilities tend to be pre-planned for long periods of time, unvaried, and fixed with no excess capacity. Since SC functions must operate within these capability restrictions, production is massive and standard, inventory is high level and unvaried, and transportation processes utilize low cost modes (e.g. Full TL) and seek to reduce transport time as much as possible.

Conversely, the aim in agile strategy is to satisfy unpredictable customer demand with short lead times and perfect fulfilment. In this case, the process model suggests a make-to-order policy in which resources are variable, varied and excess. Production policy is characterized by product variety, low product volume and short process time. Inventory is kept low but varied, postponement or quick response policies are the norm, and transportation modes are fast and flexible. Lastly, in the leagile strategy option, the process model suggests a make/assemble-to-order policy. This is a combination of make-to-stock and make-to-order policies.

The third component within the framework is the network model. This has four levels: the SCN level, the enterprise element level, the workshop element level and the production unit element level. All of these resources and the associated decision making (e.g. regarding locations, capacities and technologies) are directed towards achieving the strategic objectives of the SC using the policies suggested by the process model.

The last component in the framework is the performance model, which aims to measure SC performance and show the extent to which the SC is achieving its objectives of efficiency, responsiveness and sustainability.

6.4 Limitations and Future Work

The primary limitation of the study was the lack of real-life data about existing SCs, as this prevented the researcher from running further simulations to investigate other aspects of SCD. SCG's newness to the market was also a problem in that firstly, it affected some of the research modelling tasks (e.g. it was not possible to use more than one probability distribution to model a certain type of demand as this function is still under development) and secondly, a considerable proportion of the available research time had to be spent learning to master the software to the necessary level.

As far as future study is concerned, the lack of a common research approach to SCD raises questions about the extent to which this research is consonant with SCD practice in industry. Investigation is required of the extent to which research is supporting real-life practice, and the factors that shape this practice.

The literature review shows that simulation is rarely employed by researchers investigating SCs at the strategic level, so more research is needed to determine whether simulation-based research is making any contribution to SC practice, especially at this level, where simulation is vital. The modelling and simulation work done in this research reveals a number of ways in which the application of SCG could be extended:

 To apply the concept of strategic fit by examining empirically the optimum strategy each SC member should adopt to improve overall SC performance.

- To model different process configurations by designing SC capabilities to implement specific strategies (see Table 4.1).
- To identify the best SCN configurations in terms of centralization and decentralization strategy and business location (global, local or both) (see Tables 4.2 and 4.3).
- To further explore the interaction between the framework models, such as how process policies assist in achieving SC strategy, and the impact of different SCN configurations and decoupling point locations on strategy achievement.

References

ADHITYA, Arief, HALIM, Iskandar and SRINIVASAN, Rajagopalan (2011). Decision Support for Green Supply Chain Operations by Integrating Dynamic Simulation and LCA Indicators: Diaper Case Study. Environmental science & technology, 45 (23), 10178-10185.

AFFONSO, R. C., LIU, Y. and ZOLGHADRI, M. (2013). A global framework of activities integrating product & supply chain design. [online]. IFAC Proceedings Volumes, 46 (24), 505-510. http://www.sciencedirect.com.lcproxy.shu.ac.uk/science/article/pii/S1474667 016322388.

ALFALLA-LUQUE, Rafaela, MEDINA-LOPEZ, Carmen and DEY, Prasanta Kumar (2013). Supply chain integration framework using literature review. Production planning & control, 24 (8-9), 800-817.

AMBE, Intaher M. (2012). Determining an optimal supply chain strategy. Journal of transport and supply chain management, 6 (1), 126-147.

AMBE, Intaher M. and BADENHORST-WEISS, Johanna A. (2011). Framework for choosing supply chain strategies. African journal of business management, 5 (35), 13388-13397.

AMBE, I. M. (2014). Key indicators for optimising supply chain performance: The case of light vehicle manufacturers In South Africa. Journal of applied business research, 30 (1), 277-289.

AMER, LE and ELTAWIL, AB COLLABORATIVE SUSTAINABLE SUPPLY CHAIN NETWORK DESIGN: STATE OF THE ART AND SOLUTION FRAMEWORK.

ARONSSON, Håkan, ABRAHAMSSON, Mats and SPENS, Karen (2011). Developing lean and agile health care supply chains. Supply chain management, 16 (3), 176-183.

ARZU AKYUZ, Goknur and ERMAN ERKAN, Turan (2010). Supply chain performance measurement: a literature review. International journal of production research, 48 (17), 5137-5155.

BAGHALIAN, Atefeh, REZAPOUR, Shabnam and FARAHANI, Reza Zanjirani (2013). Robust supply chain network design with service level against disruptions and demand uncertainties: A real-life case. European journal of operational research, 227 (1), 199-215.

BAUD-LAVIGNE, Bertrand, AGARD, Bruno and PENZ, Bernard (2012). Mutual impacts of product standardization and supply chain design. International journal of production economics, 135 (1), 50-60.

BAUD-LAVIGNE, Bertrand, AGARD, Bruno and PENZ, Bernard (2012). Mutual impacts of product standardization and supply chain design. International journal of production economics, 135 (1), 50-60.

BAUMGARTNER, Kerstin, FUETTERER, André and THONEMANN, Ulrich W. (2012). Supply chain design considering economies of scale and transport frequencies. European journal of operational research, 218 (3), 789-800.

BAYAT, Ayse, et al. (2011). Sustainably driven supply chains. Industrial engineer, 43 (8), 26-31.

BEAMON, Benita M. (1998). Supply chain design and analysis:: Models and methods. International journal of production economics, 55 (3), 281-294.

BEHROUZI, F., WONG, K. Y. and BEHROUZI, F. (2011). A study on lean supply chain performance measures of SMEs in the automotive industry. In: IEEE international conference on industrial engineering and engineering management, 237-241.

BIRHANU, Dagne, LANKA, Krishnanand and RAO, A. Neelakanteswara (2014). A survey of classifications in supply chain strategies. Procedia engineering, 97, 2289-2297.

BOLSTORFF, Peter and ROSENBAUM, Robert G. (2007). Supply chain excellence: a handbook for dramatic improvement using the SCOR model. AMACOM/American Management Association.

BRAZIOTIS, Christos, et al. (2013). Supply chains and supply networks: distinctions and overlaps. Supply chain management, 18 (6), 644-652.

BÜYÜKÖZKAN, Gülçin and BERKOL, Çiğdem (2011). Designing a sustainable supply chain using an integrated analytic network process and goal programming approach in quality function deployment. Expert systems with applications, 38 (11), 13731-13748.

CABRAL, Izunildo, et al. (2011). An information model in lean, agile, resilient and green supply chains. In: Communication software and networks (ICCSN), 2011 IEEE 3rd international conference on, IEEE, 776-780.

Candace Y. Yi, E.W.T. Ngai and K-L. Moon (2011). Supply chain flexibility in an uncertain environment: exploratory findings from five case studies. Supply chain management, 16 (4), 271-283.

CARTER, Craig R. and ROGERS, Dale S. (2008). A framework of sustainable supply chain management: moving toward new theory. International journal of physical distribution & logistics management, 38 (5), 360-387.

CHAABANE, Amin, RAMUDHIN, Amar and PAQUET, Marc (2012). Design of sustainable supply chains under the emission trading scheme. International journal of production economics, 135 (1), 37-49.

CHAHARSOOGHI, S. Kamal and HEYDARI, Jafar (2011). Strategic Fit in Supply Chain Management: A Coordination Perspective. INTECH Open Access Publisher.

CHANG, Yoon and MAKATSORIS, Harris (2001). Supply chain modeling using simulation. International journal of simulation, 2 (1), 24-30.

CHARDINE-BAUMANN, Emilie and BOTTA-GENOULAZ, Val (2014). A framework for sustainable performance assessment of supply chain management practices. Computers & industrial engineering, 76, 138-147.

CHOPRA, Sunil and MEINDL, Peter (2007). Supply chain management. Strategy, planning & operation. Springer.

CHRISTOPHER, Martin (2005). Logistics and supply chain management: creating value-added networks. Pearson education.

CHRISTOPHER, Martin, PECK, Helen and TOWILL, Denis (2006). A taxonomy for selecting global supply chain strategies. The international journal of logistics management, 17 (2), 277-287.

CLOSS, David J. and MCGARRELL, Edmund F. (2004). Enhancing security throughout the supply chain. IBM Center for the Business of Government Washington, DC.

COLICCHIA, Claudia, DALLARI, Fabrizio and MELACINI, Marco (2010). Increasing supply chain resilience in a global sourcing context. Production planning & control, 21 (7), 680-694.

CONTRERAS, Homero H., et al. (2012). A Strategic and Standardized Simulation Model for Distribution Networks. In: IIE annual conference. proceedings, Institute of Industrial Engineers-Publisher, 1.

COROMINAS, Albert, et al. (2015). Methodological elements of supply chain design. International journal of production research, (ahead-of-print), 1-14.

COROMINAS, A., et al. (2015). Methodological elements of supply chain design. International journal of production research, 53 (16), 5017-5030.

COUNCIL, Supply Chain (2012). Supply chain operations reference model: Revision 11.0. Chicago, IL, . DAABOUL, Joanna, et al. (2014). Value networks: pulling the triggers. A combined approach of modelling and simulation for performance evaluation. International journal of computer integrated manufacturing, 27 (7), 609-623.

DANESE, Pamela and ROMANO, Pietro (2011). Supply chain integration and efficiency performance: a study on the interactions between customer and supplier integration. Supply chain management: An international journal, 16 (4), 220-230.

DANESE, Pamela, ROMANO, Pietro and FORMENTINI, Marco (2013). The impact of supply chain integration on responsiveness: The moderating effect of using an international supplier network. Transportation research: Part E, 49 (1), 125-140.

DE WAAL, André and KERKLAAN, Leo (2010). A performance management readiness review framework for governmental service providers. Business horizons, 53 (4), 405-412.

DU TOIT, Deirdre and VLOK, Pieter-Jan (2014). Supply chain management: A framework of understanding. South african journal of industrial engineering, 25 (3), 25-38.

DUBEY, Rameshwar, GUNASEKARAN, Angappa and CHILDE, Stephen J. (2015). The design of a responsive sustainable supply chain network under uncertainty. The international journal of advanced manufacturing technology, , 1-19.

DURGA PRASAD, KG, VENKATA SUBBAIAH, K. and NARAYANA RAO, K. (2014). Supply chain design through QFD-based optimization. Journal of manufacturing technology management, 25 (5), 712-733.

EL-AAL, Mohammad Abd, et al. (2008). SUPPLY CHAIN MANAGEMENT SIMULATION TYPES: A. Journal of engineering sciences, assiut university, 36 (3), 675-687.
ELROD, Cassandra, MURRAY, Susan and BANDE, Sundeep (2013). A Review of Performance Metrics for Supply Chain Management. Engineering management journal, 25 (3).

ELTAWY, Nesrine and GALLEAR, David (2017). Leanness and agility: a comparative theoretical view. Industrial management & data systems, 117 (1), 149-165.

ERTEM, Mustafa A., BUYURGAN, Nebil and ROSSETTI, Manuel D. (2010). Multiple-buyer procurement auctions framework for humanitarian supply chain management. International journal of physical distribution & logistics management, 40 (3), 202-227.

FABBE-COSTES, Nathalie, et al. (2014). Sustainable supply chains: a framework for environmental scanning practices. International journal of operations & production management, 34 (4), 664-694.

FARAHANI, Reza Zanjirani, et al. (2014). Competitive supply chain network design: An overview of classifications, models, solution techniques and applications. Omega, 45, 92-118.

FAWCETT, Stanley E., ELLRAM, Lisa M. and OGDEN, Jeffrey A. (2014). Supply chain management: from vision to implementation. Pearson London.

FAWCETT, Stanley E. and WALLER, Matthew A. (2014). Supply Chain Game Changers-Mega, Nano, and Virtual Trends-And Forces That Impede Supply Chain Design (i.e., Building a Winning Team). Journal of business logistics, 35 (3), 157-164.

FAYEZI, Sajad, ZUTSHI, Ambika and O'LOUGHLIN, Andrew (2014). Developing an analytical framework to assess the uncertainty and flexibility mismatches across the supply chain. Business process management journal, 20 (3), 362-391. FERREIRA, Luciano and BORENSTEIN, Denis (2011). Normative agentbased simulation for supply chain planning. Journal of the operational research society, 62 (3), 501-514.

FISHER, Marshall L. (1997). What is the right supply chain for your product? Harvard business review, 75, 105-117.

FITKOV-NORRIS, Elena (2010). Literature overview of approaches for enterprise-wide modelling, simulation and optimisation.

FORBES, Catherine, et al. (2011). Statistical distributions. John Wiley & Sons.

FUMERO, Yanina, CORSANO, Gabriela and MONTAGNA, Jorge M. (2013). Integrated modeling framework for supply chain design considering multiproduct production facilities. Industrial & engineering chemistry research, 52 (46), 16247-16266.

GAN, Thiam-Soon and GRUNOW, Martin (2013). Concurrent Product – Supply Chain Design: A Conceptual Framework & Literature Review. [online]. Procedia CIRP, 7 (Supplement C), 91-96. http://www.sciencedirect.com.lcproxy.shu.ac.uk/science/article/pii/S2212827 113002230.

GARCIA, Daniel J. and YOU, Fengqi (2015). Supply chain design and optimization: Challenges and opportunities. Computers & chemical engineering, .

GEORGIADIS, Michael C., et al. (2011). Optimal design of supply chain networks under uncertain transient demand variations. Omega, 39 (3), 254-272.

GHOSH, Anupam, DAS, Sujoy and DESHPANDE, Aashish (2014). Effect of Responsiveness and Process Integration in Supply Chain Coordination. IUP journal of supply chain management, 11 (1), 7-17.

GIANNOCCARO, Ilaria (2011). Assessing the influence of the organization in the supply chain management using NK simulation. International journal of production economics, 131 (1), 263-272.

GOFF, JA, OLSON, HC and DUNCAN, CS (2000). Correlation of side-scan backscatter intensity with grain-size distribution of shelf sediments, New Jersey margin. Geo-marine letters, 20 (1), 43-49.

GOPAL, PRC and THAKKAR, Jitesh (2012). A review on supply chain performance measures and metrics: 2000-2011. International journal of productivity and performance management, 61 (5), 518-547.

GOSLING, Jonathan, NAIM, Mohamed and TOWILL, Denis (2013). A supply chain flexibility framework for engineer-to-order systems. Production planning & control, 24 (7), 552-566.

GOVINDAN, Kannan, FATTAHI, Mohammad and KEYVANSHOKOOH, Esmaeil (2017). Supply chain network design under uncertainty: A comprehensive review and future research directions. [online]. European Journal of Operational Research, 263 (1), 108-141. http://www.sciencedirect.com.lcproxy.shu.ac.uk/science/article/pii/S0377221 717303429.

GRACIA, María D. and QUEZADA, Luis E. (2016). A framework for strategy formulation in sustainable supply chains: a case study in the electric industry. NETNOMICS: Economic research and electronic networking, 17 (1), 3-27.

GUMUS, Alev Taskin, GUNERI, Ali Fuat and KELES, Selcan (2009). Supply chain network design using an integrated neuro-fuzzy and MILP approach: A comparative design study. Expert systems with applications, 36 (10), 12570-12577.

GUMUS, Alev Taskin, GUNERI, Ali Fuat and ULENGIN, Fusun (2010). A new methodology for multi-echelon inventory management in stochastic and neuro-fuzzy environments. International journal of production economics, 128 (1), 248-260. GUNASEKARAN, A., PATEL, C. and MCGAUGHEY, Ronald E. (2004). A framework for supply chain performance measurement. [online]. International Journal of Production Economics, 87 (3), 333-347. http://www.sciencedirect.com.lcproxy.shu.ac.uk/science/article/pii/S0925527 303002561.

HASSINI, Elkafi, SURTI, Chirag and SEARCY, Cory (2012). A literature review and a case study of sustainable supply chains with a focus on metrics. International journal of production economics, 140 (1), 69-82.

HILLETOFTH, Per and LÄTTILÄ, Lauri (2012). Agent based decision support in the supply chain context. Industrial management & data systems, 112 (8), 1217-1235.

HILLETOFTH, Per (2012). Differentiation focused supply chain design. Industrial management & data systems, 112 (9), 1274-1291.

HOFFMAN, Julien I. E. (2015). Chapter 18 - The Poisson Distribution.[online].AcademicPress.259-278.http://www.sciencedirect.com/science/article/pii/B9780128023877000184.

HOLWEG, Matthias and HELO, Petri (2014). Defining value chain architectures: Linking strategic value creation to operational supply chain design. International journal of production economics, 147, 230-238.

HONG, Paul, et al. (2012). Extending the efficient and responsive supply chains framework to the green context. Benchmarking: An international journal, 19 (4/5), 463-480.

HUAN, Samuel H., SHEORAN, Sunil K. and WANG, Ge (2004). A review and analysis of supply chain operations reference (SCOR) model. Supply chain management: An international journal, 9 (1), 23-29.

HUO, Yanfang, et al. (2009). A framework and key techniques for supply chain integration. In: Supply Chain the Way to Flat Organisation. InTech, .

ISOTALO, Jarkko (2001). Basics of statistics. Finland: University of tampere, .

IVANOV, Dmitry (2010). An adaptive framework for aligning (re) planning decisions on supply chain strategy, design, tactics, and operations. International journal of production research, 48 (13), 3999-4017.

JABAREEN, Yosef (2009). Building a conceptual framework: philosophy, definitions, and procedure. International journal of qualitative methods, 8 (4), 49-62.

JAYA KRISHNA, S. (2011). Supply Chain Collaboration: Evolution Management Framework. International journal of global business, 4 (1), 23-43.

JAYARAM, Jayanth, DIXIT, Mita and MOTWANI, Jaideep (2014). Supply chain management capability of small and medium sized family businesses in India: A multiple case study approach. International journal of production economics, 147, 472-485.

JOTHIMANI, D. and SARMAH, S. P. (2014). Supply chain performance measurement for third party logistics. Benchmarking, 21 (6), 944-963.

KHAN, Muhammad Burdbar and SHEIKH, Naeem Nisar (2012). Human resource development, motivation and Islam. Journal of management development, 31 (10), 1021-1034.

KIM, Dong-Young, KUMAR, Vinod and KUMAR, Uma (2010). Performance assessment framework for supply chain partnership. Supply chain management: An international journal, 15 (3), 187-195.

KIM, Bowon (2013). Competitive priorities and supply chain strategy in the fashion industry. Qualitative Market Research: An International Journal, 16 (2), 214-242.

KISPERSKA-MORON, Danuta and DE HAAN, Job (2011). Improving supply chain performance to satisfy final customers: "Leagile" experiences of a

polish distributor. International journal of production economics, 133 (1), 127-134.

KLIBI, Walid, MARTEL, Alain and GUITOUNI, Adel (2010). The design of robust value-creating supply chain networks: a critical review. European journal of operational research, 203 (2), 283-293.

KOCAOĞLU, Batuhan, GÜLSÜN, Bahadır and TANYAŞ, Mehmet (2013). A SCOR based approach for measuring a benchmarkable supply chain performance. Journal of intelligent manufacturing, 24 (1), 113-132.

KOTHARI, Chakravanti Rajagopalachari (2004). Research methodology: Methods and techniques. New Age International.

KRISTIANTO, Yohanes, et al. (2012). A decision support system for integrating manufacturing and product design into the reconfiguration of the supply chain networks. Decision support systems, 52 (4), 790-801.

KUMAR, Sameer, SPENCE, Wendell and SWARTZER, Philip (2012). Lessons for strategic organizational growth - performance challenges for U.S. automakers. Human systems management, 31 (3), 203-218.

LAMBERT, Douglas M., GARCÍA- DASTUGUE, Sebastián J. and CROXTON, Keely L. (2005). An evaluation of process- oriented supply chain management frameworks. Journal of business logistics, 26 (1), 25-51.

LAMBIASE, Alessandro, et al. (2013). Strategic planning and design of supply chains: A literature review. Int.J.eng.bus.manag, 5 (49), 1-11.

LEE, Hau L. (2002). Aligning supply chain strategies with product uncertainties. California management review, 44 (3), 105-119.

LEONCZUK, Dorota (2016). Categories of Supply Chain Performance Indicators: an Overview of Approaches. Business, management and education, 14 (1), 103-115.

LEUKEL, Joerg and SUGUMARAN, Vijayan (2013). Formal correctness of supply chain design. Decision support systems, 56, 288-299.

LEUKEL, Joerg and SUGUMARAN, Vijayan (2013). Formal correctness of supply chain design. Decision support systems, 56, 288-299.

LI, Gang, et al. (2010). The evolutionary complexity of complex adaptive supply networks: a simulation and case study. International journal of production economics, 124 (2), 310-330.

LICHOCIK, Grzegorz and SADOWSKI, Adam (2013). EFFICIENCY OF SUPPLY CHAIN MANAGEMENT. STRATEGIC AND OPERATIONAL APPROACH. Logforum, 9 (2), 119-125.

LIN, Cheng-Chang and WANG, Tsai-Hsin (2011). Build-to-order supply chain network design under supply and demand uncertainties. Transportation research part B: Methodological, 45 (8), 1162-1176.

LONG, Qingqi (2014). Distributed supply chain network modelling and simulation: integration of agent-based distributed simulation and improved SCOR model. International journal of production research, 52 (23), 6899-6917.

LU, Dawei (2011). Fundamentals of supply chain management. Bookboon.

LYU, JrJung, DING, Jyh-Hong and CHEN, Ping-Shun (2010). Coordinating replenishment mechanisms in supply chain: From the collaborative supplier and store-level retailer perspective. International journal of production economics, 123 (1), 221-234.

MAK, Ho-Yin and SHEN, Zuo-Jun (Max) (2012). Risk diversification and risk pooling in supply chain design. IIE transactions, 44 (8), 603-621.

MALLIDIS, Ioannis, DEKKER, Rommert and VLACHOS, Dimitrios (2012). The impact of greening on supply chain design and cost: a case for a developing region. Journal of transport geography, 22, 118-128.

MALLIDIS, Ioannis, DEKKER, Rommert and VLACHOS, Dimitrios (2012). The impact of greening on supply chain design and cost: a case for a developing region. Journal of transport geography, 22, 118-128.

MANSOORNEJAD, Behrang, PISTIKOPOULOS, Efstratios N. and STUART, Paul R. (2013). Scenario-based strategic supply chain design and analysis for the forest biorefinery using an operational supply chain model. International journal of production economics, 144 (2), 618-634.

MARCHESINI, M. M. P. and ALCÂNTARA, R. L. C. (2016). Logistics activities in supply chain business process: A conceptual framework to guide their implementation. International journal of logistics management, 27 (1), 6-30.

MATOPOULOS, Aristides, BARROS, Ana Cristina and VORST, J. G. A. J. (J (2015). Resource-efficient supply chains: a research framework, literature review and research agenda. Supply chain management, 20 (2), 218-236.

MELNYK, Steven A., NARASIMHAN, Ram and DECAMPOS, Hugo A. (2014). Supply chain design: issues, challenges, frameworks and solutions. International journal of production research, 52 (7), 1887-1896.

MENTZER, John T., et al. (2001). Defining supply chain management. Journal of business logistics, 22 (2), 1-25.

METTA, Haritha and BADURDEEN, Fazleena (2013). Integrating sustainable product and supply chain design: modeling issues and challenges. Engineering management, IEEE transactions on, 60 (2), 438-446.

METTA, Haritha and BADURDEEN, Fazleena (2013). Integrating Sustainable Product and Supply Chain Design: Modeling Issues and Challenges. IEEE transactions on engineering management, 60 (2), 438-446.

MIN, Hokey and ZHOU, Gengui (2002). Supply chain modeling: past, present and future. Computers & industrial engineering, 43 (1-2), 231-249.

Ming-Hon Hwang (2010). Establishment of a comprehensive framework for strategic supply chain management. Human systems management, 29 (3), 127-137.

MISHRA, Rohita Kumar (2012). Measuring supply chain efficiency: a DEA approach. J oper supply chain manag, 5 (1), 45-68.

MITTERMAYER, Herwig and RODRÍGUEZ-MONROY, Carlos (2013). Evaluating alternative industrial network organizations and information systems. Industrial management & data systems, 113 (1), 77-95.

MOHAMMADI, Mohsen (2017). Challenges of Business Process Modeling in Logistics and Supply Chain Management. International journal of computer science and network security, 17 (6), 259-265.

MONCAYO-MARTÍNEZ, Luis A. and ZHANG, David Z. (2011). Multiobjective ant colony optimisation: A meta-heuristic approach to supply chain design. International journal of production economics, 131 (1), 407-420.

MORTENSEN, Mads and ARLBJØRN, Jan (2012). Inter-organisational supplier development: the case of customer attractiveness and strategic fit. Supply chain management: An international journal, 17 (2), 152-171.

MULA, Josefa, et al. (2013). A system dynamics model for the supply chain procurement transport problem: Comparing spreadsheets, fuzzy programming and simulation approaches. International journal of production research, 51 (13), 4087-4104.

NASLUND, Dag and WILLIAMSON, Steven (2010). What is management in supply chain management?-a critical review of definitions, frameworks and terminology. Journal of management policy and practice, 11 (4), 11-28.

NEPAL, Bimal, MONPLAISIR, Leslie and FAMUYIWA, Oluwafemi (2012). Matching product architecture with supply chain design. European journal of operational research, 216 (2), 312-325.

OH, Seungjin, RYU, Kwangyeol and JUNG, Mooyoung (2013). Reconfiguration framework of a supply network based on flexibility strategies. Computers & industrial engineering, 65 (1), 156-165.

OLIVARES-BENITEZ, Elias, RÍOS-MERCADO, Roger Z. and GONZÁLEZ-VELARDE, José Luis (2013). A metaheuristic algorithm to solve the selection of transportation channels in supply chain design. International journal of production economics, 145 (1), 161-172.

PALMA-MENDOZA, Jaime (2014). Analytical hierarchy process and SCOR model to support supply chain re-design. International journal of information management, 34 (5), 634-638.

PALMA-MENDOZA, Jaime, NEAILEY, Kevin and ROY, Rajat (2014). Business process re-design methodology to support supply chain integration. International journal of information management, 34 (2), 167-176.

PANNELL, David J. (1997). Sensitivity analysis: strategies, methods, concepts, examples. Agric econ, 16, 139-152.

PARMIGIANI, Anne, KLASSEN, Robert D. and RUSSO, Michael V. (2011). Efficiency meets accountability: Performance implications of supply chain configuration, control, and capabilities. Journal of operations management, 29 (3), 212-223.

PERSSON, Fredrik and ARALDI, Mirko (2009). The development of a dynamic supply chain analysis tool—Integration of SCOR and discrete event simulation. International journal of production economics, 121 (2), 574-583.

PETROVIC, Dobrila (2001). Simulation of supply chain behaviour and performance in an uncertain environment. International journal of production economics, 71 (1), 429-438.

PIRARD, Florence, IASSINOVSKI, Serguei and RIANE, Fouad (2011). A simulation based approach for supply network control. International journal of production research, 49 (24), 7205-7226.

PONIS, Stavros T., et al. (2013). Modeling supply chain processes: a review and critical evaluation of available reference models. In: Proceedings of the 2nd international symposium and 24th national conference on operational research, Hellenic Operational Research Society Athens, 270-276.

POPPER, Karl (1994). The Myth of the Framework., In defence of science and rationality.

PRASAD, K. G. D., SUBBAIAH, K. V. and RAO, K. N. (2014). Supply chain design through QFD-based optimization. Journal of manufacturing technology management, 712-733.

PricewaterhouseCoopers and American Production and Inventory Control Society(2014), Sustainable supply chains: Making value the priority, P6

QRUNFLEH, S. and TARAFDAR, M. (2013). Lean and agile supply chain strategies and supply chain responsiveness: The role of strategic supplier partnership and postponement. Supply chain management: An international journal, 18 (6), 571-582.

QRUNFLEH, Sufian and TARAFDAR, Monideepa (2013). Lean and agile supply chain strategies and supply chain responsiveness: the role of strategic supplier partnership and postponement. Supply chain management, 18 (6), 571-582.

RANDALL, Taylor R., MORGAN, Ruskin M. and MORTON, Alysse R. (2003). Efficient versus responsive supply chain choice: an empirical examination of influential factors. Journal of product innovation management, 20 (6), 430-443.

RAO, P. H. (2014). Measuring Environmental Performance across a Green Supply Chain: A Managerial Overview of Environmental Indicators. Vikalpa, 39 (1), 57-74. RASHID, Shahid and WESTON, Richard (2012). Design of an integrated methodology for analytical design of complex supply chains. Advances in decision sciences, 2012.

REDDI, Krishna R. and MOON, Young B. (2013). Modelling engineering change management in a new product development supply chain. International journal of production research, 51 (17), 5271-5291.

REZAPOUR, Shabnam, et al. (2014). Designing a new supply chain for competition against an existing supply chain. Transportation research part E: Logistics and transportation review, 67, 124-140.

REZAPOUR, Shabnam, et al. (2015). Competitive closed-loop supply chain network design with price-dependent demands. Journal of cleaner production, 93, 251-272.

Roland Berger Strategy Consultants, (2009). Global SCM excellence study, p.5.

ROH, James, HONG, Paul and MIN, Hokey (2014). Implementation of a responsive supply chain strategy in global complexity: The case of manufacturing firms. International journal of production economics, 147, 198-210.

ROMERO, Juan Camilo, et al. (2012). Collaborative Methodology for Supply Chain Quality Management: Framework and Integration With Strategic Decision Processes in Product Development. Proceedings of the european conference on information management & evaluation, , 418-427.

ROMSDAL, Anita, STRANDHAGEN, Jan Ola and DREYER, Heidi Carin (2014). Can differentiated production planning and control enable both responsiveness and efficiency in food production? Proceedings in food system dynamics, 324-333.

RONI, Md S., et al. (2014). A supply chain network design model for biomass co-firing in coal-fired power plants. Transportation research: Part E, 61, 115-134.

SABET, Ehsan, et al. (2017). Supply chain integration strategies in fast evolving industries. The international journal of logistics management, 28 (1), 29-46.

SABRI, Ehap H. and BEAMON, Benita M. (2000). A multi-objective approach to simultaneous strategic and operational planning in supply chain design. Omega, 28 (5), 581-598.

SANDERS, Nada R. (2011). Supply chain management: A global perspective. Wiley Global Education.

SANTA-EULALIA, Luis-Antonio, et al. (2011). Agent-based experimental investigations of the robustness of tactical planning and control policies in a softwood lumber supply chain. Production planning & control, 22 (8), 782-799.

SATYA, Parkash Kaushik and VEERENDER, Kumar Kaushik (2014). A Simplified, Result Oriented Supplier Performance Management System Testing Framework for Sme. Logforum, 10 (2), 191-203.

SCHALTEGGER, Stefan and BURRITT, Roger (2014). Measuring and managing sustainability performance of supply chainsReview and sustainability supply chain management framework. Supply chain management, 19 (3), 232-241.

SEURING, Stefan, et al. (2008). Sustainability and supply chain management-an introduction to the special issue, .

SHAHI, Shashi and PULKKI, Reino (2013). Supply Chain Network Optimization of the Canadian Forest Products Industry: A Critical Review. American journal of industrial and business management, 3 (07), 631.

SHAHZAD, Kashif M. and HADJ-HAMOU, Khaled (2013). Integrated supply chain and product family architecture under highly customized demand. Journal of intelligent manufacturing, 24 (5), 1005-1018.

SHAO, Junyang, MOSER, Roger and HENKE, Michael (2012). Multidimensional supply performance framework: A conceptual development and empirical analysis. International journal of production economics, 138 (1), 26-34.

SHAO, Junyang, MOSER, Roger and HENKE, Michael (2012). Multidimensional supply performance framework: A conceptual development and empirical analysis. International journal of production economics, 138 (1), 26-34.

SHEN, Z. (2007). Integrated supply chain design models: a survey and future research directions. Journal of industrial and management optimization, 3 (1), 1.

SHUKLA, Aviral, LALIT, Vishal Agarwal and VENKATASUBRAMANIAN, Venkat (2011). Optimizing efficiency-robustness trade-offs in supply chain design under uncertainty due to disruptions. International journal of physical distribution & logistics management, 41 (6), 623-647.

SINGH, Rajesh Kumar and SHARMA, Milind Kumar (2014). Prioritising the alternatives for flexibility in supply chains. Production planning & control, 25 (2), 176-192.

SOMUYIWA, Adebambo and ADEBAYO, Toyin (2013). Firm's competitiveness through supply chain responsiveness and supply chain management practices in Nigeria. Journal of poverty, investment and development, 1, 142-147.

SONI, Gunjan and KODALI, Rambabu (2013). A critical review of supply chain management frameworks: proposed framework. Benchmarking: An international journal, 20 (2), 263-298.

SPEIER, Cheri, et al. (2011). Global supply chain design considerations: Mitigating product safety and security risks. Journal of operations management, 29 (7), 721-736.

SPEIER, Cheri, et al. (2011). Global supply chain design considerations: Mitigating product safety and security risks. Journal of operations management, 29 (7), 721-736.

Stefan Schaltegger, Prof Roger Burritt, Dr, et al. (2014). Framing sustainability performance of supply chains with multidimensional indicators. Supply chain management: An international journal, 19 (3), 242-257.

SUKATI, Inda, et al. (2012). Competitive advantage through supply chain responsiveness and supply chain integration. International journal of business and commerce (IJBC), 1 (7).

SYNTETOS, Aris A. and BOYLAN, John E. (2005). The accuracy of intermittent demand estimates. International journal of forecasting, 21 (2), 303-314.

TAKO, Antuela A. and ROBINSON, Stewart (2012). The application of discrete event simulation and system dynamics in the logistics and supply chain context. Decision support systems, 52 (4), 802-815.

TATICCHI, Paolo, TONELLI, Flavio and PASQUALINO, Roberto (2013). Performance measurement of sustainable supply chains A literature review and a research agenda. International journal of productivity & performance management, 62 (8), 782-804.

TERZI, Sergio and CAVALIERI, Sergio (2004). Simulation in the supply chain context: a survey. Computers in industry, 53 (1), 3-16.

THATTE, A. A., RAO, S. S. and RAGU-NATHAN, T. S. (2013). Impact of SCM practices of a firm on supply chain responsiveness and competitive advantage of a firm. Journal of applied business research, 29 (2), 499-530.

TSAO, Yu-Chung (2015). Design of a carbon-efficient supply-chain network under trade credits. International journal of systems science: Operations & logistics, (ahead-of-print), 1-10.

TSENG, Tzu-Liang, GUNG, Roger R. and HUANG, Chun-Che (2013). Performance Evaluation for Pull-Type Supply Chains Using an Agent-Based Approach.

UM, Juneho and UM, Juneho (2017). Improving supply chain flexibility and agility through variety management. The international journal of logistics management, 28 (2), 464-487.

UM, Juneho, et al. (2017). Product variety management and supply chain performance: A capability perspective on their relationships and competitiveness implications. [online]. International Journal of Production Economics, 187 (Supplement C), 15-26. http://www.sciencedirect.com.lcproxy.shu.ac.uk/science/article/pii/S0925527 31730035X.

UMEDA, Shigeki and JAIN, Sanjay (2004). Integrated Supply Chain Simulation System (ISSS)-Modeling Requirements and Design Issues. National institute of standards and technology, US department of commerce.washington, DC, .

VAHDANI, B., ZANDIEH, M. and ROSHANAEI, V. (2011). A hybrid multistage predictive model for supply chain network collapse recovery analysis: a practical framework for effective supply chain network continuity management. International journal of production research, 49 (7), 2035-2060.

VAN KAMPEN, Tim and VAN DONK, Dirk Pieter (2014). Coping with product variety in the food processing industry: the effect of form postponement. International journal of production research, 52 (2), 353-367.

VARSEI, M., et al. (2014). Framing sustainability performance of supply chains with multidimensional indicators. Supply chain management, 19 (3), 242-257.

VERDOUW, C. N., et al. (2011). A framework for modelling business processes in demand-driven supply chains. Production planning & control, 22 (4), 365-388.

VIDALAKIS, Christos, TOOKEY, John E. and SOMMERVILLE, James (2011). The logistics of construction supply chains: the builders' merchant perspective. Engineering, construction and architectural management, 18 (1), 66-81.

VLAJIC, Jelena V., VAN, der Vorst and HAIJEMA, René (2012). A framework for designing robust food supply chains. International journal of production economics, 137 (1), 176-189.

WAGNER, Stephan M., GROSSE-RUYKEN, Pan Theo and ERHUN, Feryal (2012). The link between supply chain fit and financial performance of the firm. Journal of operations management, 30 (4), 340-353.

WANG, HS (2009). A two-phase ant colony algorithm for multi-echelon defective supply chain network design. European journal of operational research, 192 (1), 243-252.

WANG, Jun (2005). A review of operations research applications in workforce planning and potential modeling of military training, .

WANG, Sufang, FANG, Feihu and CHENG, Nuo (2011). Research on the framework of supply chain network design. In: Artificial intelligence, management science and electronic commerce (AIMSEC), 2011 2nd international conference on, IEEE, 1451-1454.

WANG, Fan, LAI, Xiaofan and SHI, Ning (2011). A multi-objective optimization for green supply chain network design. Decision support systems, 51 (2), 262-269.

WIELAND, Andreas (2013). Selecting the right supply chain based on risks. Journal of manufacturing technology management, 24 (5), 652-668.

WILMSMEIER, Gordon, et al. (2014). Efficiency: a key ingredient towards sustainable supply chains.

WILSON, Martha C. (2007). The impact of transportation disruptions on supply chain performance. Transportation research part E: Logistics and transportation review, 43 (4), 295-320.

XU, Jiuping, LIU, Qiang and WANG, Rui (2008). A class of multi-objective supply chain networks optimal model under random fuzzy environment and its application to the industry of Chinese liquor. Information sciences, 178 (8), 2022-2043.

XU, He, LIU, Zhen Zhen and ZHANG, Sheng Hao (2012). A strategic analysis of dual-channel supply chain design with price and delivery lead time considerations. International journal of production economics, 139 (2), 654-663.

YOUN, S., YANG, M. G. and ROH, J. J. (2012). Extending the efficient and responsive supply chains framework to the green context. Benchmarking, 19 (4), 463-480.

YU, Wantao, et al. (2014). Integrated green supply chain management and operational performance. Supply chain management: An international journal, 19 (5/6), 683-696.

ZAILANI, Suhaiza, et al. (2012). Sustainable supply chain management (SSCM) in Malaysia: A survey. International journal of production economics, 140 (1), 330-340.

ZANJANI, Masoumeh Kazemi, AIT-KADI, Daoud and NOURELFATH, Mustapha (2010). Robust production planning in a manufacturing environment with random yield: A case in sawmill production planning. European journal of operational research, 201 (3), 882-891. ZHANG, Abraham, LUO, Hao and HUANG, George Q. (2013). A bi-objective model for supply chain design of dispersed manufacturing in China. International journal of production economics, 146 (1), 48-58.

ZHOU, Honggeng, et al. (2011). Supply chain integration and the SCOR model. Journal of business logistics, 32 (4), 332-344.